## IN SITU TECHNO-ENVIROECONOMIC ANALYSIS OF A ROOFTOP PHOTOVOLTAIC (PV) BACKUP SYSTEM IN THE TROPICAL CLIMATE

BY

## WILKINS KOSGEI CHERUIYOT

# A THESIS SUBMITTED IN PARTIAL FULFILMENT OF THE REQUIREMENTS FOR THE AWARD OF DEGREE OF DOCTOR OF PHILOSOPHY IN PHYSICS, UNIVERSITY OF ELDORET, KENYA

JULY, 2022

#### DECLARATION

#### **Declaration by the Candidate**

This thesis is my original work and has not been presented for a degree in any other University. No part of this thesis may be reproduced without the prior written permission of the author and/ or University of Eldoret.

**Cheruiyot Wilkins Kosgei** 

#### SC/PHD/011/16

Date

#### **Declaration by the Supervisors**

This thesis has been submitted for examination with our approval as University Supervisors.

**Prof Joel K. Tonui** Department of Physics University of Eldoret, Kenya

**Dr. Samuel L. Chelimo** Department of Physics University of Eldoret, Kenya. Date

Date

#### DEDICATION

To my Dad George Sirma Kosgei for his unconditional love, advice and support. To my dear wife Sellah Jesang, for her love and care and to my lovely children Abigael Jelagat and Abijah Jelimo.

To the faculty and staff and colleagues of the University of Eldoret, Physics department.

To my mentor and friend, Mr. Mbeche Fanuel, I couldn't have done this without you. To my friends and neighbours, thank you all for your support along the way.

Above all, I dedicate this work to our almighty God who showers us His blessings in our everyday lives, especially for the strength, courage, patience, wisdom, time, and guidance in realization of this work.

#### ABSTRACT

Solar PV generators are increasingly being deployed in the built environment as stand-alone or backup power systems to supply electricity either solely or during power outages respectively. Diesel generators (DGs) are applied routinely as standby power systems by many enterprises and institutions, especially in developing countries where power outages are real and frequent due to unstable national grids. The current pursuit of low-carbon and sustainable source of energy places photovoltaic (PV) power system in advantageous position as a substitute for a DG backup system. An existing off-grid 780 Wp PV system installed as a backup power supply in a learning institution in western part of Kenya was studied both experimentally and theoretically. Technical, economic and environmental analyses were carried out to determine its performance under the local outdoor conditions at the site for a period of one year in 2020. Irradiance estimation models were also validated by experimental data to determine appropriate model(s) for the site. Planeof-array (POA) solar radiation was measured with solar cell sensor installed at the surface of PV modules and a charge controller/inverter unit with the capability to measure and log real time output current (I) and voltage (V) was used to generate I-V characteristic data. PVsyst software was used to simulate the PV system and generate optimized theoretical results for the site which were compared with experimental results. Available energy was determined as 3202.80 kWh/year, average array efficiency of 11.71%, FF of 0.66, array yield of 4.89 kWh/kW, reference yield of 5.51 kWh/kW, capacity factor as 19.8%, annual average performance ratio (PR) as 76.0%, and average array losses as 0.54 kWh/kW. Economic results for the PV system show that the payback time (PBT) is~ 6.38 years, LCC of \$3057.93, levelized cost of energy (LCOE) of 0.045 \$/kWh and operation and maintenance (O&M) is 17.32 \$/year. For the diesel generator (DG),PBT was 4.25 years and O&M was 262.80 \$/year for a lifespan of 5 years and assuming that it operates 2 hours per day of blackouts, LCC of \$7792.75 and LCOE of 0.324 \$/kWh. Environmental results show that the total annual amount of CO<sub>2</sub> emissions avoided when PV is used instead of DG power backup system was 5.84 tCO<sub>2</sub>/year giving an average cost parameter (penalty for CO<sub>2</sub> generation) of \$9.62. Simulation results gave the available energy as 3746.40 kWh/year, reference yield of 5.55 kWh/kW, array yield of 4.18 kWh/kW, array losses of 0.61 kWh/kW, capacity factor as 21.23%, FF as 0.68, PR as 73.6% and PV array efficiency of 13.19%. The average amount of CO<sub>2</sub> emission avoided was 7.95 tCO<sub>2</sub>/year with annual environmental cost of \$116.18. Angstrom-Prescott and Iqbal models were found to be the most accurate for the site having the lowest values of mean absolute percentage error (MAPE) of 8.5% and 8.9% and root mean square error (RMSE) of 0.252 and 0.302 respectively. In conclusion, technically, the low FF (<<1) indicate that the system is not operating at its optimum, which can be attributed to how the PV system was installed. The LCOE results show that PV power is cheaper by a factor of seven than that of the diesel generator, and the amount of  $CO_2$ avoided is at least 0.44 tCO<sub>2</sub>/month. The PV power presents net benefits over diesel power in all performance indices evaluated, and hence can be used as a reliable and affordable replacement for DG backup systems in tandem to the global quest to transition to clean and sustainable energy sources and attainment of the SDGs 7 and 13.

## TABLE OF CONTENTS

DECLARATIONii
DEDICATION
ABSTRACTiv
TABLE OF CONTENTSv
LIST OF TABLESix
LIST OF FIGURESx
LIST OF ABBREVIATIONS AND SYMBOLSxi
ACKNOWLEDGEMENT
CHAPTER ONE
INTRODUCTION1
1.1 Background1
1.2 Electricity Generation from Solar PV Systems
1.3 Progress in solar PV Technology
1.4 Statement of the problem
1.5 Research Objectives
1.5.1 General objective
1.5.2 Specific objectives
1.6 Justification of the study14
1.7 Significance of the study15
1.8 Scope and Limitations of the study15
1.9 Thesis organization

CHAP	ΓER	TWO	17
THEOI	RY A	AND LITERATURE REVIEW	17
2.1	Int	troduction	17
2.2	Ele	ectricity in Kenya	17
2.3	So	lar Radiation	22
2.3	3.1	Ground Observation	25
2.3	3.2	Interpolation method	26
2.3	3.3	Satellite Remote Sensing	27
2.3	3.4	Empirical methods	27
2.3	3.5	Clearness index	32
2.4	So	olar energy conversion technologies	33
2.5	Th	e PV cells	35
2.6	Th	e PV module and array	39
2.7	PV	/ performance	41
2.7	7.1	Single diode model	41
2.7	7.2	PV module model	43
2.8	So	olar PV application	43
2.9	Th	e PV Balance of System	45
2.9	9.1	Inverters	45
2.9	9.2	Energy Storage systems	46
2.10	Ph	otovoltaic system performance	50

2.10.	1 Efficiency of PV cell/module	51
2.10.	2 Effect of solar irradiance on PV performance	51
2.10.	3 Effect of ambient temperature on PV performance	53
2.10.4	4 Installation setup and performance	55
2.11 I	EC Standards	56
2.12 I	Literature review	57
CHAPTE	R THREE	67
METHOI	DOLOGY AND MATERIALS	67
3.1 I	Introduction	67
3.2	Study location	67
3.3 I	Experimental setup	69
3.3.1	The PV modules	70
3.3.2	Inverter/ Controller unit	71
3.3.3	Battery Storage System	73
3.3.4	Loads	73
3.4 I	Instrumentation and data collection	74
3.5 I	Performance functions of PV modules	75
3.5.1	Efficiency, fill factor and output energy	76
3.5.2	Yield factors, performance ratio and losses	77
3.5.3	Capacity factor	79
3.6 I	Economic performance indicators	79

3.6	5.1 Life cycle cost analysis	80
3.6	5.2 Levelized cost of energy	81
3.7	Environmental analysis	82
3.8	Diesel generator fuel consumption model	83
3.9	PVSyst Simulation tool	84
CHAP	TER FOUR	86
RESUL	TS AND DISCUSSIONS	86
4.1	Introduction	86
4.2	Operating conditions	86
4.3	Electrical performance of the PV system	92
4.3	I-V characteristic of PV modules	92
4.3	E.2 Energy Production Performance parameters	94
4.4	Economic performance	97
4.5	Environmental performance of the PV system	101
CHAPT	TER FIVE	104
CONC	LUSIONS AND RECOMMENDATIONS	104
5.1	Conclusions	104
5.2	Recommendations	107
LIST O	F REFERENCES	108
APPEN	IDICES	137

## LIST OF TABLES

Table 3. 1: Technical specification of PV module	71
Table 4. 1: Comparison performance of the four selected models on predictin	g solar
irradiance at the site	91
Table 4. 2: Electrical PV module performance parameters.	94
Table 4. 3: Monthly average values of energy performance parameters	95
Table 4. 4: Summary of PV system versus diesel generator economic perform	nance 100

## LIST OF FIGURES

Figure 1. 1: Annual PV installations globally (Source: IEA, 2019)11
Figure 2. 1: The proportion trend of electricity generated by source from 2016 to 2020 (Source: Economic survey, 2021; KPLC, 2020)17
Figure 2. 2: Spectral distribution of extraterrestrial radiation. Source: Renewable resource data center (Duffie and Beckman, 2013)
Figure 2. 3: Structure of a solar PV cell. (Luceño-Sánchez et al., 2019)36
Figure 2. 4: Schematic showing the classification of solar PV cells
Figure 2. 5: Structure of a PV module with 72 cells wired in series. (Source: Müller <i>et al.</i> , 2016)40
Figure 2. 6: Basic structure of a PV array
Figure 2. 7: Equivalent circuit of a PV cell (Dadjé et al., 2017)42
Figure 2. 8: Classification of PV systems (Messenger and Ventre, 2003)44
Figure 2. 9: Classification of energy storage systems
Figure 2. 10: TheI-V characteristic at varying solar irradiance under constant temperature (Source: PVSyst simulation software library)
Figure 2. 11: I-V characteristics of a PV module at constant solar irradiance (Source: PVSyst simulation software library)
Figure 3. 1: (a) Map of Kenya showing study region - Nandi County; (b) Map of Mosop constituency showing the study site
Figure 3. 2: Schematic diagram of the studied PV backup system69
Figure 3. 3: Photograph of roof-mounted PV Array at the Institute's70
Figure 3. 4: Picture of Integrated Inverter/Controller Unit72
Figure 3. 5: Picture of Battery Bank System73
Figure 3. 6: A picture of a digital recorder and the pyranometer(model SPM-1116SD)
Figure 3. 7: Modified schematic backup system layout in PVSyst software
Figure 4. 1: Measured solar irradiance representative results
Figure 4. 2: Monthly clearness index
Figure 4. 3: Monthly average ambient temperatures and wind speeds at the site89
Figure 4. 4: Comparison between estimated (by the four models) and measured values of monthly average Daily global solar irradiance at the site90
Figure 4. 5: PV module I-V characteristics
Figure 4. 6: PV backup system versus diesel generator life cycle costs
Figure 4. 7: Monthly carbon reduction and environmental cost102

## LIST OF ABBREVIATIONS AND SYMBOLS

А	Module area (m <sup>2</sup> )
a-Si	Amorphous Silicon
BOS	Balance of system
CF	Capacity factor
CO <sub>2</sub>	Carbon dioxide
COVID	Corona-virus disease
CSP	Concentrating solar power
CSV	Comma-separated values
DAQ	data acquisition systems
DG	Diesel generator
DHI	Diffuse Horizontal Irradiance
DNI	Direct Normal Irradiance
$E_{Dc}$	DC output energy (kWh)
FF	Fill Factor
FiT	Feed-in Tariff
$G_{t}$	Measured solar irradiance (W/m <sup>2</sup> )
GHG	Greenhouse gases
GHI	Global Horizontal Irradiance
I <sub>mp</sub>	Nominal current (A)
I <sub>sc</sub>	Short-circuit current (A)
I <sub>max</sub>	Maximum current (A)
I <sub>mp</sub>	Maximum power current (A)
IEC	International Electrotechnical Commission

IEEE	Institute of Electrical and Electronic Engineers
IRENA	International renewable energy agency
$J_{sc}$	Short-circuit current density
KAM	Kenya Association of Manufacturers
KenGen	Kenya electricity generating company
KOSAP	Kenya Off-Grid Solar Access Project
KPLC	Kenya power and lighting company
KNBS	Kenya national bureau of statistics
KTTI	Kaiboi Technical Training Institute
$L_{ m A}$	Array capture loss (kWh/kW)
LCE	Life cycle emissions
LCOE	Levelized cost of energy
MAPE	Mean absolute percentage error
MBE	Mean bias error (MBE),
MPPT	Maximum Power Point Tracking
NPV	Net present value
$P_o$	Module rated power (W)
P <sub>max</sub>	STC power rating (W)
pc-Si	Poly-crystalline silicon
POA	Plane of array
PPA	Power Purchase Agreement
PR	Performance ratio
PSH	Peak Solar Hour
PV	Photovoltaic
PV/T	Photovoltaic/Thermal

PWM	Pulse Width Modulation
$R^2$	Coefficient of determination
R&D	Research and Development
$R_s$	Series resistance
$R_{sh}$	Shunt resistance
RES	Renewable energy sources
RMSE	Root mean square error
ROI	Return on investment
SDG	Sustainable Development Goals
SE4All	Sustainable Energy for All
SHC	Solar heating and cooling
SHS	Solar home system
SDHW	Solar domestic hot water heating system
SSA	Sub-Sahara Africa
STE	Solar thermal electricity
STC	Standard Test conditions
t-stat	<i>t</i> -statistic method
TC	Thermal collector
$T_c$	Cell temperature ( <sup>0</sup> C)
$T_{amb}$	Ambient temperature ( <sup>0</sup> C)
T <sub>NOCT</sub>	Nominal operating cell temperature ( <sup>0</sup> C)
V <sub>t</sub>	Thermal voltage (V)
V <sub>mp</sub>	Maximum power voltage (V)
V <sub>max</sub>	Maximum voltage (V)

V <sub>oc</sub>	Open circuit voltage (V)
VRB	Vanadium Redox Battery
WEO	World economic outlook
Y <sub>A</sub>	Array yield (kWh/kW)
Y <sub>r</sub>	Reference yield (kWh/kW)
$\eta_{cell}$	Cell efficiency (%)
β	Cell temperature parameter (K <sup>-1</sup> )
$\eta_{\scriptscriptstyle Array}$	PV array efficiency (%)
$\eta_{\scriptscriptstyle ref}$	Reference module efficiency (%)

#### ACKNOWLEDGEMENT

I am always thankful to the Almighty God for perseverance, achievements and for His grace upon my life throughout my Ph.D. work.

I am grateful to the University of Eldoret for giving me the opportunity to pursue a Doctorate degree programme in Physics.

I also use this opportunity to thank my Supervisors, Prof. Joel K. Tonui and Dr. Samuel C. Limo for their supervision, guidance and support, which greatly contributed to the success of this thesis. They spent their quality time throughout the research review in advising and making healthy suggestions and corrections to improve the quality of this thesis.

I acknowledge and appreciate Kaiboi Technical Training Institute for allowing me to use their PV power backup system for data collection. I also recognize the head of KTTI Electrical and Electronics Department Mr. Richard A. Kinyamal for his support and technical knowledge during data collection process.

Finally, I salute all my lecturers, Department of Physics and fellow postgraduate students who constantly encouraged me. I want to appreciate my wife, Sellah for her prayers, understanding and support throughout my Ph.D programme.

#### **CHAPTER ONE**

#### **INTRODUCTION**

#### **1.1 Background**

Energy is an essential resource globally for economic, social development and improvement of quality of life. However, energy creation and use have far-reaching consequences, affecting big themes such as climate change and geo-political factors, as well as many more "trivial" elements of our daily lives.

Energy is an essential resource globally for economic and social development in any country. Efficient use of energy comes with benefits such as industrial development and improvement of the standards of living. On the contrary, energy generation and use have far-reaching consequences, which have created the present big themes such as climate change and geo-political factors, as well as many more "trivial" elements of our daily lives. Several important aspects of energy are currently experiencing a period of transformation. These include energy production (with regards to both fossil fuels and renewable energy sources); transportation and distribution (e.g., electricity and distribution networks); consumption (including demand-side management and energy efficiency); and energy security and access (Kristie et al., 2019). Energy transitions also play a crucial role in climate change mitigation. Energy technologies are developing rapidly, allowing new and more efficient ways to deliver energy on varying scales. Globally, energy is evolving from a centralized system, with large production plants, to a distributed system, in which an individual becomes a "prosumer", both producing and consuming energy (World Bank, 2021). As a result, consumers have an increasingly important role to play in improving the efficiency of these systems.

Electricity, from a practical viewpoint, catalyzes economic productivity and industrial growth and is central to the operation of any modern economy and livelihood. Provision of electricity is now considered globally as a necessity along with food and water (Hemza et al., 2019). However, universal access to electricity is highly skewed with the majority if not all households in developed countries connected to grid electricity, and vice-versa in the developing and transitioning economies like Africa and South Asia (Corfee-Morlot et al., 2019; World Bank, 2019). The percentage of people with access to electricity has been steadily increasing over the last few decades where a rise of more than 50% of new connections has been realized between the year 2000 and 2021 (World Bank, 2021). Electricity is a high quality and versatile form of energy and lack of access to it leads to adverse socio-economic consequences. According to the World Bank technical report (ESMAP, 2007), lack of access to electricity services entrenches poverty, constrains the delivery of social services, limits opportunities for women and girls and erodes environmental sustainability at the local, national and global levels. The lack of universal access to electricity by all in the world, prompted global agencies such as the United Nation (UN), the World Bank Group and others to come up with strategies to provide electricity for all at the start of this millennium or 21<sup>st</sup> century (IEA, IRENA, UNSD, WB, WHO, 2021).

Global campaign for universal electrification across the world begun with the Millennium Development Goals (MDGs) in 2000 at the beginning of 21<sup>st</sup> century. However, the MDGs initiative did not capture explicitly energy sector unlike its predecessor, the Sustainable Development Goals (SDGs) which has a substantive agenda on energy, SDG 7 (Kumar *et al.*, 2016). The 17 SDGs, cover everything from energy and climate; to water, food and ecosystems; to health and poverty; to jobs and innovation; education, gender equity, among a number of other objectives (McCollum

et al., 2018). The SDG 7 is subdivided into three distinct, but related targets or pillars to be achieved by the year 2030 (UN, 2015; McCollum et al., 2018; Neriniet al., 2018; Warner and Jones, 2017). The SDG Target 7.1 is to ensure universal access to affordable, reliable, and modern energy services (7.1.1 focuses on the proportion of the population with access to electricity and 7.1.2, on the proportion relying primarily on clean fuels and technologies for cooking). Target 7.2 is to increase substantially the share of renewable energy in the global energy mix, while Target 7.3 is to double the global rate of improvement in energy efficiency (UNDP, 2015). Another global initiative for universal access to electricity is the Sustainable Energy for All (SE4All) of 2012 (AfDB, 2015). The SDG 7 is a fundamental objective to sustainable development, because all the other SDGs are intrinsically linked to energy. This means that the efforts undertaken to achieve its targets may either facilitate or compromise achievement of the targets of other SDGs. Since Kenya is a UN member, it is obliged to ratify the SDG goals, and the country employed institutional approach to achieve the SDG targets within the specified time frame. The country established the Rural Electrification Authority (REA) in 2006 and national development blue print document prominently called 'Kenya Vision 2030' (GoK, 2007). The REA was tasked to hasten rural electrification in the country through extension of national grid in order to promote sustainable socioeconomic development by 2030 (Roche & Blanchard, 2018). Following the enactment of Energy Act of 2019 (Act, EPRA, 2019), REA has been transformed to Rural Electrification and Renewable Energy Corporation (REREC) and its mandate expanded to include also Target 7.2. Thus, REREC is tasked with implementing rural electrification projects (Target 7.1.1) and spearheading Kenya's green energy drive (Target 7.2).

Reliable supply of electricity is crucial for both modern livelihood and economy. Electrical power outages lead to a variety of negative business outcomes, such as reduced sales (loss of revenue), damaged machinery (reduce productivity), data loss, loss of customers, and product spoilage (Maende and Alwanga, 2020). In addition, it disrupts daily life; interfere with services such as lighting, refrigeration and air conditioning as well as causing tragic consequences to critical health and social services, such as life-saving medical devices and to the extreme end loss of life. Power outages in developing countries are rampant and are made worst with increasing severe weather events and disasters as well as aging grids (Farquharson *et al.*, 2018). The World Bank Enterprise Surveys report that most enterprises in Africa encounter regular power outages, which could number to as high as 100 times in a month (The World Bank, Enterprise Surveys, 2018). The grid electricity is vulnerable to a range of mechanical, operational, environmental, and human-related hazards (Preston *et al.*, 2016). Mechanical faults, line shorts, fires, and human interference are common causes of outages during transmission.

Power outages have compelled many businesses, institutions and even individuals to seek self-generation options. The widely adopted option, in developing countries mainly, is the diesel generators (portable or immovable) because they are readily available and have a relatively low initial investment cost, but carry health and environmental risks (Babajide and Brito, 2021). In addition, the cost of running diesel generator can be quite high, especially in countries where gasoline prices are high. Solar photovoltaic systems offer a cleaner viable option for the African continent because it is endowed with abundant solar resource throughout the year. This solution is suitable for individuals and businesses alike because of its modularity and do not incur high running cost unlike the traditional DG, but has not been embraced in the

African continent, including Kenya. Therefore, for decision makers or prospective consumers to make informed choices, it is important to understand the cost and reliability associated with Solar PV backup power in order to be accepted as a possible replacement of the present fuel-power generators.

#### 1.2 Electricity Generation from Solar PV Systems

The sun is the driving force for all atmospheric processes, and every known form of energy, except nuclear and geothermal energy, originates from the sun, either directly or indirectly. Solar energy refers to energy that is directly attributed to the light of the sun or the heat that sunlight generates (Timilsina *et al.*, 2011). It is a huge resource and is considered as the only inexhaustible renewable energy (RE) source because it will last as long as the sun exists and has minimal impact on the environment (Freris and Infield, 2008). It can be harnessed using modern technologies to provide the two widely sought forms of energy viz electricity and heat energy. Since solar energy is dilute (spread throughout the earth's surface facing the sun) and intermittent (not available at night and non-steady during the day), it needs to be collected, concentrated and stored for it to be useful. Solar collectors are devices used to collect and concentrate solar energy to make it adequate and available continuously. Two commonly used types of solar collectors are photovoltaic (PV) and solar thermal collector (TC).

The PV converts sunlight directly into electricity while TC converts sunlight into heat energy through a working fluid (water or air or any other heat transfer fluid). These two technologies are available commercially and have been deployed all over the world but to different degrees with developed countries leading the developing countries. The skewed deployment has been attributed to technological, financial and institutional barriers, which are experienced mainly in developing countries (Timilsina *et al.*, 2011). The PV applications have been growing exponentially across the world, with installed capacity of more than 750 GW by the end of 2020, and now contribute ~3% of total electricity in the world (IEA PVPS, 2020). The present development in PV is characterized by ongoing research that aim at increasing the low field efficiencies of about 15% to 20% currently in order to reduced PV electricity cost and expand its applications globally.

Photovoltaic has the potential to provide electricity in every corner of Kenya, both in urban dwellings and remote rural villages(Kiprop et al., 2018). PV can promote development across all sectors of economy, especially in regions with enough solar radiation to meet electricity needs and cost of the consumers (Jiménez-Castillo et al., 2019). PV applications are usually classified as either solar home system (SHS), minigrid, grid-connected or hybrid system (PV-thermal power, PV-wind, or PV-hydro). However, SHS and mini-grid systems can play important role in rural electrification in any country (Barnes and Foley, 2004). The SHS has been cited in literature as having provided individual solution, while mini-grid can offer a collective solution at a relatively lower cost. In addition, the mini-grid can facilitate basic needs as well as productive use of electricity, and hence promoting local economic engagements (Bhattacharyya and Palit, 2016). For these PV systems to be competitive, they should be able to supply reliable electricity to support income generation activities for the poor on a regular basis. Grid-connected PV system consists of a large number of PV modules interconnected in series-parallel to form a PV array that can generate a great amount of electricity (hundreds of kW to MW), which is then fed to the national grid (Felten et al., 2006). Hybrid PV system, on the other hand, consists of a PV generator combined with convectional electricity generation plants such as thermal power (fossil fuels), wind generator or mini-hydropower plants. In hybrid systems, generators are fused together to complement each other with respect to time mainly and hence supply more reliable and stable electricity than the individual system throughout the year (Kougias *et al.*, 2016).

Solar electricity can be cost competitive in many electricity markets today if solar panels can perform to warranted specifications for at least the length of their warranty, which is typically 25 years. But it can be difficult to identify which PV modules will meet their warranted performance level, and hence the need for long term outdoor performance test of different PV technologies at the site of application (Olchowik et al., 2006; Kurtz and Jordan, 2013; Bashir et al., 2014). Power output of a PV module is reported by manufacturers at standard test conditions (STC), which are defined by international standard IEC 61836 as one-sun irradiance of 1000 W/m<sup>2</sup>, cell temperature of 25°C, air mass 1.5 and normal incidence (IEC TS 61836, 2016) or PVUSA Test Conditions (PTC) with same irradiance but ambient temperature  $(T_{amb})$ of 20 °C (Topic et al., 2011). However, these conditions cannot be experienced under realistic outdoor operation anywhere in the world. In real installed PV systems, the module output is strongly affected by various environmental conditions such as irradiance, temperature, spectral effects, angle of incidence, wind speed and direction. In addition, different PV technologies have different patterns of behaviour with respect to each climatic factor on amount of energy produced. Hence, modules of different technologies may be more suited to certain specific climate (Carr& Pryor, 2004). In order to help PV users to make more accurate performance predictions and assist them in system planning and financing, it is important to evaluate or predict the performance of the module at the local climatic conditions.

Monitoring of outdoor performance of a PV module is challenging because the operating outdoor weather conditions change simultaneously making it difficult to investigate the effect of one parameter independently. In most cases, these changes are intertwined to one another and it becomes difficult to separate the contribution of each effect on the overall performance of the PV module. In tropical regions, weather conditions tend to be constant throughout the year except during rainy season where some days are fully or partially overcast. The temperature affects the performance of PV module either negatively or positively depending on the cell technology. The effect of air mass is due to the filtering of incident solar radiation beam by atmospheric constituents, which affect its spectral distribution (Chantana *et al.*, 2020). Since these factors vary geographically and seasonally as well as time of the day, it is important to investigate experimentally or model their effects on different PV technologies in order to characterize them and hence help the users purchase cost-effective modules for the site.

A wealth of literature on performance of solar off-grid PV, either SHS or mini-grid or hybrid systems is available in various journals, technical reports and conference proceedings. This is due to the growing share and relevance of PV as a future clean and sustainable energy system throughout the world. However, their performance in different climates or geographical locations over extended period of time is not completely established (Guerra *et al.*, 2017). The data collected during the operation of PV systems are of great interest in determining the system performance, which will help to detect malfunctions and to optimize the system. For instance, field studies have reported that one PV system was oversized by more than 40%, a rural electrification project in Laos reported 65% of solar home systems were not operating, and another project at Guatemala reported that 45% of solar home systems ceased to operate after five years of installation (Jiménez-Castillo *et al.*, 2019).

The installation of a solar PV power system to either replace or offset a portion of electricity generated from DGs has been cited as an option to consider for remote residential homes in Nigeria (Ani, 2016).However, this author observed that complete replacement of DGs with PV systems may not be feasible during rainy season, and recommended the use of PV/diesel hybrid systems. It is noted that combined solar/diesel system is more reliable and cost effective, but it must be sized optimally. The DG component is operated only during low irradiance wet seasons, and hence reduces fuel consumption, which in turn reduce fuel cost and CO<sub>2</sub> emission. The merit of a PV power backup system is that it will accord the institutions or individuals an opportunity to contribute towards the global quest on low-carbon and sustainable sources of energy. Additionally, unlike a DG that is operated only during power outages, a PV system has the potential of producing electricity continuously, whether grid is on or off. Therefore, it can free up the limited utility electricity, and hence assist in stabilizing up the grid, contribute to energy diversity and security with little burden on the government.

#### **1.3** Progress in solar PV Technology

The development in solar PV technology has been growing very fast in recent years due to technological improvement, cost reductions in materials and government support for renewable energy-based electricity production. The cost of PV systems has been declining steadily and significantly over the past recent decades, and this cost decline has been attributed to intense research, technological development and market growth spurred by government subsidies (Lang *et al.*, 2016). The efficiency of solar cell is one of the important driving parameters which are required to establish

and make this technology competitive in the market. The decreasing module costs combined with increasing efficiencies have resulted in a compound decrease in the cost of electricity from PV modules (Fukushima Renewable Energy Institute, 2014; Shubbak, 2019). In addition, the robust growth is backed up by continuous reduction in cost per watt where currently on average the cost is below a dollar per watt in the United States (US) and for a PV system it is slightly above a dollar per watt (Fu *et al.*, 2020). The growth is further attributed to the realization of levelized costs of electricity (LCOE) that are now generally very low compared to other energy sources (Jordan and Kurtz, 2013).

According to International Energy Agency (IEA) report of 2019, PVs have been gaining an increasing market share over the last few decades and is one of the fastest growing renewable energy technologies expected to play a major role in the future global electricity generation mix. As a result, global PV capacity is projected to increase from 700 GW to 880 GW in a span of five years from 2020 to 2025 (IEA, 2019). Figure 2.1 shows the trend of annual PV installations in leading world markets from the year 2000 to 2020, indicating clearly the rapid growth in the last 2010 to 2020 decade. The installed PV systems in this market increased by 156 TWh or a 23% growth in 2020 as compared to 2019. This placed the PV sector in the second position behind wind and ahead of hydropower in the annual growth of electricity generation from renewable energy sources. The growth of solar PV in major world markets from 2019 to 2020 includes 45% in United States, 75% in China, and 15% in European Union. In the emerging world markets, Brazil continued to lead Latin America's solar PV capacity where annual installed capacity increased by 58% to reach 3 GW in 2020 as compared to 2019. However, despite the increasing generation of electricity from



#### Figure 1. 1: Annual PV installations globally (Source: IEA, 2019).

Kenya has a huge potential for PV since it is located astride the equator, hence endowed with a high level of solar insolation with large part of the country receiving average insolation ranging from1900 kWh/m<sup>2</sup> to 2500 kWh/m<sup>2</sup> (Oloo *et al.*, 2015; Kiplagat *et al.*, 2011; SolarGIS, 2013; Jacobson, 2007; Rabah, 2005). Due to the huge abundance of solar irradiance in Kenya, PV systems offer a cleaner and viable solution for both primary (sole supply) and secondary (backup system) electricity generation (Jacobson, 2007; Rabah, 2005; Mohandes *et al.*, 2019).The most common PV systems in the country are the solar home system (SHS) and constitute the largest percentage of the installed capacity (Simuyu *et al.*, 2014).Currently, the country is considered as among the most active PV market in the African continent, with an estimated 250,000 rural households using SHS and annual PV sales estimated to be between 25,000 to 30,000 PV systems. With a recent major government decision to reinstate the VAT exemptions will support the increased uptake of the solar products and significantly contribute towards the achievement of universal access to electricity in the country. PV is projected to grow at 15% annually in Kenya, but this is still marginal compared to its potentials, (ERC, 2019).

#### **1.4** Statement of the problem

Accessibility to reliable and efficient supply of electricity is regarded as a key conduit for economic growth and development across the world. However, majority of African countries, Kenya included, experience numerous power outages. The costs of a power outage to a business can be substantial, including product losses, revenue, productivity, customers and even loss of life (in case of medical facilities). With increasing severe weather events and disasters triggering greater numbers of costly power outages, there is a growing interest in independent generators that can provide reliable electricity whenever power outages occur. The widely used power backup systems are the conventional DG power systems that can reliably bridge the gap during power blackout, but are associated with pollution effects through CO<sub>2</sub> emission due to fossil fuel used and noise as well as high fuel cost. These adverse effects of diesel backup systems prompt for gradual replacement of these fossil powered systems with clean alternatives that conform to SDG 7 and 13 targets. The current pursuit of low-carbon and sustainable sources of energy makes PV power backup systems ideal substitute for a diesel backup system because they are clean, low noise pollution and reliable. However, due to intermittent and irregular nature of PV power generation, there is a need for monitoring PV power output, and generated data on its performance has a wealth of information for diverse stakeholders (consumers, vendors, manufacturers, governments, etc). Most PV systems are installed in the built environment, most notably on the rooftops of residential, commercial, and industrial

buildings. Thus, in many cases, module orientation (i.e., tilt and azimuth) of these PV systems are not optimized, because they are installed at heterogeneous tilt and azimuth angles, often determined by roof characteristics, especially for pitched roofs. Consequently, they are not operating optimally as it should be, and hence there is a need to perform assessment of technical, economic and environmental performance of PV systems to showcase its viability as power backup over DGs, and hence enable stakeholders make informed decision on its adaptability. The technical, economic and environment analyses are done on an already installed solar PV system at Kaiboi Technical Training Institute (KTTI) in western part of Kenya as a case study for possible substitute of the DG backup systems.

#### 1.5 Research Objectives

#### 1.5.1 General objective

To carry out technical, environmental and economic analysis of an installed off-grid PV system at KTTI, in western region of Kenya as a possible substitute for diesel backup generator.

#### **1.5.2** Specific objectives

The specific objectives of this research are:

- To determine the solar irradiance model(s) that best fit the solar radiation data collected at the site.
- 2. To evaluate in-situ outdoor electrical performance of the PV power system, both experimentally and theoretically.

- 3. To perform comparative economic analysis of electricity generation from the solar PV system relative to that of a similar diesel backup generator.
- To calculate the amount of the CO<sub>2</sub> emission avoided or mitigated when solar PV power system replaces a similar diesel backup generator.

#### **1.6** Justification of the study

Many countries throughout the world are increasingly incorporating locally available renewable energy (RE) into their energy supply mix. This trend is informed mainly by the perceived scarcity of fossil fuels and increasing restriction on their use as a result of their adverse effects on climate change. Kenya is endowed with high solar irradiance (average of 5 kWh/m<sup>2</sup>/day) (Oloo et al., 2015; Kariuki and Sato, 2018) throughout the year, hence huge potential which can be exploited to generate green electricity for every household throughout the country. The PV technology allows electricity to be obtained from solar radiation with minimal or no impact to the human-beings and environment, unlike the diesel which has health and environmental hazards. Deployment of PV backup systems has the potential to supply a significant share of the energy demand and to make a positive impact on combating climate change and global warming. However, a major part of PV electricity production is characterized by a large degree of intermittency driven by the natural variability of climate factors such as air temperature, wind velocity, solar radiation, and precipitation. Consequently, it is beneficial to investigate the performance of any PV technology installed under the local weather conditions, because the generated data on energy yield can be consumed by different stakeholders for optimization of power flow management, evaluation of profitability by the private PV power producers and detection of system anomaly. In addition, PV power backup systems will enable the

enterprises and institutions to operate their business and operations smoothly with low operational costs, and also enable them to contribute to the increase use of green energy at individual level and hence synergize with the government in the achievement of SDG 7 and its ripple effect on the other interconnected SDGs.

#### **1.7** Significance of the study

The benefit of this study is that it will showcase to the decision makers in institutions or individual entrepreneurs and governments to make informed decision on affordably replacing the DG backup system with PV power backup system. The experimental data collected on solar irradiance and electrical performance of the PV system are important to solar energy community. The developers of simulation models will use radiation data to validate their models and electrical measurements will be useful for degradation analysis for the PV modules. Further, the economic and environmental analyses will guide the consumers on the initial/ running cost and environment impact respectively for each backup system.

#### **1.8** Scope and Limitations of the study

The findings from this study are based on the local outdoor climatic conditions measured at Kaiboi Technical Training Institute (KTTI) in western region of Kenya collected for the year 2020, which may and may not be replicated in subsequent years. Consequently, generalizing the results to other years or places may be subjected to variations.

#### **1.9** Thesis organization

The thesis is structured as follows:

- Chapter two presents background of PV theory, PV power system and literature review.
- Chapter three presents methodology and materials. It covers experimental setup, instrumentation and analytical techniques used to generate results.
- Chapter four presents the results and discussions, based on the four objectives.
- Chapter five presents the conclusions and recommendations.

#### **CHAPTER TWO**

#### THEORY AND LITERATURE REVIEW

#### 2.1 Introduction

This chapter presents a brief overview of electricity scenario in Kenya, solar radiation, photovoltaic theory and literature review. Photovoltaic theory focuses on available conversion technologies, PV cell structure and operations, applications and the available PV system configurations. Literature review on technical, economic and environmental performance analysis of PV systems are presented.

#### 2.2 Electricity in Kenya

Electricity generation in Kenya is largely dominated by renewable energy sources (RES) which comprise of hydro, geothermal, wind and solar resources. Figure 2.1 presents the installed proportion trend of electricity generated by source from 2016 to 2020 where their corresponding data is given in Appendix I which include their effective capacities (KPLC, 2020 and Economic survey, 2021).



Figure 2.1: The proportion trend of electricity generated by source from 2016 to 2020 (Source: Economic survey, 2021; KPLC, 2020)

As can be seen from Figure 2.1, the geothermal installed capacity of electricity remained at a constant capacity about 712.5 MW from 2016 to 2018 since there was no additional well. With an additional Olkaria V well in 2019, there was an increase in capacity by 4.2% to 863.1 MW from 2019 to 2020. In addition, geothermal effective capacity declined by 10.9 MW to 805.1 MW in 2020. In addition, the Kenyan Government through KenGen has four ongoing geothermal projects with a total estimated capacity of 314 MW. Similarly, wind installed capacity of electricity showed a sharp rise in 2018 to about 400 MW as a result of commissioning of new wind power plants e.g., the Lake Turkana wind power plant. Solar energy installed capacity of electricity has as well shown a slight rise from the year 2018 by 3.0% to the current 52.5 MW in 2020. This increase was as a result of commissioning of the Garissa solar power plant with a capacity of about 50.0 MW. This capacity is expected to increase in the near future with more power plants being developed e.g., the Gitaru solar power project with an estimated capacity of 42.5 MW (Economic survey, 2021).

Hydro capacity rose by 7.8 MW to 834.0 MW in 2020 mainly due to favourable rainfall experienced in the year. Its capacity over the previous years has shown slight or no change since there has been any installations of new hydro power plants. Hydropower in Kenya is mainly in the form of dammed hydro power plants with production susceptible to drought. On the other hand, thermal capacity decreased slightly to 749.1 MW in 2020 and has been on downward trend since 2018 in an effort to scale up access to clean energy by replacing it with solar and wind energy sources hence saving the country from heavy expenditure on petroleum imports. Generation of electricity from thermal sources, which is the only non- renewable source in the country accounted for 26.4% of the total electricity generation during the review

period where the rest of electricity was generated from renewable sources which accounted for a total 73.6%. Similarly, Co-generation installed capacity dropped from 28.0 MW in 2019 to 2.0 MW in 2020 due to decommissioning of Mumias Sugar power plant in February 2020 following expiry of its Power Purchase Agreement (PPA) (Economic survey, 2021).

The total installed capacity of electricity in the country increased slightly to 2,836.7 MW in 2020 from 2,818.9 MW in 2019. In addition, the total maximum theoretical electric output power has seen a huge increase of 509.7 MW from the year 2016 to 2020. However, effective electricity generation capacity dropped by 1.1 per cent to 2,705.3 MW in 2020. Total electricity generation declined marginally from 11,620.7 GWh in 2019 to 11,603.6 GWh in 2020 (Economic survey, 2021).

Energy demand in Kenya is on the rise due to rapid increase in population as well as growth in the economy and more affordable electronic devises, though a decline was registered in the year 2020 where total domestic demand for electricity reduced by 57.6 GWh to 8,796.4 GWh (KNBS, 2020;Economic survey, 2021;KPLC, 2021). In addition, demand for electricity registered an increase in all categories except large and medium (commercial and industrial) category whose demand reduced by 3.6% to 4,281.0 GWh in 2020. Demand for street lighting and rural electrification categories increased by 16.7% and 7.5% to 74.5 GWh and 611.9 GWh, respectively during the same period. Similarly, demand for domestic and small commercial category increased by 49.0 GWh to 3,829.1 GWh. This was attributed to, possibly, the working from home measures instituted to counter the spread of COVID-19. Large and medium (commercial and industrial) category accounted for 48.7% of the total domestic demand for electricity in 2020, while domestic and small commercial

category accounted for 43.5% (Economic survey, 2021; Energy Regulatory Commission, 2020).

Available data shows that the cost of electricity in Kenya is relatively high compared to other countries which result to high cost of locally produced goods (KAM, 2019). The high cost of electricity is attributed to power production plants which operate under low efficiencies as reported by Njeru *et al.*,(2020). For instance, the thermal power plants average efficiency was found to be 70.62% for both the grid connected and the isolated power plants. The mean price of electricity which is inclusive of power production costs, distribution and taxes for the past few years in Kenya is 0.211\$/kWh for households and 0.169 \$/kWh for business (Economic survey, 2021). Comparatively, in the neighboring countries the household price of electricity in Tanzania is 0.099 \$/kWh, in Uganda is 0.189 \$/kWh and in Ethiopia is 0.007 \$/kWh. Globally, the average price for electricity is 0.136 \$/kWh for households and 0.123 \$/kWh for businesses respectively (Grant *et al.*, 2020).

During the same review period 2016 to 2020, transmission and distributive losses totaled to 2,790.7 GWh, accounting for 24.3% of the total domestic generation in 2020 (KNBS, 2021; Economic survey, 2021). These losses coupled with frequent power outages act as a decelerating factor on economic growth (World Bank, 2019). In addition, according to Cissokho and Seck, (2013) and George *et al.*, (2019) show that interruption in electricity supply could have serious consequences on consumers and losses which have led to many Kenyan companies lose nearly 10% of their production and on average; firms reported losing 8.6% of the total annual sales due to power outages (Enterprise Surveys, 2016). Power outages in Kenya are partly blamed on poor infrastructural facilities, fraud and vandalism of infrastructures such as

transformers and power cables and in extension corruption, politics and poor governance in the power institutions (Pless and Fell 2017; Maende and Alwanga 2020). These power outages experienced in the country according to the World Bank (2013) occur 6.3 times in a month and can last an average of 5 hours which translate to an average of 1.5 hours to 2 hours in a day across the year. During this period of power outages, many businesses, institutions and households look for an alternative source of power as backup solution to these interruptions.

Lack of reliable electricity in Kenya has prompted the need for self-generation and consumption of electricity (Jiménez-Castillo et al., 2019). During power blackouts, DGs, with typical power ratings ranging from  $\sim 0.1$  kW to 100 kW, are usually used by many businesses and institutions to power their operations. The DGs are preferred power backup systems because of their availability, low initial and operational knowhow by many users (Research and Markets, 2017; Energy Regulatory Commission, 2018). In addition, the recent push to expand rural grid coverage nationwide has resulted in higher levels of electricity access in Kenya (Lee et al., 2014). These authors collected data on spatial and economic factors to determine local electrification rates in the densely populated counties of Western Kenya, and their findings were interesting. They found, to their surprise, that even in an ideal setting, where there is high population density and extensive grid coverage, electrification rates remain very low, averaging 5% for rural households and 22% for rural businesses. They also observed that the pattern holds across time and applies to both poor and relatively well-off households and businesses. An interesting finding from this study is that half of the unconnected households are "under grid," or clustered within just 200 m of a low-voltage power line, and that the same results may apply across many countries in SSA. Thus, the assumption that majority of the un-electrified
are 'off grid' is not true and hence this signal for alternative methods for electrification of rural households and enterprises in SSA. In her efforts to achieve universal electrification for all by 2030, the Kenya's government has rolled out an ambitious flagship rural electrification project called 'Kenya Off-Grid Solar Access Project (KOSAP)', and is tasked to increase electricity access using PV systems in 14 underserved counties (KPLC, 2012).

# 2.3 Solar Radiation

The energy emitted by the sun is called solar radiation and is the earth's primary natural source of energy, and has been exploited since antiquity to date. Solar radiation is a term used to describe ultraviolet, visible and infrared regions of the electromagnetic spectrum where their wavelengths range from 0.25  $\mu$ m to 4.5  $\mu$ m (Goswami *et al.*, 2000). Figure 2.3 shows the spectral distribution of terrestrial and extraterrestrial solar radiation. Extraterrestrial and terrestrial radiation falls between 0.15  $\mu$ m and 1.20  $\mu$ m and the radiation of practical importance to the solar energy users falls between 0.15  $\mu$ m and 2.50  $\mu$ m, which include infrared used in thermal collectors. Visible radiation wavelengths lie between 0.40  $\mu$ m and 0.80  $\mu$ m. The power density emitted by the sun is of the order of 64 MW/m<sup>2</sup> of which ~1370 W/m<sup>2</sup> also known as the solar constant reaches the top of the Earth's atmosphere with no significant absorption in the space (Duffie and Beckman, 2013).



Figure 2.2: Spectral distribution of extraterrestrial radiation. Source: Renewable resource data center(Duffie and Beckman, 2013)

The spectral and intensity distribution of solar radiation is altered as it passes through the atmosphere by absorption, reflection and scattering and the amount of energy absorbed depends on the length of the path in which the solar beam traverses (Shafey and Ismail, 1990). This has important implications in many natural and technological processes, including meteorology, biology, ecology, and energy production. The solar radiation received at ground level by a horizontal surface has a direct component and a diffuse component (Zhang *et al.*, 2014).In addition, for solar energy collectors, there is also a third component called albedo, which is solar radiation reflected to the surface of the collector from the ground/snow, buildings and trees(Rao *et al.*, 1984). The sum of the direct and diffuse components on a horizontal surface is usually called global horizontal radiation (GHI) (Karuppu and Sitarama, 2019; Duffie and Beckman, 2013). Solar harvesting devices are usually mounted on inclined building roofs or inclined on flat roofs; hence the radiation intercepted is different from the GHI. In such cases, the plane-of-array (POA) solar radiation is used instead, and consists of the three components mentioned above. Solar radiation data bases provide the GHI and simulation models have been developed for estimating solar radiation in sites where there are no measurements for both GHI and POA data (Hofmann and Seckmeyer, 2017).

The earth receives about 174 PW which is approximately 70% of solar insolation at the upper atmosphere where some of the incoming radiation is absorbed in the atmosphere by vapor, dust, ozone and some are reflected back into space mostly by clouds (Duffie and Beckman, 2013). Available data show that most of the Earth land mass receive insolation levels of about 3.5 KWh/m<sup>2</sup>/day to 7.0 KWh/m<sup>2</sup>/day which is substantial and supplies about 0.1% of the world energy consumption (Karuppu and Sitarama, 2019; Hofmann and Seckmeyer, 2017; Duffie and Beckman, 2013). It is estimated that Kenya on average receives solar insolation of about 4.0 KWh/m<sup>2</sup>/day to 6.0 KWh/m<sup>2</sup>/day (Oloo *et al.*, 2015; Kiplagat *et al.*, 2011; Rose *et al.*, 2016). Insolation received at a given location depends upon the relative position between the sun and the earth, time of the day, seasons of the year, local weather and geographical location (Castaner, 2002; Bosch *et al.*, 2010; Museruka and Mutabazi, 2007).

Availability of solar radiation data are beneficial in areas of agriculture, water resources, day lighting and architectural design, solar conversion devices and climate change studies (Katiyar *et al.*, 2010; Al-Kayiem and Mohammad, 2019). In solar energy conversion and utilization devices, data on solar radiation and its components at a given location are very essential for their evaluation and deployment. These data are required in order to design and size a cost-effective solar collector in terms of cost and energy demand of the load at the site. Daily global solar radiation reaching the earth surface at hourly basis (or even sub-hourly) is very important for analysis of

performance of solar energy conversion and utilization devices (e.g., PV module), but monthly averaged solar radiation data may be sufficient for agricultural applications (Al-Sanea *et al.*, 2004). Solar radiation data can be obtained from several sources including ground observation, satellite remote sensing, interpolation, and empirical models (Huang *et al.*, 2019; Dahmani *et al.*, 2016; Myers *et al.*, 2002). Each method has its merits and deficiencies. The radiation data are specified in units of Wm<sup>-2</sup> or kWm<sup>-2</sup> and when integrated over a day, give daily irradiation which is expressed in MJm<sup>-2</sup>day<sup>-1</sup> or kWhm<sup>-2</sup>day<sup>-1</sup>. In addition, solar radiation varies both spatially and temporarily. Thus, for a given site and regardless of type, solar radiation can be quantified with respect to time as hourly, daily, monthly or annual averages (Cucumo *et al.*, 2007).

## 2.3.1 Ground Observation

Ground observation is the most direct and reliable way to obtain solar radiation at the site of interest, and high-quality instrumentation and maintenance often provides baseline solar radiation data. These stations have been in operation as early as 1964, and in particular, the World Meteorological Organization (WMO) had set up the World Radiation Data Centre to collect, archive and publish global measurements of GHI, and other components (Huang *et al.*, 2019).However, these ground stations are few and sparsely distributed across the world due to the high cost of measuring instruments (Sengupta *et al.*, 2015; Myers *et al.*, 2002) and is its weakest point.

Pyranometers (e.g., Kipp and Zonen) are usually used to measure the GHI, and are mounted on flat surfaces (Shenoy *et al.*, 2018). The GHI component may vary greatly depending on the surface elevation, location, time of day, seasons and on other meteorological factors (Awasthi and Poudyal, 2018). For solar energy conversion

devices, the pyranometers are also used to measure POA radiation by mounting them on the surface of the collector (Oyelami et al., 2020). The pyranometers come in various models based on type of sensor used, which can be based on either heating effects (thermopiles) or electric effects (PV cell) to generate electric voltage that can detected and measured. Modern digital pyranometer use pyroelectric elements that generate voltage when a change in temperature is detected. Each pyranometer is usually calibrated by the manufacturers, but need to be calibrated at regular intervals to ensure accurate measurements (Tohsing et al., 2019). Another common instrument used for solar radiation measurements is the pyrheliometer, which is used to measure direct or beam solar radiation only (Garg, 1993). A pyrheliometer has a narrow aperture to ensure that only solar radiation from the sun is measured, usually has a full angle of about 0.53° (Frohlich et al., 1973; Duffie and Beckman, 2013). The pyrheliometer needs to be pointed to the sun and is normally used for calibration. The portable pyranometer and pyrheliometer are usually used in conjunction with voltmeter to read the measurements or with data loggers for continuous recording and storage (Jacovides et al., 2006).

#### 2.3.2 Interpolation method

Interpolation method is used to estimate solar radiation in a location where there is no measured data, and is used in conjunction with ground measured data or global radiation maps. This method is used to estimate radiation values at a location between two ground measuring stations and usually uses interpolation with respect to spatial dimensions between the observation stations and area of interest (Longley *et al.*, 2001). However, spatial interpolation is restricted to a radius of 30 km from the station (Al-Sanea *et al.*, 2004 and Ascencio-V´asquez *et al.*, 2020).Interpolation

techniques that are usually used include linear regression; nearest neighbor; inverse distance weighting, among others (Pessanha *et al.*, 2021).

## 2.3.3 Satellite Remote Sensing

Satellite remote sensing has been recognized as a method that offers a unique means to monitor and estimate solar radiation regionally or globally and is better than interpolation and numerical modeling methods. In particular, satellite data are preferable than those of interpolation method, and is usually recommended to be used for locations that further than 30 Km from ground observation stations. The satellite derived data have accuracies that approach those of ground observations(Huang *et al.*, 2019). In this method, radiometers mounted on the satellites are used to measure the radiation in space and relay the data to ground stations where they are recorded and archived (Ruiz-Arias *et al.*, 2010).Satellite measurements provide continuous data for all regions of the earth over many years with almost uniform resolution (Cebecauer *et al.*, 2010).

# 2.3.4 Empirical methods

Solar radiation data in any location on the earth surface can also be generated using numerical or empirical or theoretical models that have been developed over the years. The empirical method is used in cases where measured or satellite data are inaccessible and is usually used in simulation models that are usually utilized in the design and analysis of solar energy conversion devices. Several models with varying degrees of complexity, detail and accuracy have been developed and used in different parts of the world to generate radiation data and its components. These models are based on either empirical correlations or statistical regression between satellite and or ground measurements and various meteorological parameters (Doorga *et al.*, 2019).

The most commonly used meteorological parameters are air temperature (Gairaa and Bakelli, 2013; Bocca *et al.*, 2018), relative humidity (Rao *et al.*, 2012; Qing andNiu, 2018), precipitation (Yu *et al.*, 2019; Maleki *et al.*, 2017), cloudiness cover (Paulescu and Blaga, 2016) and bright sunshine duration (Muneer, 2014). Additional input parameters used are solar altitude, aerosol concentrations and global warming factor (Hocaoglu and Serttas, 2017; Despotovic *et al.*, 2016).The correlation coefficients in each model are site dependent, and hence are adjusted for each location when using them. In this study, four models for predicting GHI were selected and validated with the ground measured data at a site of study to select the most appropriate. These models are Angstrom-Prescott, ASHRAE, Hargreaves-Samanni and Iqbal, which are based respectively on sunshine hours, sky clearness, ambient temperature and the atmospheric transmittance. Each model has got unique merits and demerits depending on the number and availability or ease of determination of predictor variable(s) (Doorga *et al.*, 2009).

Radiation prediction models are expected to estimate as closely as possible the monthly average daily global solar irradiation on a horizontal surface from applicable meteorological parameters (Ulgen and Hepbasli, 2002). The four selected models were chosen because they are diverse and involve input parameters that are easily available or determined at the site. The models considered can be grouped into three categories: sunshine, temperature and hybrid-parameter based models (Hofmann and Seckmeyer, 2017). The Angstrom-Prescott model was developed by Angstrom but was modified by Prescott, hence the name. Prescott simplified the equation by replacing clear sky global solar irradiation with extraterrestrial solar irradiation on a horizontal surface. The Angstrom-Prescott model assumed linear relationship between sunshine duration and solar radiation, but various researchers have reported better

coefficients for second order (quadratic), third order (cubic), exponential, power and logarithmic fits (Doorga *et al.*, 2009).Hargreaves-Samanni's model is based on air temperature (maximum and minimum) and most appropriate when data on sunshine hours is lacking. ASHRAE clear sky model offers a simple method and is widely used as a basic tool for solar heat load calculation for air conditioning systems and building designs by engineers and architects (Maleki *et al.*, 2017). Iqbal model takes into account the scattering-transmittance of the atmosphere contributed by ozone, gases, water and aerosol components (Wong and Chow, 2001). In this study, these selected models will be validated by the solar radiation data measured in the study area.

#### a) Angstrom-Prescott model

This model correlates the daily extraterrestrial,  $H_o$  and terrestrial,  $H_s$  solar radiation and sunshine duration as having linear relationship (Duffie and Beckmann, 2013; Ångström, 1924; Da-Silva *et al.*, 2017; Tymvios *et al.*, 2005):

$$H_{s} = H_{o} \left[ a + b \left( \frac{S}{S_{o}} \right) \right]$$
(2.1)

where *S* is daily number of hours of bright sunshine,  $S_0$  is daily number of hours of possible sunshine (daylight between sunrise and sunset), *a* and *b* values are regression coefficients to be determined for the site (Duffie and Beckmann, 2013; Sabzipavar *et al.*, 2013). The  $H_0$  can be calculated from the equation:

$$H_o = \frac{24}{\pi} G_{sc} \left[ 1 + 0.33 \cos\left(\frac{360n}{365}\right) \right] \left[ \cos \delta \cos \varphi \sin \omega_s + \frac{\pi \omega_s}{180} \sin \varphi \sin \delta \right] (2.2)$$

where  $G_{sc}$  is solar constant,  $\omega_s$  is the sunset angle,  $\varphi$  is the latitude angle of the site,  $\delta$  is the sun declination angle and *n* is the day number of the year, with *n*=1 for 1<sup>st</sup>January and *n*= 365 for 31<sup>st</sup> December (Bandyopadhyay *et al.*, 2008). The

coefficients a and b in equation (2.2) is dependent on the site altitude and atmospheric transmittance and are given by Duffie and Beckman, (2013) as;

$$a = 0.110 + 0.235\cos\varphi + 0.323\left(\frac{S}{S_o}\right)$$
(2.3)

$$b = 1.449 + 0.553\cos\varphi - 0.694\left(\frac{S}{S_o}\right)$$
(2.4)

#### b) ASHRAE model

ASHRAE clear sky model is given by (ASHRAE, 1985; Wong & Chow, 2001; Al-Sanea *et al.*, 2004; Perez *et al.*, 1992; Jamil and Khan, 2014):

$$H_s = G_{sc} e^{\left(-A/\cos\phi_z\right)} \left(\cos\phi_z + B\right)$$
(2.5)

where *A* is the atmospheric extinction coefficient, *B* is the diffuse sky factor, and  $\phi_z$  is the zenith angle. The coefficient*A* and Cos  $\phi_z$  can be calculated respectively from (Basharat and Mohd, 2014; Perez *et al.*, 1992; Bakirci, 2009):

$$A = 360(n-81)/365 \tag{2.6}$$

$$\cos\phi_Z = \cos\delta\cos\varphi\cos w_s + \sin\delta\sin\varphi \tag{2.7}$$

#### c) Hargreaves-Samanni's model

Hargreaves-Samanni'stemperature-based model is given by (Hargreaves and Samani, 1982; Gavalian *et al.*, 2005):

$$H_{s} = K_{RS} \left( \sqrt{T_{\text{max}} - T_{\text{min}}} \right) H_{o}$$
(2.8)

where  $K_{RS}$  is an empirical coefficient related to atmospheric transmittance, and  $T_{max}$ and  $T_{min}$  are respectively maximum and minimum temperatures. The  $K_{RS}$  is included to account for possible pollution at the site with values of 0.16 (used in this study) and 0.19 respectively for inland and coastal sites. The  $H_o$  is given by (Despotovic *et al.*, 2016):

$$H_o = \frac{1440}{\pi} G_{sc} E_o \left( \omega_s \sin \varphi \sin \delta + \cos \delta \cos \varphi \sin \omega_s \right)$$
(2.9)

where  $E_{0}$  is the eccentricity correction factor of the earth's orbit and is expressed as (Wong & Chow, 2001):

$$E_0 = 1.0 + 0.033 \cos 2\pi \left(\frac{n-1}{365}\right) \tag{2.10}$$

#### d) Iqbal model

Iqbal model is based on atmospheric transmittance and is given by (Iqbal, 1983; Batlles *et al.*, 2000; Wong and Chow, 2001):

$$H = \left(0.9751G_{sc}E_{o}\tau_{s}\cos\theta_{z} + H_{d}\right)\left(\frac{1}{1-\rho_{g}\rho_{a}}\right)$$
(2.11)

where the factor 0.9751 is included to show that spectral interval considered is 0.3-3  $\mu$ m of the electromagnetic spectrum, $\tau_s$  is the scattering-transmittance (equal to product of ozone, gas, water and aerosol scattering fractions), and  $H_d$  is diffuse component contributed by Rayleigh and aerosols scattering after passing through the atmosphere. The parameters  $\rho_g$  and  $\rho_a$  are the ground albedo respectively. Evaluation of  $\tau_s$ ,  $H_d$ ,  $\rho_a$  and other parameters are given explicitly by Wong and Chow (2001); Batlles *et al.*, (2000) and Despotovic *et al.*, (2016).

#### e) Sensitivity analysis

Several statistical indicators have been proposed and used to test the accuracy of the solar radiation models. Among them are mean bias error (MBE), root mean square

error (RMSE), absolute percent error (MAPE), coefficient of determination ( $R^2$ ), *t*-statistic method (*t*-stat), among others (Toffalis, 2015; Al-Aboosi, 2020; Doorga, *et al.*, 2019).

The RMSE represents the sample standard deviation of the differences between estimated and measured values and is given as:

$$RMSE = \sqrt{\frac{1}{n} \sum_{i=1}^{n} (H_{i,r} - H_{i,e})^2}$$
(2.12)

The MAPE is a measure of relative overall fit, and has the advantage of being scaleindependent (Doorga *et al.*, 2019):

$$MAPE = \frac{100}{n} \sqrt{\sum_{i=1}^{n} \left(\frac{H_{i,r} - H_{i,e}}{H_{i,r}}\right)}$$
(2.13)

where  $H_{i,r}$  and  $H_{i,e}$  represent the *i*<sup>th</sup> measured and estimated values respectively, and *n* is the number of observations.

### 2.3.5 Clearness index

Clearness index,  $K_T$  is the ratio of the measured solar irradiation in a given location relative to the calculated extraterrestrial irradiation (Woyte *et al.*, 2007). It is a dimensionless quantity and can vary from 0 to 1 and can be determined on hourly, daily or monthly basis (Mellit *et al.*, 2008). The monthly average clearness index is the ratio of monthly average measured daily radiation on a horizontal surface (H) (W/m<sup>2</sup>/day) to monthly average daily extraterrestrial radiation on a horizontal surface ( $H_o$ ) (W/m<sup>2</sup>/day) expressed as (Duffie and Beckman, 2013; Bandyopadhyay *et al.*, 2008):

$$K_T = \frac{H}{H_o} \tag{2.14}$$

where  $H_o$  is the extraterrestrial solar radiation as expressed in equation (2.2).

Clearness index is useful in characterization of solar energy resource for effective sizing of energy conversion systems and for a predictive purpose (Li and Lam, 2000). Using clearness index, segmentation of a given site is characterized by three types of clusters, namely cloudy (KT < 0.3), partly cloudy ( $0.3 \le KT \le 0.5$ ) and clear sky/sunny (KT > 0.5) (Harrouni *et al.*, 2005; Al Aboosi, 2020; Lee *et al.*, 2015).

## 2.4 Solar energy conversion technologies

The available technologies through which solar energy is harnessed presently include the photovoltaics (PV), concentrating solar power systems (CSP), solar domestic hot water heating system (SDHW), solar heating and cooling (SHC) systems and Photovoltaic/ Thermal (PV/T) systems.

The solar heating and cooling (SHC) systems absorb solar radiation and convert to heat energy through heat transfer fluid such as water, air or oil (Faninger, 2010). They are utilized in various applications in commercial and industrial applications. They comprise of solar thermal collectors, heat exchange system, a thermal storage system, and a cooling tower (Luzzi and Lovegrove, 2004). The CSP system also known as solar thermal electricity (STE) uses heat from the sun to run a steam engine, which turn turbine to generate electricity (Collado and Guallar, 2013). They use mirrors to concentrate energy from the sun in order to heat water to steam. The SDHW system uses solar energy for heating where a black absorbing surface absorbs solar radiations and transfers the heat energy to water flowing through it. They consist of solar collector and an insulated storage tank to store hot water to be used latter anytime (Burzynski *et al.*, 2010). Recent development in solar energy utilization is the combination of PV and solar thermal collector to yield hybrid PV/T solar system

where heat and electricity are generated simultaneously (Ali *et al.*, 2017, Tonui and Tripanagnostopoulos, 2008, 2007). These systems transfer the otherwise unused waste heat from the PV modules to heat transfer fluid which then channel for useful applications and at the same time cools the cells improving electrical conversion efficiency. The PV/T system attains higher overall conversion efficiency compared to when they operate independently (Kalgirou and Tripanagnostopoulos, 2007).

The PV convert solar radiations directly to electricity through a process known as photoelectric effect, a phenomenon that was first discovered experimentally in 1839 by Bequerel (1839). During his experiments involving an electrochemical cell, he noted that the voltage of the cell increased when its silver plates were exposed to sunlight. It is through this process that PV cells generate voltage and electric current upon exposure to light. The solar PV cells produce direct current (DC) power. Upon this discovery, it opened up a modern application of solar energy for electricity generation and spurred great interest in research and technological development in solar energy field. An explosion of research and development in solar photovoltaics have been witnessed since the oil crisis of the 70s to date when countries who do not produce oil was awakened to the seek alternative energy sources to mitigate against oil crisis in the future. The PV industry has been striving throughout the world in applications and market penetration, which has been attributed to advances in technology (increasing efficiency and reliability), government subsidies (policies), mass production, and the need to switch to low carbon energy sources (Mittag et al., 2019).

Figure 2.3 shows the structure of a PV cell. As Illustrated in the figure, the PV cell consists of two layers, the p-type semiconductor and the n-type semiconductor that are joined together to form a p-n junction diode (Luceño-Sánchez *et al.*, 2019). The n-layer has high electron concentration while the p-layer has high hole concentration, and when they are joined, electrons (holes) diffuse from n-layer (p-layer) to p-layer (n-layer)and fill the holes (electrons) adjacent to the junction and create negatively (positively) charge ions. Diffusion of free charge carriers continue until equilibrium is reached, when the negative (positive) ions on p-layer (n-layer) stop further diffusion of electron (holes). Thus, the region adjacent to the junction is depleted of free carriers and is usually called depletion layer, and no free carriers (electrons or holes) can cross it.

A solar cell is a large area p-n junction diode so as to capture sufficient solar radiation. It is structured such the p-layer forms the top or front layer and is intentionally made very thin as compared to back n-layer in order to allow solar radiation to penetrate into the p-n junction. When light is incident on the p-n junction through the front thin p-layer, light photons absorbed supply sufficient energy to ions in the junction to create a number of electron-hole pairs. If an external circuit is connected across the cell, then the photon generated charge carriers will be separated, and made to flow in opposite directions creating an external current in the process. The amount of current depends on the number of charges created, which in turn depends on the amount of solar radiation incident on the cell (Kumaresh *et al.*, 2014; Boyle, 2004). Typical shapes of solar cells are either square or circular, though other shapes are also possible, with typical dimension of 10 cm x 10 cm and thickness of

about 0.3 mm(Archer and Hill, 2001).Solar cells are also provided with metallic contacts attached to the two layers, which are used to collect the photon generated charge carriers and pass them to the external circuit. The front contact collects electrons, hence forms the negative terminal, while the back contact collects the holes and the positive terminal (Luceño-Sánchez *et al.*, 2019).



Figure 2.3: Structure of a solar PV cell. (Luceño-Sánchez et al., 2019)

Front contacts are plated into grooves in the cell so as to minimize shading of the PV cell surface. The back contact for modules is patterned in order to separate one cell from the next. The characteristics of the material to be used in making of the front and back contacts are those that are good conductors, comparatively cheap, temperature stable and with a smooth surface. The front and back contacts have great influence on efficiency as well as performance of the PV cell (Rached and Mostefaoui, 2008). The entire assembly of the PV cell is encapsulated between thin glass to provide mechanical stability and environmental protection (Hossain *et al.*, 2019).

The art, design and development of solar cells are categorized into three generations, and each generation aims to increase efficiency and reduce cost as compared to the previous one. Figure 2.4 shows classification of solar PV technologies based on the generation.



Figure 2.4: Schematic showing the classification of solar PV cells

First generation is based on cells from silicon and is categorized into two subgroups namely the single/ mono-crystalline silicon solar cells and poly/multi-crystalline silicon solar cells (Sampaio and González, 2017). They are solar cells produced from silicon wafers, hence referred to as wafer-based cells and the oldest and most established technology due to high conversion efficiencies. Their popularity is also attributed to easy ways of making an insulating layer on top of silicon by exposing it to oxygen while heating it to form SiO<sub>2</sub> layer which is a strong insulating material (Esmaili and Nasiri, 2009).

Second generation is more economical as compared to the first generation silicon wafer solar cells and they comprise of a-Si thin film solar cells, Cadmium Telluride (CdTe) thin film solar cell and Copper Indium Gallium Di-Selenide (CiGS) Solar cell which offer added advantages like environmental friendliness and lower costs(Choubey *et al.*, 2012; Gorter and Reinders, 2012 and Nikolaidou *et al.*, 2019).For a long period, silicon technology has been leading in manufacturing of PV

cells due to its availability but comes with high cost of production that has prompted researchers to come up with thin-film technology as a substitute since their layers are thinner, hence uses less material(McCann *et al.*, 2001). Thin-film technologies are maturing fast and soon may challenge the market share of crystalline Silicon devices though crystalline silicon devices are likely to remain dominant for the next decade because of their abundant availability as compared to thin-film technologies (Green and Emery, 2008; Shah *et al.*, 2014). Solar cells based on thin films constitute ~10% of global PV module market nowadays (Nikolaidou *et al.*, 2019). Generally, thin-film cells are made through fabrication processes, which may reduce manufacturing capital expense and material usage. This category extends from commercial technologies based on nano-structured materials. At the moment, three thin film PV technologies which are developed to commercial phase are hydrogenated amorphous silicon (a-Si:H), cadmium telluride (CdTe) and copper indium gallium diselenide (CuInxGa1–xSe2, or CIGS) (Ludin *et al.*, 2018).

The third generation comprises of nanostructures and organic materials which are relatively expensive but very efficient cells. These types include the Nano crystal-based solar cells, polymer-based solar cells, dye-sensitized solar cells and concentrated solar cells (Nikolaidou *et al.*, 2019; Habibi *et al.*, 2014). They are the new promising technologies but are not yet commercially deployed. This generation aim to develop devices with high efficiencies using the thin film method which may lead to an increment in the area cost, but the cost per watt peak would be reduced (Conibeer, 2007; Das *et al.*, 2014; Sethi *et al.*, 2011; Serrano *et al.*, 2009).

#### 2.6 The PV module and array

Figure 2.5 shows a structure of a PV module with 72 PV cells connected in series. A PV module is made of a number of PV cells connected in series or parallel to respectively increase current and produce a higher voltage. Most commercial modules contain 36 PV cells or 72 PV cells connected in series where the36 PV cells modules are commonly used for large power production in the PV industry. These number of cells account for any unexpected reduction in PV module voltage due to temperature, shading and other outdoor operating factors. The output voltage of a PV module is y chosen so as to be compatible with a 12V battery which may require voltage of about 15 V or more to charge. The current output from the PV module is determined by the solar irradiance and the number of PV cells while voltage is dependent on the size of the PV cells and cell temperature (Joseph and Kamala, 2013; El-Adawi and Al-Nuaim, 2007). A single crystal PV cell is often  $15.6 \times 15.6 \text{ cm}^2$  in size which gives a total current of almost 10 A from a module of 72 PV cells (Yang *et al.*, 2015). In addition, under optimum tilt conditions, the current density from a commercial solar cell is approximately between 30 mA/cm<sup>2</sup> to 36 mA/cm<sup>2</sup> (Green *et al.*, 2009).

Tempered glass or some other suitable transparent material encapsulates PV modules on the front surface while the back surface is covered with a thin polymer protective sheet (typically tedlar) which is a water proof material. These materials guarantee protection of the cells within the module from water, dust, mechanical compressions etc. To ensure that the module is well mounted, an aluminum frame holding everything together in a mountable unit is used in all the module edges for mechanical strength. At the back of the module there is a junction box, with wire leads that provide external power extraction line via the charge controller.



Figure 2.5: Structure of a PV module with 72 cells wired in series. (Source: Müller *et al.*, 2016)

A PV array is constructed from a number of individual PV modules that are connected together either in series or parallel or mixture of the two and usually mounted on the roof or on the ground field. For high power applications, this kind of connection is desirable so as to increase the output power to meet the load demand since a single PV module may not produce sufficient power. To increase the voltage, modules are connected in series and to increase current they are connected in parallel but under outdoor conditions, PV modules in an array should not be connected in series so as to obtain higher power output (Khatoon and Ibraheem, 2014). An entire surface of a PV array would not maintain a uniform irradiance level sometimes due to partial shading which causes a significant power loss (Micheli et al., 2014). To avoid these power losses, blocking diodes are used in series with the PV modules so as to prevent current from flowing back into them and also prevent the fully charged batteries from discharging or draining back through the PV array at night. In addition, bypass diodes are used in parallel with a PV module so as to provide an alternative path to current around the shaded module (Müller et al., 2016). Figure 2.6 shows a PV array consisting of four PV modules. As illustrated in the figure, the PV array produces two parallel branches in which there are two PV panels that are electrically connected together to produce a series circuit.



Figure 2.6: Basic structure of a PV array

## 2.7 PV performance

Characterization of the performance of PV cells/ modules/ arrays can be done either experimentally or theoretically. Experimental method involves setting up a PV circuit and measuring voltage, current, temperature and POA irradiance under different loading and plotting the I-V curve. From the curve, various characteristic parameters can be determined. On the other hand, theoretical analysis involves constituting equations from the equivalent circuit of a solar cell. The commonly used equivalent circuit of a solar cell is the single-diode model and PV module model (Bandou *et al.*, 2013; Ghani *et al.*, 2015; Akinyele *et al.*, 2015).

## 2.7.1 Single diode model

Figure 2.7 shows a single-diode equivalent circuit model of a solar cell, which consists of a series resistance  $(R_s)$ , a shunt resistance  $(R_{sh})$ , and a linear independent current source in parallel to a diode (Dadjé *et al.*, 2017). This model shows the parameters that are of interest such as the short-circuit current  $(I_{sc})$ , open-circuit voltage  $(V_{oc})$  and the PV maximum power  $(P_{max})$  that is expressed as the product of

maximum power point voltage ( $V_{mpp}$ ) and maximum power point current ( $I_{mpp}$ ) (Ghani et al., 2015; Tian et al., 2012 and Akarslan, 2012). The  $V_{oc}$  is voltage measured when the solar cell has no load; hence the current through the cell is zero. Essentially,  $V_{oc}$  is the maximum DC voltage while  $I_{sc}$  represents the maximum DC current on the I-V characteristics curve. The  $I_{mpp}$  and  $V_{mpp}$  are values of current and voltage when a PV cell or module is operating at maximum power point respectively (Tian et al., 2012).



Figure 2.7: Equivalent circuit of a PV cell (Dadjé et al., 2017)

From Figure 2.7, the PV cell output current, *I* could be estimated by applying the Kirchoff's current law as (Tian *et al.*, 2012; Ghani *et al.*, 2015):

$$I = I_l - I_d - I_{sh} \tag{2.15}$$

where *l* is the output current,  $I_l$  is the solar generated current,  $I_d$  is the diode current and  $I_{sh}$  is the shunt leakage current.

Employing the Shockley's ideal diode equation *I* is given as:

$$I = I_o \left( \exp\left(\frac{V + IR_s}{n_{id}V_t}\right) - 1 \right)$$
(2.16)

where

$$V_t = \frac{kT_c}{q} \tag{2.17}$$

and

$$I_{sh} = \frac{(V + IR_s)}{R_{sh}}$$
(2.18)

These equations (2.16), (2.17) and (2.18) are then substituted to equation (2.14) where the output current of the solar cell is then represented as (Bandou *et al.*, 2013; Mahmoud *et al.*, 2012):

$$I = I_l - I_{dr} \left( \exp\left(\frac{V + IR_s}{n_{id}}\right) - 1 \right) - \frac{V + IR_s}{R_{sh}}$$
(2.19)

The five parameters (i.e.,  $I_l$ ,  $I_{dr}$ ,  $R_s$ ,  $R_{sh}$  and  $n_{id}$ ) of the model are very useful in the development of the I-V curve for PV performance analysis.

## 2.7.2 PV module model

For a PV module, with cells connected in series, the same current flow through the cells while voltage is cumulative and is equal to the product of cell's voltage and number of cells (Akinyele *et al.*, 2015). The single-diode model equivalent for a PV module is given by(Tian *et al.*, 2012; Moharil and Kulkarni, 2009; Sangram, 2016):

$$I = I_l - I_{dr} \left( \exp\left(\frac{V_m + \alpha I_m R_s}{n_{id} \alpha V_t}\right) - 1 \right) - \frac{V_m + \alpha I R_s}{\alpha R_{sh}}$$
(2.20)

where  $V_m$  is the module voltage,  $I_m$  is the module current and  $\alpha$  is the number of modules in an array.

## 2.8 Solar PV application

Application of PV power systems is usually categorized into either grid-connected or standalone systems as shown in Figure 2.8 and vary greatly in size and application (Bhandari *et al.*, 2015).



Figure 2.8: Classification of PV systems (Messenger and Ventre, 2003).

A stand-alone system supplies electricity solely to the premise and are broadly classified into hybrid systems, PV with storage systems and PV without storage systems (Messenger and Ventre, 2003; Enerray, 2017). Standalone systems vary widely in size and application from wristwatches or calculators to remote buildings or spacecraft. Those without storage are used to power the loads directly only during sunshine hours making them suitable for applications such as ventilation fans, water pumps and small water pumps for water heating systems or fans in solar air dryers (Ramirez *et al.*, 2018; Weis, 2013). In non-portable applications where weight is not an issue, such as in buildings, batteries are commonly used (Karakaya and Sriwannawit, 2015).

The grid connected systems can be directly connected to the main utility grid where all the power generated is supplied to the grid or can be connected to the utility grid via the locale electrical network where only surplus power generated is supplied to the grid (Elhodeiby *et al.*, 2011). Depending on the solar radiations and the electric energy generated by the PV system, the load can take all of the required energy either from the PV system or can be shared between the PV and the electric grid. Grid connected systems can be enabled through net metering system where producers can sell their surplus electricity to the utility grid and as well buy it during low PV power production (AbdelHady, 2017). Usually, grid connected system does not need battery storage, because all surplus power is transferred automatically to the linked utility grid. These grids connected systems range from small residential and commercial rooftop systems to large utility-scale solar power stations (De-Lima *et al.*, 2017).

#### 2.9 The PV Balance of System

In a PV power system, PV module is the core component and other components needed to avail electricity generated to the loads are called collectively the balance of system (BOS). The BOS included are charge controller, inverters, batteries, wiring and switches. The inverter and the storage systems form the major parts of the subsystem because of their impact on overall PV system efficiency.

#### 2.9.1 Inverters

The inverter is an essential component of BOS subsystem that changes the direct current (DC) from the PV array to alternating current (AC) for use by appliances that operate on AC supply (VignolaFotis *et al.*, 2008). There are two general types of inverters namely, square wave inverters (line frequency switching) and pulse width modulation (PWM) inverters (high frequency switching) depending on the switching techniques used. Nowadays, inverter manufacturers list the overall conversion efficiency in the range of 95% to 98% (Panwar *et al.*, 2017). These efficiencies are

obtained under standard testing conditions (STC) for a PV system in which the array is properly sized for the inverter. This conversion efficiency usually affects the overall performance of the PV systems (Saiful *et al.*, 2006). This efficiency reduces gradually at full load depending on the solar irradiance and inverter operating temperature (Burger and Rüther, 2006).

The choice of the right type of an inverter for any PV system has a greater influence on the optimum performance of the system to meet the different requirements for any application. Modern inverters have incorporated solar charge controllers with maximum power point tracking (MPPT) capability (Alpesh *et al.*, 2020; Danandeh and Mousavi, 2018). Such a device has a better capacity to maximize the solar energy output at any given irradiance levels.

## 2.9.2 Energy Storage systems

One of the major challenges of many renewable energy technologies and particularly the PV technology is intermittency nature of its electrical power output. It is due to this feature that they are installed with storage systems capable of smoothing out the fluctuating output power (Akinyele and Rayudu, 2014; Chen *et al.*, 2009). Various energy storage technologies are already in existence and are classified into four types (Denholm *et al.*, 2010) as shown in Figure 2.9. The electrical energy storage systems include the capacitor storage and superconducting magnetic energy storage systems (Vazquez *et al.*, 2010);The mechanical energy storage systems include the flywheel energy storage, compressed air energy storage and the pumped-hydro storage systems (Young-Min *et al.*, 2012);The chemical energy storage systems include the battery energy storage system (Chen *et al.*, 2013);Finally the thermal energy storage systems which includes the aquiferous thermal energy storage, cryogenic energy storage, hot thermal energy storage and pumped-heat electrical storage systems (Kaldellis *et al.*, 2007).



**Figure 2.9: Classification of energy storage systems** 

The studied PV system uses the battery type of chemical energy storage system. Batteries are the most widely used energy storage systems in PV applications and their popularity is driven by their availability in the market, modularity and their high efficiencies compared to other technologies (Akinyele, 2014; Burke, 2007). There are several existing types of batteries which include Lead-acid, Lithium-Ion (Li-ion), Nickel Cadmium (Ni-Cd), Bickel metal Hydride (Ni-MH), Sodium Sulphur (NaS), Zinc Bromide (ZnBr), Sodium-Nickel Chloride (ZEBRA) and vanadium redox (VRB) (Chen *et al.*, 2009).

Traditionally, lead-acid batteries, invented in 1859 by Gaston Planté, are the oldest technology of rechargeable batteries and have been used widely with standalone PV systems. These lead-acid batteries consist of an electrolyte (concentrated solution of sulfuric acid), electrodes that are the positive (lead oxide) and negative (lead) conductors and the primary part of connected cells (Denholm *et al.*, 2010). These batteries are classified as both the flooded or vented types which require frequent distilled water addition to avoid the battery's electrolyte from depletion and the sealed types that do not require water addition (Young- Min *et al.*, 2012). They have a round-trip efficiency of more than 70% and a life span of 5 - 15 years and a cycle life

of about 2000- 2500 cycles with a low self-discharge loss of 0.1 %/day - 0.3 %/day (Denholm *et al.*, 2010). However, they suffer from deep discharging and overcharging and may bring about lots of pollution if improperly discarded, poisoning and leaks contaminating environment and damaging ecosystem.

Lithium-ion is recognized not only in the renewable energy (PV sector) and electronics field but is also well known in the transportation industries (Whittingham *et al*, 2012).Their low self-discharge losses of 0.1 %/day to 3 %/day and round trip efficiency of about 100% with a life span of 5-15 years make them popular among the other battery technologies (Vazquez *et al.*, 2010). Despite these, they are much more expensive upfront compared to lead acid but they are maintenance free and have longer lifespan.

Nickel cadmium electrode material is useful in many different battery technologies that include the nickel-zinc (Ni-Zn), nickel-hydride (Ni-H2), nickel cadmium (Ni-Cd), nickel metal hydride (Ni-MH) and nickel-iron (NiFe). Among these alloys, Ni-Cd and Ni-MH technologies are the most popular and widely used in the industry. These batteries have a life span of ten to twenty years. However, adverse environmental impact of producing cadmium is their major shortcomings (Denholm *et al.*, 2010).

Vanadium Redox Battery (VRB) technology stores energy through the transference of electron between dissimilar ionic vanadium materials (Vazquez *et al.*, 2010). It has a lifespan of 5 to 10 years with a round-trip efficiency of over 85% with small self-discharge loss per day. However, it has an adverse environmental impact of generating toxic remnants (Denholm *et al.*, 2010).

Zinc bromide technology presents a relatively higher performance though expensive compared to the lead acid batteries. It has a life span of 5 - 10 years, cycling capacity

of over 2,000, a round-trip efficiency ranging from 72% to 85% and a small daily self-discharge loss (Vazquez *et al.*, 2010; Burk, 2007).

The NaS possesses a high power and energy density of than those of lead acid battery however; the battery operates at a temperature ranging from 300 °C to 350 °C. This is because an external heating arrangement is a requisite for its efficient operation. It has relatively higher self-discharge losses of about 20%, a cycling capacity of about 2,500 and a life span 10 to 25 years (Diaz- Gonzalez, 2012).

The ZEBRA is a high energy battery technology with electrodes of nickel chloride as the cathode conductor and liquid sodium as the anode conductor. It possesses a high operating temperature property of ~  $300^{\circ}$ C, cycling capacity of over 2,500, self-discharge losses of ~ 15 %/day and life span of between 10 to 14 years. They can be easily recycled and be developed into new batteries (Diaz- Gonzalez, 2012). In addition, this technology possesses a lower energy and power density compared to the NaS battery systems (Lim *et al.*, 2012).

Metal-air technology presents battery system in form of cell technology that uses a metal as fuel while the air serves the purpose of an oxidizing agent (Lim *et al.*, 2012). It is eco-friendly and can also offer cost-efficient energy storage in the future. However, it possesses a low round trip efficiency and a poor recharging capability (Diaz- Gonzalez, 2012). Its cycling capacity is between 100 and 300, even though it has very small self- discharge losses.

Stand-alone PV systems, like any other renewable energy system, are weather dependent and batteries play a role of ensuring continuous and stable power supply during overcast conditions or autonomous days (Hubble and Ustun, 2018). Operation lifetime of a battery is greatly affected by fluctuations in weather conditions because

of frequent charging and discharging cycles. As a result, battery performance is affected, life time will be decreased and the maintenance and replacement costs will increase. Use of charge controllers help to safeguard batteries from unnecessary deep and over charging and discharging especially the lead acid batteries hence enhance PV system's performance and minimize associated costs (Kan *et al.*, 2006).

## 2.10 Photovoltaic system performance

Once a PV module or array has been installed in the field, an essential requirement is the understanding of the performance exhibited by the system (De-Lima et al., 2017). This is because PV system outdoor performances are different from those described in manufacturer data-sheets which are always obtained at standard test conditions (STC). Use of appropriate performance parameters is vital in the determination of PV system output power over a given specified period and as well facilitates the comparison of PV systems that may differ with respect to design, technology, or geographic location and the prevailing weather conditions at a given location (Gongsin and Saporu, 2020; Beyer et al., 2011 and Okello et al., (2015). In addition, It is largely reported that disparities in the performance of PV systems are attributed to various factors such as BOS (Congedo et al., 2013), meteorological factors (D'Orazio et al., 2014 and Pietruszko et al., 2012), solar irradiance (Al-Addous et al., 2017), cell temperature (Bai et al., 2016 and Zaoui et al., 2015) and effects of dust (Fouad et al., 2017 and Chanchangi et al., 2020). Therefore, there is need to determine performance of PV system at site of application and these parameters be investigated. The most significant of these factors that affect the amount of PV power generated by the PV array is the solar radiation incident on the module (Al-Aboosi, 2020).

#### 2.10.1 Efficiency of PV cell/module

The efficiency of solar PV cell/ module is an important parameter normally used in the determination of its performance and its acceptance rate in the energy market. Currently there is a lot of research work going on and new techniques have been developed with the aim of increasing the conversion efficiency (Ghaffarzadeh and Azadian, 2019). An example of such a technique is the tandem solar cells where the light with a shorter wavelength is absorbed by the outermost material with a wide band gap, while the light with a longer wavelength is transmitted through and absorbed by the material with the narrower band gap (Zhu *et al.*, 2020; Aydin *et al.*, 2020). Investigations on PV module efficiency have led to the development and growth of PV technologies for both domestic and commercial use and in extension reduced its prices (Ozden *et al.*, 2017 and Gaglia *et al.*, 2017).

Nominal efficiency measured at STC is always provided by the manufacturers and usually carried out in controlled laboratories. However, under outdoor conditions, these conditions (e.g., solar radiation) are rarely met and vary with location hence varying cell/module efficiency(Ghazi and Ip, 2014). Moreover, the outdoor operating conditions overlap making it complex to analyze each independently to determine the extent of its influence on system efficiency. But solar radiation and temperature, since they directly influence the energy production, which affect PV module efficiency, can be determined individually (Olchowik *et al.*, 2006; Tanesab *et al.*, 2017; Lorenzo *et al.*, 2014 and Louwen *et al.*, 2017).

## 2.10.2 Effect of solar irradiance on PV performance

Solar irradiance is the most important parameter for assessing a PV plant performance and its measurement is an important exercise for performance evaluation (Salih *et*  *al.*,2012). Solar radiation directly affects the PV array current where, as solar radiation increases, the short circuit current, maximum power, and conversion efficiency also increase (Irwanto *et al.*,2010). Development and growth of PV technology has significantly led to the development of methods and technology for solar radiation measurement with the aim of using the data to rate PV system performance on various scales under outdoor conditions. In PV system performance studies, solar insolation data play a significant role in characterization and evaluation of PV modules performance and development of new technologies.

Figure 2.10 shows the effect of varying solar irradiance on the I-V characteristics of a PV module under constant temperature. From the figure, there is a huge impact on output current with a slight change in solar radiation. The output voltage show slight change with increase in solar radiation. There is a point on the curve known as the maximum power point (MPP) where generated power from the PV modules is maximized under given solar irradiance (Yali *et al.*, 2014). This point shifts upwards (vertically) with increasing solar insolation e.g., from 18.4 W for solar radiation of 200 W/m<sup>2</sup> to 100.5 W for solar radiation of 1000 W/m<sup>2</sup>. These points of maximum power generated by a PV module are tracked by the Maximum Power Point Tracking (MPPT) schemes also called MPPT trackers incorporated in the inverters (Mohamed *et al.*, 2013; Venkata and Muralidhar, 2016; Emilio *et al.*, 2014 and Mahmoud *et al.*, 2012).



Figure 2.10:TheI-V characteristic at varying solar irradiance under constant temperature (Source: PVSyst simulation software library)

# 2.10.3 Effect of ambient temperature on PV performance

Ambient temperature is an outdoor operating condition factor that plays an important role in the PV module energy conversion process where its variation affects the module efficiency (Fesharaki *et al.*, 2011; Siddiqui and Bajpai, 2012).Cell/ module temperature which is a factor of ambient temperature and solar irradiance as shown in equation 2.22 affect the performance of a PV module where module performance decreases with increasing cell temperature (Kawajiri *et al.*, 2011). This is because of increased internal carrier recombination in the semi-conductor material caused by increased carrier concentration. Both the electrical efficiency and the power output of a PV module depend on the cell temperature where it can reduce output efficiency by 10% - 15% (Skoplaki and Palyyos, 2009).

The open circuit voltage of a PV module and the fill factor decreases substantially with temperature due to increased thermally excited electrons that begin to dominate the electrical properties of a PV cell. On the other hand, the short circuit current slightly increases with temperature (Thong *et al.*, 2016; Skoplaki and Palyvos, 2009). Thus, the electrical efficiency of a PV module is dependent on temperature and its net effect leads to a linear relation expressed as (Kawajiri *et al.*, 2011):

$$\eta_{\rm mod} = \eta_{\rm ref} [1 - \beta_{\rm ref} (T_c - T_{\rm ref})] \tag{2.21}$$

where  $\eta_{ref}$  is the module's electrical efficiency at the reference temperature (i.e., STC),  $T_{ref}$ ,  $\beta_{ref}$  is the temperature coefficient of silicon and  $T_c$  is the PV cell/module temperature. The quantities  $\eta_{ref}$  and  $\beta_{ref}$  are normally given by the PV manufacturer. The PV cell/ module temperature in most cases is not measured but normally evaluated from nominal operating cell temperature,  $T_{NOCT}$  (measured under opencircuit conditions), and is expressed as (Anis *et al.*, 1983; Debbarma *et al.*, 2017; Griffith *et al.*, 1991):

$$T_c = T_{air} + (T_{NOCT} - 20) \frac{G(t)}{800}$$
(2.22)

where G(t) is the measured solar irradiance and  $T_{air}$  is the ambient temperature.

Figure 2.11 shows the behavior of PV module current and voltage with variations in temperature between 10 °C and 70 °C, in steps of 15°C under constant solar irradiance (Mayfield, 2012). The Figure was obtained from the PVSyst simulation software library. From Figure 2.11, the module's voltage decreases as the cell/ module temperature increases, while the current shows little or no increase with temperature at constant irradiance (Irwanto *et al.*, 2010). In addition, there is a drop shift of the MPP point position with increase in cell/ module temperature. Shifting of the MPP point position is largely attributed to changes in the PV cell/ module temperature.

PV module: Generic, Poly 190 Wp 54 cells



Figure 2.11: I-V characteristics of a PV module at constant solar irradiance (Source: PVSyst simulation software library)

### 2.10.4 Installation setup and performance

The PV system performance is affected by its design, configuration and BOS characteristics (Díez-Mediavilla *et al.*, 2012). The design (installation setup) is characterized mainly by the PV module tilt angle and orientation (azimuth angle).Tilt angle of a PV module is the angle measured between the PV module and a horizontal surface or the vertical surface representing the x or y direction respectively. For the PV stand-alone systems with battery storage and permanently fixed, their tilt angle is an angle equal to the latitude of the installation site (Joshua *et al.*, 2014). In addition, Degradation over time of the PV system components decreases the performance of the PV system (Makrides *et al.*, 2014). Degradation technique is used to determine the decrease in maximum power point performance over time and depends on several factors, such as the technology, the operating conditions and the cumulative history of environmental exposure (Phinikarides *et al.*, 2014).

### 2.11 IEC Standards

Standards for PV module performance evaluation have been developed by the International Electrotechnical Commission (IEC) established in the year 1995 though some countries still have their own national PV standards (Spooner and Harbidge, 2001). The IEC is a worldwide organization for standardization, and its objective is to promote international co-operation on all issues concerning standardization in the electrical and electronic fields. It develops and publishes the international standards for electro- technology for a wide range of technology with the help of experts from various sectors such as academia, industry, and government sector and research laboratories. The IEC standards in PV were developed by TC82 Solar photovoltaic energy systems (Wohlgemuth, 2012).

This International Standard recommends procedures for monitoring of energy-related PV system characteristics such as in-plane irradiance, array output, storage input and output, power conditioner input and output and for the exchange and analysis of monitored data (Mcnutt *et al.*, 2000). The IEC standard 61724 (Photovoltaic system performance monitoring, 1998) basically concentrates on the PV system performance monitoring through methodologies for evaluating performance of the system in respect to site prevailing operating conditions. It provides procedures and recommendations on the PV system performance guidelines for measurement, data collection and analysis (El-Chaar *et al.*, 2011). The IEC/TS 61836 presents definitions, terms and symbols of PV energy systems. The IEC/TS 62257-7 consider the PV system components selection, installation, sizing and operation and maintenance for small renewable energy systems. The IEC/TS 62257-7-1 present technical specifications for small renewable energy systems for their design and

development for off grid applications. The IEC/TS 62257-9-1 provide standards for system selection for the right locations considering their technical specifications. The IEC/TS 62257-2 emphasize on methodologies useful for evaluating the socio-economic aspect of stand-alone renewable energy systems from specified energy demand requirements. The IEC/ TS 62257-4 provide technical specifications and recommendations for system selection and design for small renewable energy and hybrid systems with emphasis on user's energy needs (Hibberd, 2011; Wohlgemuth, 2012).

Three of the IEC standards 61724-1 (2021) performance parameters may be used to define the overall system performance with respect to the energy production, solar resource, and overall effect of system losses. These parameters are the final PV system yield, reference yield, and performance ratio. The IEC standard parameters that are used to characterize the performance of solar cells are the peak power ( $P_{max}$ ), the short-circuit current density ( $J_{sc}$ ), the open circuit voltage ( $V_{oc}$ ), efficiency and the fill factor (*FF*) (Al-Aboosi and Al-Aboosi, 2021).

## 2.12 Literature review

A lot of research work on PV systems has been done and reported in peer reviewed journals, conference proceedings and technical reports. The work includes research on different PV systems, technical performance, economic and environmental performance of PV systems, degradation tests, applications, regional performance tests etc. These work findings are reviewed in this section.

Samoita *et al.*, (2020) explored barriers and solutions for increasing integration of PV systems in Kenya's electricity mix as a means of diversifying and stabilizing electricity supply. In their study, they pointed out a variety of technical, economic,
institutional and political barriers which currently restrict further growth of PV technology deployment in Kenya. Key among the technical barriers includes the lack of PV performance information as well as lack of operation know-how of PV systems among potential Kenyan customers. These barriers, as stated by the authors, can be overcome with robust policy regulations, additional investments in education, training, research and development, better regulation of the electricity sector and improved coordination between key actors. Their results may be applicable to other sub-Saharan African countries of whom many are faced with similar challenges.

A research study on techno-economic analysis of off-grid solar PV system for rural electrification in Punjab, Pakistan was discussed by Irfan et al., (2019). The research results reveal that there is an excellent solar irradiance in the rural areas of Punjab for electricity generation. In addition, the authors also found that conventional energy sources are more expensive than the off-grid solar PV system as electricity generation from conventional sources is 0.092 \$/kWh, while it is only 0.032 \$/kWh for the offgrid solar PV system. The study further reveals that 617,020 metric tons of CO<sub>2</sub> could be mitigated annually by electrifying 100% rural households with the off-grid solar PV system. In their conclusion, the authors suggested essential policy recommendations that would serve as a guideline for the government and stakeholders to maximally deploy the off-grid solar PV rural electrification programs. A similar research has been presented by Omar and Mahmoud, (2018), for a selected number of home systems in Palestine, showing a payback in less than five years for a 5 kW system. Additionally, Allouhi et al., (2016) studied the energy production from two PV technologies in an institutional building in Morocco by evaluating the economic and environmental aspects and comparing them with other PV plants worldwide.

A study has been presented on experimental performance evaluation of a PV array with respect to maximum power voltage, maximum power current, maximum power output, power output losses and fill factor (FF) values by Abdullah et al., (2021). The setup involved a PV array with series configuration of 2×4 PV modules. Their work aimed to provide information on PV array performance as a whole considering site shading effect. Such information is useful for planning and system site installation. Results showed that in the case of the series configuration of a PV array, even if only one PV module fails to generate electricity due to an event of any failure or partial shading, the total amount of power generated is reduced. The power output of a PV array as discussed by Micheli et al., (2014) is a factor of various features such as shading pattern, number and size of modules and wiring scheme of the modules in a PV array. Musanga et al., (2018) reported that output current is a factor of solar irradiance while the module temperature has more effect on the output voltage which in essence the two controls the power output of a PV array. These two operating conditions, most significantly the solar irradiance, provide insight into PV array electrical characteristic performance information.

Nkhonjera and Wu, (2013) in their study on performance analysis of a battery based standalone system (SHS) installed in Malawi to expedite the country's electrification. They considered both low and high insolation sites of Malawi. Results showed that the system is capable of operating annually with a mean performance ratio(PR) of 0.68, PV array production factor of 0.88 and PV performance ratio of 78% which indicate a good system performance. In addition, if the system is under-sized, its performance ratio is reduced considerably and the system becomes less reliable. However, the authors noted that the system is not totally reliable all year round since

it cannot meet the demand in 9% of the total yearly operational time if located in high insolation areas and 13% of the year if located in low insolation areas.

Bhuvaneswari *et al.*, (2018) carried out a study on performance analysis of a 15 kW standalone solar PV system installed in Vellore District, Tamil, and reported  $E_{Dc}$  ranging from 6,500–7,000 kWh, *PR* of 78%, and utilization factor of 6.97%. In a similar study, Ezenugu *et al*, (2016) carried out performance analysis of stand-alone PV system in a health clinic in Nigeria, and reported annual DC energy output ( $E_{Dc}$ ) of 5269 kWh/year, *PR* of 58.4%, moderate operating efficiency of 8.83% and average system loss of 7.1%.

Kessaissia *et al.*, (2015) characterized two silicon-based PV modules i.e. monocrystalline and polycrystalline silicon based on site prevailing outdoor conditions. The I-V characteristics were obtained using two mathematical models i.e. the implicit model and the explicit model. In the implicit model, five coefficients were calculated analytically whereas in the explicit model, two coefficients were calculated from experimental measurements. Outdoor conditions considered are solar insolation and temperature in addition to data provided by manufacturers. The results show a strong agreement between the implicit model and the experimental characteristics. In their conclusion, the five parameters model is better and more accurate than the explicit model where it is preferable to use the method according to IEC 891. In a similar study, Bhowmik and Amin, (2019) carried a comparative simulation analysis study of m-Si and p-Si PV modules to study the performance of the modules at the specific tilt angle of 32°. The parameters measured include solar intensity, output voltage and current, ambient temperature, humidity, wind velocity, output power efficiency and module efficiency. They concluded that m-Si PV module performs

better with the module efficiency of around 8.5% higher than that of p-Si PV module. Another comparative study between m-Si and p-Si PV modules at an optimized tilt angle was reported by Cornaro and Musella, (2010). Their results showed that p-Si module is highly stable with an average performance ratio (PR) of 88%.

Olatomiwa *et al.*, (2014), in their study, employed the Hybrid Optimization Model for Electric Renewables (HOMER) tool to simulate hybrid energy systems such as the PV/diesel generator, PV/Wind turbine/ diesel generator and diesel generator only. Using remote telecommunication tower in Nigeria, they focused on the technical and economic assessment of the energy options and also compare the amount of carbon emissions produced by the systems. The peak load and average demand per day of the telecommunication system are 3.3 kWh and 37 kWh respectively. The sizes of PV, diesel generator, and wind turbine are 8.0 kW, 5.5 kW, and 1.0 kW, respectively. The research results reveal that the energy configurations that incorporate renewable energy systems offer the most economical solutions as well as contribute to the reduction of carbon emissions.

A paper on the analysis of PV system based on energy output and carbon credit earned for the climate of Uttah Pradesh, India has been published by Rajput *et al.*, (2017). It focuses on yearly performance of the system whose lifespan is 25 years. Embodied energy and Energy Payback Time (EPBT) is also analyzed for calculation of reduction in CO<sub>2</sub> emission while generating electrical energy. The research results reveal that embodied energy of the installed PV system is 8493.16 kWh and EPBT of the system is 5 years. Reduction in CO<sub>2</sub> emission; when using PV system against the coal-based plant for the generation of electricity is 2.525 tCO<sub>2</sub>eq and the monetary saving due to carbon credit is 143162.19  $\gtrless$  for life time of 35 years. Further, they recommended that EPBT can be reduced if the output of the system increased further with a greater number of clear days. However, their research did not consider monetary savings due to carbon credit during economic analysis of PV system.

Integration of PV systems into the rooftops of residential buildings offers a considerable amount of clean and renewable energy that could slightly move towards achieving the concept of zero-energy buildings and contribute to lower CO<sub>2</sub> emissions as discussed in a research study by Abdelhafez *et al.*, (2021) in Hail, Saudi Arabia. Results reveal that there is a considerable amount of energy production from the use of all residential rooftops in Hail, reaching more than 346 GWh. In addition, there is a significant reduction in the amount of CO<sub>2</sub> emissions reaching nearly 5.9tCO<sub>2</sub>/year. Further, the PR and the system efficiency were affected by tilt angle of the PV module. The efficiency increases with higher tilt angles because with decrease in PV module temperature its efficiency increases. Besides, the results pointed out that the  $30^{0}$ -tilt angle PV produced the highest amount of energy, whereas the  $75^{0}$  tilt PV records the lowest amount of energy though it achieves the best possible efficiency.

González-Mahecha *et al.*, (2019) carried out research on greenhouse gas (GHG) mitigation potential and abatement costs in the Brazilian residential sector. Their findings show that energy efficiency measures in the cooking end-use and solar PV systems would represent together more than 70% of the abatement potential. In addition, the total avoided emissions would be 642 MtCO<sub>2</sub> in Brazil over the period of 2010–2050. In a similar study, Tiwari *et al.*, (2015) carried out enviroeconomic analysis of a stand-alone PV system and their results show that the amount of CO<sub>2</sub> emitted per kWh is approximately 960 g. However, from their findings, this amount

rises to 2.0 kg of  $CO_2$  per kWh if transmission losses (40%) and distribution losses (20%) are considered.

An economic analysis of an off-grid PV system designed for household electrification was presented by Ghafoor and Munir, (2015). The paper focuses on a single household application in Faisalabad, Pakistan that requires 1.928 kW of solar PV to support an average daily demand of 5.9 kWh. The results indicate that the system's life cycle cost and the cost of energy are lower than those of conventional systems. However, the study does not consider energy losses, battery state of charge, reliability, load growth and the life cycle impact analyses.

A life-cycle cost (LCC) analysis of various combinations of PV and diesel generators for a school in India was done by Kolhe *et al.*, (2002). Their study drew a conclusion that a stand- alone PV system is the most viable option when the power needs for the school are minimal and that PV systems will become more competitive as their costs decline with improved technology and efficiency.

Liu *et al.*, (2017) used measured data from 15 radiation stations to validate different empirical estimation methods of daily solar irradiation over the Tibetan Plateau. The highly rated sunshine- based Ångström model and temperature-based Bristow model were selected for the site application. Their result through calibration indicates that sunshine- based site- dependent models perform better than temperature-based ones. To achieve better performance, the Ångström- type model was improved using altitude and water vapor pressure as the leading factors. However, they inferred that the model should be further validated against observations before its applications in other plateau mountainous regions. Several computer software tools have been developed and employed to analyze the performance of PV systems in different parts of the world. These tools include PVsyst, SolarMAT, HOMER, among others. The PVsyst software has been used widely by many researchers to simulate and study the performance of PV system and was selected for this study. Spea and Khattab, (2019) designed and carried out performance analysis of stand-alone PV systems using PVSyst software for a location in Egypt. The system studied consists of PV panels, batteries, inverter, and charge controller. They employed PV panels of 450 W and 260 W in their study based on watt-hour demand calculations for comparisons. Their results showed that both PV panels can feed the desired load. A 450W PV panel is better than 260W PV because higher energy conversion efficiency was obtained; higher value of energy produced per year, and occupied a lower area. Kumar et al., (2020) employed PVsyst tool to design and simulate a standalone solar PV system in India. Analysis of performance ratio and losses are reported. Their results showed that the energy required was 1086.24 kWh and the energy available through solar panel was 1143.6 KWh, whereas energy supplied to the user was 1068.12 kWh with average PR for the year was 72.8%. Sifat et al., (2015) designed a 2 kW Stand-alone PV System with backup in Bangladesh Using PVsyst, Homer and SolarMAT software tools. The PVsyst showed mismatch with practical data to a small extent and a good choice for economic analysis while output results generated by SolarMAT has a great similarity with practical data. HOMER would be preferable for hybrid solar system designs.

Ammu *et al.*, (2017), in their study, carried out a cost-effective assessment of PV system as a backup source of power in an Urban Indian context. They compared five different battery technologies and conventional grid backup systems such as inverter-battery versus diesel engine options. Life cycle costing (LCC) and the energy payback

time (EPBT) tools were used to determine the economic viability conditions of the battery technologies and the grid backup systems while the net energy ratio (NER) was used to determine their technical aspects. Their results show that Li-ion battery technology has high NER and lowest annualized energy requirement. Further, the two grid backup systems have initial comparable costs with inverter- battery system showing relatively low long term costs and at the same time have high NER.

Ndwali *et al.*, (2020) proposed an optimized operation control strategy of micro grid connected PV system with diesel generator backup system. Their aim was to reduce energy purchased from utility grid and the fuel consumption cost of the diesel generator based on the daily load profile of an Engineering workshop at Jomo Kenyatta University of Agriculture and Technology (JKUAT), Kenya. Their results show a daily energy saving increase of up to 52.1%, daily cost saving of 20%, and daily generated income of \$17. The study did not deal with uncertainties and disturbances associated with micro grid systems.

Rahil *et al.*, (2018) presented an economic and environmental analysis of PV energy system as an option to meet the electrical energy demand in a reliable and sustainable manner for regions without grid connection in a remote village in South Africa. The PV systems presented are the PV system with battery storage and PV systems with diesel generator and battery storage. Their results show that the total cost in a PV system with battery storage represents only 26% of the entire PV system. In addition, the PV with Battery system release minimal harmful emissions compared with nearly 6.0 tCO<sub>2</sub>/year in the PV with Diesel generator system. Both the systems sufficiently meet the load demand with no interruptions.

Diesel generators are known to have a relatively low initial investment cost compared to PV systems but carry health and environmental risks as reported by Babajide *et al.*, (2021). In their work, they used measured data from a monitoring campaign in Lagos, Nigeria for analysis. Their work addressed making cleaner electricity through solar PV more attainable, increasing access to more reliable power, and reducing or eliminating the use of diesel generators. Their results showed that a cost saving of about 60% - 65% over the project life can be attained by use of PV system in place of diesel generators for backup power generation.

#### **CHAPTER THREE**

#### **METHODOLOGY AND MATERIALS**

#### 3.1 Introduction

This chapter presents description of the study location, instrumentation and data collection, PV backup system performance parameters and the simulation tool used in this thesis. The study was carried out both experimentally and theoretically.

# 3.2 Study location

The study was carried out at Kaiboi Technical Training Institute (KTTI) located in Nandi County, western Kenya. Figure 3.1 (a) shows the map of Kenya and the region where the study was carried out. Figure 3.1 (b) shows the map of Mosop constituency in Nandi County and the site location where the study was carried out. The site has geographical coordinates of 0.42 <sup>0</sup> N (hence is almost to the equator) and 35.03 <sup>0</sup> E with elevation of 1993 m above sea level. The site is located in a flat terrain, hence not affected by terrain shading in the morning or evening. The location experiences moderate to warm temperatures with mean of between 18 °C to 28 °C with an average rainfall ranging from 1200 mm to 2000 mm per annum. The location falls within tropical climatic zone, and as such is expected to have abundant solar radiation throughout the year.







**(b)** 

Figure 3.1: (a) Map of Kenya showing study region - Nandi County; (b) Map of Mosop constituency showing the study site.

#### **3.3** Experimental setup

The experimental system studied was an existing PV power system installed at the social/dining hall of KTTI. The system was installed in the year 2016 and has been operating since then as a power backup system to electrify the appliances in the social/ dining hall whenever power outages occur. Figure 3.2 is a block diagram showing the components and connections of the PV system studied and how it is integrated into the electrical power network of the social/ dining hall as a backup system. The basic components of the system are the PV modules, charge controller, battery bank, an inverter, loads (appliances), and the switch over control system.



Figure 3.2: Schematic diagram of the studied PV backup system

#### **3.3.1** The PV modules

The PV module is the back-born of any PV power plant system because it is the generator of the electrical energy. Four identical polycrystalline PV modules each with a power rating of 195 W are used in the system and are mounted on the roof of the social/ dining hall of the institution as depicted in Figure 3.3. Thus, the tilt angle and orientation (azimuthal) of the PV array is dictated by the pitch angle and the orientation of the roof of the building. The roof of the social/ dining hall has a pitch angle of 15° and oriented in the NE-SW direction, which are respectively the tilt and azimuthal of the PV modules. As it can be observed from Figure 3.3, the PV modules were mounted on a metallic frame for securing them to the roof and an air gap of about 18 cm was provided at the rear of the arrays to provide cooling during dry seasons by allowing natural (buoyancy) air circulation. The natural cooling is necessary because the electrical conversion efficiencies of the PV modules degrade with increasing temperatures of the solar panels.



Figure 3.3: Photograph of roof-mounted PV Array at the Institute's Social/ dining hall.

Table 3.1 presents the technical specifications of the modules used. In the present system, the PV array consists of two pairs of series connected modules, which are in turn connected in parallel. Therefore, the resulting bus voltage of the array is 24 V. From Table 3.1, the rated output peak Watt ( $W_P$ ) of each module is 195 W, hence, the nominal output power is 780 W. In addition, the nominal efficiency of each module is specified by manufacturer as 15.6%.

Parameter	A (m <sup>2</sup> )	V <sub>oc</sub> (V)	I <sub>sc</sub> (A)	P <sub>o</sub> (W)	NOCT (°C)	V <sub>mp</sub> (V)	I <sub>mp</sub> (A)	$\eta_{ m ref}$ (%)	β (°C <sup>-1</sup> )
Value	0.4	44.5	5.70	195	45±2	36	5.56	15.6	0.0041

 Table 3.1: Technical specification of PV module

# **3.3.2** Inverter/Controller unit

The charge controller and the inverter are usually separate components, but in the present system, they are integrated together as one unit, and their functions are respectively to monitor charging/ discharging processes of the battery and conversion of dc power from PV modules to ac power for integration into the ac network of the social/ dining hall. The switch over control system is also integrated in this inverter/ controller unit, and is a logic switch which triggers automatically the switching from either grid to solar and vice-versa. Whenever a blackout occurs, the control switch automatically connects the PV system to power the social /dining hall appliances, and when grid power is restored, it disconnects the PV system. In addition, this unit has an integrated data acquisition system which records electrical quantities such as current, voltage and energy generated by the PV system when operating.

Figure 3.4 shows the picture of the inverter/ controller unit. The inverter in the studied PV system has maximum output ac voltage of  $230 \pm 0.05$  V and 50 Hz line frequency (i.e., ac grid frequency in Kenya). The charge controller monitor the state of charge of the battery during charging and discharging cycles by disconnecting the batteries automatically from the PV array to prevent overcharging during charging or loads during discharging to prevent deep discharge. The output DC voltage and current of the PV array were measured and recorded in an internal data storage card integrated in the inverter/ charge controller unit. The electrical data were measured using a digital multimeter integrated in the inverter that measures the voltage and amperes being generated by the PV array. To measure the DC voltage the "DC V" was selected and to measure the DC current the "DC I" was selected manually and time duration and interval set for the period that data will be recorded automatically after turning on the multimeter. The technical characteristics of the inverter and charge controller are given in Appendices II and III respectively.



Figure 3.4: Picture of Integrated Inverter/Controller Unit.

#### **3.3.3 Battery Storage System**

The studied system uses a battery bank consisting of six (6) lead-oxide batteries mounted on a wooden cage to ensure electrical safety in its handling and use. Spacing is provided between the batteries to provide air channels for natural convective air cooling. Each battery has a power rating of 200 Ah and nominal output voltage of 24 V. The batteries are connected in parallel to match the PV module bus voltage of 24 V and to increase output current, hence output power to match the load demand. Figure 3.5 shows the picture of battery bank, and the specifications of the batteries are presented in Appendix IV.



Figure 3.5: Picture of Battery Bank System.

# 3.3.4 Loads

The PV system is adopted as a backup system; hence it will power the existing loads in the social/dining hall building during grid power outages. The power rating of each load/appliance in the building was assessed and the findings are listed in Appendix V. A worst-case scenario of two hours blackout per day was assumed and used to estimate the daily power that the PV system must supply to meet the demand. The total power rating excluding all the losses was approximately 2.076 kWh/day. Fridge was operated as base load and since the location is part of the tropical region, fans are operated mainly during midday hours in the months that experience relatively high insolation and ambient temperatures.

# **3.4** Instrumentation and data collection

Outdoor measurements of the PV operating conditions and electrical parameters of the PV system were done at the site and recorded. The operating conditions measured were the solar radiation, ambient temperature, and wind speed, while the electrical parameters were the output voltage and current of the PV array. The data were measured at intervals of five minutes and logged by data acquisition cards incorporated in a radiometer for solar radiation and inverter/controller unit for electrical parameters. The data were collected for a period of one year in 2020.

Figure 3.6shows the pyranometer readout meter (model SPM-1116SD) that was used to measure solar irradiance intercepted by the PV modules. It has an extension probe connected to the digital recorder with an internal data acquisition and storage card integrated in the meter system where data were measured and recorded. The solar sensor was mounted on the surface of the PV array. Technical specifications of the meter are given in Appendix VI. The data were downloaded from the card as an Excel file and stored in the computer for analysis.



Figure 3.6: A picture of a digital recorder and the pyranometer (model SPM-1116SD)

The ambient temperature and wind speed data were obtained from a nearby meteorological station. Ambient temperature was measured using type-K thermocouple, which consists of alloy compositions NiCr (+) and NiAlSi (-)mounted on a secure shaded region to prevent direct heating from the solar radiation while wind speed was measured in m/s using a rotating cup anemometer mounted in an open at 10 m from the ground level.

# 3.5 **Performance functions of PV modules**

The parameters recommended in IEC61724-1 (2021) standard to qualify the performance of a PV system include conversion efficiency, the fill factor (FF), yield ratios (reference yield,  $Y_r$ , final yield,  $Y_f$  and system losses,  $L_s$ ), performance ratio

(PR), and capacity factor (CF). These parameters were evaluated and used to characterize the performance of the studied PV system.

# 3.5.1 Efficiency, fill factor and output energy

The energy conversion efficiency of a PV array,  $(\eta_{array})$  is the ratio between the maximum electrical power that the array can produce to the amount of solar irradiance received at the surface of PV array (Ayompea *et al.*, 2011; Rakhi and Tiwari, 2012):

$$\eta_{array} = \frac{P_{max}}{A \times G_t} \times 100\% \tag{3.1}$$

where A is the PV array surface area,  $G_t$  is the POA solar irradiance and  $P_{max}$  is the PV array output power generated. The  $P_{max}$  is given as (Al-Addous *et al.*, 2017):

$$P_{max} = V_{oc}I_{sc}FF = V_{max} \times I_{max}$$
(3.2)

where  $V_{oc}$  is the open circuit voltage,  $I_{sc}$  is the short circuit current,  $V_{max}$  is the maximum voltage and  $I_{max}$  is the maximum current values both determined at the MPP on the I-V characteristic curve of the PV module, and *FF* is the fill factor. Fill factor is the ratio of the actual maximum obtainable power (represented by the area of the largest rectangle which will fit in the I-V curve) to the product of short  $I_{sc}$  and  $V_{oc}$ . It gives an idea of the quality of the array, and the closer it is to unity (1), the more power the array can produce. Typical values are between 0.7 and 0.8 (Ayompea *et al.*, 2011; Jain, 2004).

The energy output from the PV array is an integral factor of the measured solar irradiance and the array efficiency as determined in equation (3.1) and is given as (Abawi *et al.*, 2016; Jain, 2004; Al-Addous *et al.*, 2017; Kim *et al.*, 2021):

$$E_{PV} = \int_{t=1}^{n} \frac{1}{6000} G_{t} A \eta_{array}$$
(3.3)

# 3.5.2 Yield factors, performance ratio and losses

The  $Y_r$  and  $Y_f$  ratios are defined by IEC 61724 standard (Usman *et al.*, 2020; Aguilera and Espinoza, 2019; AlYahya and Irfan, 2016; Satsangi *et al.*, 2018; Almarshoud, 2017) respectively as:

$$Y_r = \frac{G_t}{G_o} \tag{3.4}$$

And

$$Y_f = \frac{E_{PV}}{P_o} \tag{3.5}$$

where  $G_0$  is reference irradiance (i.e., irradiance at STC or 1000 W/m<sup>2</sup>),  $E_{PV}$  is the net dc output energy generated by the PV systems over a specified period (e.g., a day, month, or year) and  $P_0$  is the nominal DC output power of the PV array. The  $Y_r$  is a measure of the number of hours when the solar irradiance is above 1000 W/m<sup>2</sup> and hence has units of kWm<sup>-2</sup>, and is normally referred to also as solar peak hours (PSH) in PV literature. Also,  $Y_r$  is a function of the location, orientation of the PV array and seasons of the year (Ketjoy *et al.*, 2013; Malvoni *et al.*, 2017; Aguilera and Espinoza, 2019, Pearsall, 2016; Almarshoud, 2017; Skoczek *et al.*, 2011). The  $Y_f$ , on the other hand, represents the number of hours that the PV array would need to operate at its rated power to provide the same energy and is useful when estimating the return on investment (ROI) of a PV system. The unit for  $Y_f$  is hours or kWh/kW and it is also used as a convenient parameter to compare the energy produced by PV systems of differing sizes (Ferrada *et al.*, 2015; Muñoz *et al.*, 2016; Aguilera and Espinoza, 2019; Sreenath *et al.*, 2020).

Performance ratio (PR) also known as quality factor, gives a measure for the degree of utilization of an entire PV system (Sidi *et al.*, 2016). It indicates the overall effect of losses on the overall performance of the PV system, and includes effects of PV array temperature, incomplete utilization of irradiation; system component limited efficiencies, and failures (Reich *et al.*, 2012). It is defined in the standard IEC 61724-1 (2021) as the ratio of final PV system yield ( $Y_f$ ) to the reference yield ( $Y_r$ ) expressed as (Trillo-Monteroet *et al.*, 2014; Almarshoud, 2017; Chioncel *et al.*, 2010; Reich, 2012):

$$PR = \frac{Y_f}{Y_r} \tag{3.6}$$

The relation indicates whether the system is operating optimally and quantifies the overall effect of losses on the rated output power. These losses are caused by many factors including the inverter losses, wiring and mismatch, module temperature, incomplete use of irradiance by reflection from the module front surface, soiling or snow, system down-time and component failures (Usman *et al.*, 2020; Louwen *et al.*, 2017). For the studied system, PR values are reported on monthly basis.

The system losses ( $L_s$ ) were obtained through evaluation of the difference between PV array productivity and the overall productivity which in essence represents the inverter conversion losses expressed as (Cherfa *et al.*, 2015; Ayompea *et al.*, 2011; Sidi *et al.*, 2016; Bajpai and Yadav, 2018; Martinez *et al.*, 2019; Žnidarec *et al.*, 2019):

$$L_s = Y_r - Y_f \tag{3.7}$$

Capacity factor (CF) is a dimensionless ratio of an actual electrical power output from a PV system ( $P_o$ ) over a given period of time to the energy generated by the PV system when it operates at its full rated power  $P_{max}$  over similar period. It indicates the extent of use of the power generating PV system in percentage and is expressed as (Usman *et al.*, 2020, Kumar *et al.*, 2014; Pundir *et al.*, 2017; Emziane and Al Ali, 2015):

$$CF = \frac{P_o}{P_{\text{max}}} \times 100 \tag{3.8}$$

In addition, the CF measures PV actual generation compared to the rated maximum generation over a given period of time and is bound by the number of sunshine hours (Ayompea *et al.*, 2011). The average monthly capacity factor was considered for performance analysis in this study.

# **3.6** Economic performance indicators

The study compares the economic performance of the PV system with that of an equivalent diesel power generator where a life cycle cost (LCC) analysis technique is used to evaluate their costs. The LCC of a system is the sum of all the costs incurred during the respective system life time (Allouhi *et al.*, 2019; Abd El-Shafy, 2009). It takes into account the applicable inflation and interest rates. Inflation rate measures the degree to which the value of money has declined. Interest rate provides the information about the amount of profit that is obtainable from saving a sum of money (Akinyele and Rayudu, 2016). The inflation and interest rates world over change with time, which is why all costs and evaluation are usually reviewed and updated. The LCC analysis starting year was set to be 2016 which is the year the PV system was

installed and in order to reflect the PV module and DG market status. The analysis period was set to be 25 years, which is the service life of the PV modules (Rodrigues *et al.*, 2016; Kaplanis and Kaplani, 2011). The current reflecting average annual value of 2.6% is taken for inflation rate and 6.0% for interest rate respectively (Economic survey, 2021).

#### 3.6.1 Life cycle cost analysis

The PV system and the DG costs were determined by application of the *LCC* technique. The LCC technique entails determination of all costs from component acquisition, operating, maintenance to replacement expressed as the present worth given as (Assad, 2010; Allouhi *et al.*, 2019; Pacca *et al.*, 2007; Braker *et al.*, 2011; Branker *et al.*, 2011; Dally, 2013; Gulaliyev *et al.*, 2020; Pillai and Naser, 2018):

$$LCC = \sum_{r=1}^{6} C_r \tag{3.9}$$

where  $C_r$  is the individual component present worth expressed as:

$$r \in \langle 1, 2, 3, 4, 5, 6 \rangle \equiv \left[ C_{PV}, C_B, C_{CON}, C_{INV}, C_{INST}, C_{O\&M} \right]$$
(3.10)

where  $C_{PV}$  is the PV array cost,  $C_B$  is the battery cost,  $C_{CON}$  represent charge controller cost,  $C_{INV}$  represent the inverter cost,  $C_{INST}$  represent installation cost and  $C_{O\&M}$  represent operation and maintenance costs.

To determine the present worth of the battery we use the equation as expressed below (Assad, 2010):

$$C_{B} = C_{BO} + \sum_{k=1}^{j} C_{BO} \left(\frac{1+d}{1+i}\right)^{n}$$
(3.11)

where *i* (%) is the interest rate, d (%) is the inflation rate,  $C_{BO}$  is the initial cost of the battery, *j* is the number of replacements and *n* is the battery life span.

For the operation and maintenance  $costs C_{O\&M}$ , the present worth is expressed as (Oliver and Jackson, 2001):

$$C_{O\&M} = C_{O\&M/y} \left(\frac{1+i}{1+d}\right) \frac{1 - \left(\frac{1+d}{1+i}\right)^{N}}{1 - \left(\frac{1+d}{1+i}\right)^{N}}$$
(3.12)

where  $C_{O\&M/y}$  is the operation and maintenance cost per year and N is the life span of the PV module

The number of years it takes to recover the initial investment cost (i.e. the number of years needed to balance cumulative discounted cash flows and the initial investment), the payback time (*PBT*) is calculated using the following equation (Lai and McCulloch, 2017; Rodrigues *et al.*, 2017; Alsema, 2000):

$$PBT = \frac{LCC}{Q_{AP} \times UEC_{KE}}$$
(3.13)

where  $UEC_{KE}$  is the cost of electricity supply in Kenya and  $Q_{AP}$  is the annual energy production from either the PV system or the DG expressed as:

$$Q_{AP} = 8760 E_{dc} \tag{3.14}$$

The PBT variation is dependent on factors such as the type of solar cell, irradiation at the location, capacity of the system and degrading factor of the PV module.

#### **3.6.2** Levelized cost of energy

This is a widely used technique that gives a more accurate energy cost calculation and is defined as the ratio between the LCC of the PV system to the whole life cycle produced energy (LCE). Furthermore, it ascribes all future costs to the present value, resulting in a present price per unit energy value expressed as (Rehman *et al.*, 2007; Kost *et al.*, 2013; Mulligan *et al.*, 2015; Rodrigues *et al.*, 2017; Muslim *et al.*, 2018; Mohammed *et al.*, 2012 and Darling *et al.*, 2011):

$$LCOE = \frac{LCC}{LCE}$$
(3.15)

The *LCE* is calculated using Equation (3.16) (Bhakta and Mukherjee, 2017):

$$LCE = \sum_{i=0}^{n} \frac{AEP \times (1 - f_{PV})}{(1 + r)^{r}}$$
(3.16)

where *AEP* is the expected annual energy produced during the PV system life span and *r* is the discount rate where a value of 7.0% is considered in this study (Economic survey, 2021). Generally the lower the discount rate the smaller the range of LCOE and in extension the PV technology has greatly improved leading to much lower capital costs across all discount rates. As the system time goes by, the output power yield will be degraded with a factor  $f_{PV}$  (maximum value is one) that was used here to get a better energy cost forecast.

# **3.7** Environmental analysis

The environmental impact of the PV system installed on a given site can be defined as the amount of the produced or mitigated  $CO_2$  emissions. In this study, almost entire life cycle of the PV system is considered (except the disposal), where non-renewable energy sources are used in the manufacturing process of the semiconducting materials (Peng *et al.*, 2013). According to Jungbluth, (2005) the environmental analysis is a method that helps quantify and determine the price of  $CO_2$  emissions into the environment for a given system and  $CO_2$  emissions rate is useful index to know to what extent the PV is effective to the global warming. The environmental cost analysis is based on CO<sub>2</sub>emission price and emitted carbon quantity. The amount of CO<sub>2</sub> emitted per month/ annum for a PV system is expressed as (Tiwari *et al.*, 2015; Alsema, 2000):

$$\Phi_{CO_2} = \frac{\Psi_{CO_2} \times E_{overall}}{1000}$$
(3.17)

where  $\Phi_{CO_2}$  is CO<sub>2</sub> emission reduction (tCO<sub>2</sub>/annum),  $\Psi_{CO_2}$  is the average CO<sub>2</sub> equivalent intensity for electricity generation from coal (2.0 kgCO<sub>2</sub>/kWh) and  $E_{overall}$ is the annual overall energy generated from the PV system. The environmental cost is given as (Tiwari *et al.*, 2015):

$$C_{CO_2} = c_{CO_2} \times \Phi_{CO_2}$$
(3.18)

where  $C_{CO_2}$  is the carbon price. From the international carbon price, the average value is 14.5\$/tCO<sub>2</sub> (Den *et al.*, 2011).

## **3.8** Diesel generator fuel consumption model

Diesel generators in many occasions in Kenya are used as primary power source for electrifying a particular location or as backup to the main grid supply in certain periods when there are power outages. The generator's fuel consumption rate  $F_c$  is given by Equation 3.19 as (Akinyele *et al.*, 2014):

$$F_c = XP_{out} + YP_{rated} \tag{3.19}$$

where  $P_{out}$  is the operating output power (kW), X is the generator fuel gradient (typically 0.246 litre/kWh),  $P_{rated}$  is the rated generator power (kW) and Y is the fuel curve intercept coefficient (typically 0.08415 L/kWh) (Oparaku, 2003; Idika, 1995).

#### **3.9 PVSyst Simulation tool**

Theoretical investigation of the PV backup system was performed using a freely available PVSyst 6.7.0 software tool to determine the various PV backup system performance parameters e.g. the system energy yield, system performance ratio (PR), fill factor (FF), capacity utilization factor (CUF) etc. (Belmahdi and Bouardi, 2020; Satish *et al.*, 2020; Shukla *et al.*, 2016). The operating system that the tool runs on must be windows 7 and above (32-bit or 64-bit). It is widely used simulation software for optimization of a PV system under outdoor conditions of a given location and in estimating their performance parameters (Odeh and Nguyen, 2021). Furthermore, the tool enables one to assess the main performance attributes including technical, economic and environmental parameters of the PV systems. Economic evaluation was performed using respective present worth of individual component prices. In addition, a performance comparison is subsequently made between the PV system and diesel generator when each is used as an independent backup system. Figure 3.7 shows a modified schematic of the PV system together with a diesel generator.



Figure 3.7: Modified schematic backup system layout in PVSyst software.

The tool has wide options and built-in features that enable one to input in-situ measured data or use the already existing PVSyst databases (built in Meteo database) e.g. the solar irradiance and ambient temperature (Meteonorm, 2010). The weather dataset used for evaluation is extracted from PVSyst database and has the attributes; solar radiation and ambient temperature. The array was designed based on the architecture design of the house and upon the roof (Figure 3.3). The data input include the characteristic details of the PV modules (type and manufacturer), orientation and the site details which include latitude, longitude, altitude and time zone were entered. Also the load demand per day where power consumed by each electronic device is estimated as hourly load and daily load were keyed in. Hence inverter selection is based on hourly load whereas battery selection is done on daily load (Rekhashree *et al.*, 2018). Output results can be obtained on weekly, daily, or hourly basis depending on the user's needs in form of project details, tables and graphs for performance characterization of the PV system.

#### **CHAPTER FOUR**

#### **RESULTS AND DISCUSSIONS**

#### 4.1 Introduction

This chapter presents technical, economic and environmental performance results and discussions of the installed PV backup system obtained through experimental measurements and simulation using PVSyst software tool. Typical measured data include the solar irradiance, ambient temperature and wind speed which constitute the outdoor operating conditions of the PV system. The measured solar irradiance data were used in the assessment of existing irradiance models that best fit the experimental data for the site and were also used in the characterization of the PV backup system's performance. The technical performance indicators evaluated include the electrical conversion efficiency, fill factor and system energy production of the PV system. The economic and environmental performance comparison results between PV system and DG, when utilized individually as backup power system, are reported in terms of LCC, LCEO and amount of CO<sub>2</sub> avoided respectively.

# 4.2 **Operating conditions**

Data were collected daily for one year in 2020, hence enormous data were recorded, and as such representative results have been presented for each month based on clear sky conditions. The monthly selected days were 21<sup>st</sup> January, 18<sup>th</sup> February, 11<sup>th</sup> March, 16<sup>th</sup> April, 20<sup>th</sup> May, 10<sup>th</sup> June, 14<sup>th</sup> July, 9<sup>th</sup> August, 11<sup>th</sup> September, 16<sup>th</sup> October, 23<sup>rd</sup> November and 18<sup>th</sup> December of the year 2020.

Figure 4.1 shows the monthly representative raw results of the measured solar irradiance.



Figure 4.1: Measured solar irradiance representative results

The graph is a plot of solar irradiance against time (5-minute interval) and the corresponding data are given in Appendix IX. These results show that the solar radiation at the site can be as high as  $1200 \text{ W/m}^2$  with a value of over  $800 \text{ W/m}^2$  being realized as early as from 9 am in the morning up to about 5 pm in the evening throughout the year. In addition, the  $Y_r$  value or in other words the peak solar hour (PSH) obtained from these results is between 4 to 5 hours and is realised between 10:30 am to 3:30 pm, mainly in the dry season in the months of January to March and August to December as indicated in Figure 4.2, which presents clearness index ( $K_T$ ) of the site.

Figure 4.2 shows the monthly clearness index ( $K_T$ ) for the site and the corresponding data are given in Appendix X.



#### Figure 4.2: Monthly clearness index

These results show that the dry season has KT > 0.5 (sunny) and occurs in the months of January to March and August to December while the wet season has  $0.3 \le KT \le$ 0.5 (partly cloudy) in the months of April to July. These results demonstrate that the site is sunny most of the year, eight months to be specific and partly cloudy for three months only, implying optimum operation of the installed PV system. The region, therefore, being in the tropics, has abundant solar radiation throughout the year, and can be harnessed for energy production (electricity and thermal) affordably and reliably in all the seasons.

Figure 4.3 shows the monthly average of ambient temperature and wind speed measured at the site and the corresponding data is given in Appendix XI. From these results, the monthly average ambient temperature varies from 16.3  $^{\circ}$ C to 19.0  $^{\circ}$ C. However, the daily ambient temperature at the site can be as low as 8.8  $^{\circ}$ C in the wet

season and can be as high as  $28.4 \, {}^{0}$ C in the dry season. The monthly average wind speed at the region varies from 1.07 m/s to 3.78 m/s. Low and high wind speeds occur during the wet and dry seasons respectively. Therefore, these wind speeds can be sufficient to provide a natural air cooling of the PV array through the air gap provided for as shown in Figure 3.3.



Figure 4.3: Monthly average ambient temperatures and wind speeds at the site.

The measured solar irradiance data was utilized in the selection of the models that best fit the solar radiation collected at the site. Figure 4.4 present comparison of monthly average solar radiation for the estimated data from the four selected models (based on input parameters that are easily available or determined at the site) and the measured data. These results show that all the models slightly underestimate the amount of solar radiation at the site, hence will present acceptable results when used to estimate solar radiation. For instance, using the estimated data to size any solar conversion device at the site will result to slightly oversized system, which is acceptable with regard to energy production, but then will lead to a higher initial cost of the system, and hence a longer payback period of the system.



# Figure 4.4: Comparison between estimated (by the four models) and measured values of monthly average Daily global solar irradiance at the site.

The sensitivity analysis of how the models predict the solar irradiance at the site was carried out to determine the most accurate model(s) under the local climate of the site. Table 4.1 presents statistical tests of the models based on MAPE and RMSE indicators. From these results, the MAPE test gives average monthly relative uncertainties of 8.5%, 11.0%, 12.6% and 8.9% respectively for Angstrom-Prescott, ASHRAE, Hargreaves-Samanni, and Iqbal models. From the literature, a MAPE value of less than 10% is regarded as having a very good fit and hence provides a very good forecast (Doorga *et al.*, 2019). Based on this criterion, therefore, Angstrom-Prescott is the best model for the site at 8.5%, followed very closely by Iqbal at 8.9%. The other two, however, are outside the good limit and in descending order of uncertainties are ASHRAE and Hargreaves-Samanni models.

Month	Angstrom		ASHRAE		Hargreaves		Iqbal	
	MAPE	RMSE	MAPE	RMSE	MAPE	RMSE	MAPE	RMSE
Jan	8.1	0.217	10.8	0.365	12.7	0.325	9.1	0.187
Feb	8.1	0.521	11.1	0.119	12.9	0.356	8.7	0.822
Mar	8.8	0.317	11.5	1.023	12.8	0.723	8.9	0.124
Apr	8.4	0.117	10.9	1.823	12.9	0.117	8.6	0.562
May	8.6	0.214	10.8	0.256	12.5	0.692	8.9	0.098
Jun	8.9	0.154	11.0	0.113	12.3	0.127	9.2	0.156
Jul	9.6	0.111	10.9	0.21	12.7	0.114	9.1	0.165
Aug	7.9	0.231	11.2	0.112	12.6	0.123	8.6	0.462
Sep	8.6	0.562	11.0	0.145	11.8	1.035	8.8	0.165
Oct	8.2	0.321	11.5	0.732	12.7	0.601	9.1	0.633
Nov	8.8	0.145	11.4	0.561	12.9	0.532	8.7	0.126
Dec	8.0	0.124	10.7	0.085	12.5	0.096	9.0	0.125
Average	8.5	0.252	11.0	0.462	12.6	0.403	8.9	0.302

 Table 4.1: Comparison performance of the four selected models on predicting solar irradiance at the site.

On the other hand, the RMSE test gives standard deviation between estimated and measured values of 0.252, 0.403, 0.462, and 0.302 respectively for Angstrom-Prescott, Hargreaves-Samanni, ASHRAE and Iqbal models. These test results again show that Angstrom-Prescott and Iqbal are the most accurate models for the site because of their lowest standard deviations. However, unlike MAPE test, Hargreaves-Samanni model performs better than ASHRAE model. From these results, the Angstrom-Prescott and Iqbal models give closer prediction, followed respectively by ASHRAE and Hargreaves-Samanni. This means that Angstrom-Prescott and Iqbal models are again

more accurate for estimating the solar radiation for the site. Therefore, from the two tests, the Angstrom-Prescott and Iqbal models are the models that can be used for estimation of solar radiation in the region.

# **4.3** Electrical performance of the PV system

#### 4.3.1 I-V characteristic of PV modules

Figure 4.5 shows typical representative I-V characteristic curves for the selected days for the twelve months of the year 2020 where the corresponding data is given in Appendix XXI. Their respective solar irradiance values are presented for each day in the graph and are instantaneous values recorded at 12.24 pm. From these curves, the  $V_{\text{max}}$  and  $I_{\text{max}}$  points were determined and used to evaluate the PV array efficiency ( $\eta_{array}$ ) and fill factor (*FF*) as presented in equations (3.1) and (3.2) respectively. In addition, the MPP points for the respective days are 122.80 W on22<sup>nd</sup> January, 78.83 W on 13<sup>th</sup> February, 105.99 W on 7<sup>th</sup> March, 104.80 W on 5th April, 108.82 W on 22<sup>nd</sup> May, 79.54 W on 8<sup>th</sup> June, 138.10 W on 12<sup>th</sup> July, 146.76 W on 6<sup>th</sup> August, 119.73 W on 2<sup>nd</sup> September, 138.83 W on 16<sup>th</sup> October, 150.08 W on 3<sup>rd</sup> November and 112.78 W on 18<sup>th</sup> December.

Table 4.2 gives a summary of the experimental representative results obtained from Figure 4.5. The representative values of  $\eta_{array}$  varied from 7.26% on 8<sup>th</sup> February, 2020 to 11.29% on 11<sup>th</sup> March, 2020 while the FF varies from 0.39 on 10<sup>th</sup> June, 2020 to 0.61 on 23<sup>rd</sup> November, 2020. The efficiency results are lower than the STC value, and are expected for outdoor test. In addition, the conversion efficiency fluctuates with the season of the year because of changing operating conditions. On the other hand, the values of the FF are almost one-half of the expected value of one (1),

meaning that the POA solar radiation is not optimized. The possible reason for these low values of FF is that the studied PV system was not installed optimally with regard to tilt angle (should be equal to latitude of the site) and azimuthal angle or orientation.



Figure 4.5: PV module I-V characteristics.
Day (2020)	Solar irradiance $(W/m^2)$	$V_{\max}\left(\mathbf{V}\right)$	$I_{\max}(\mathbf{A})$	FF	$\eta_{array}(\%)$
$\frac{2020}{21^{\text{st}}}$ Jan.	926.4	25.07	4.89	0.48	9.22
18 <sup>th</sup> Feb	645.8	23.25	4.01	0.46	7.26
11 <sup>th</sup> Mar	1026.2	22.36	4.99	0.48	11.29
16 <sup>th</sup> Apr	766.6	29.41	4.12	0.44	9.25
20 <sup>th</sup> May	718.3	28.41	4.25	0.47	9.21
10 <sup>th</sup> Jun	588.5	20.49	3.99	0.39	8.55
14 <sup>th</sup> Jul	722.8	29.47	4.56	0.56	8.11
9 <sup>th</sup> Aug	963.2	30.95	4.85	0.59	9.22
11 <sup>th</sup> Sep	841.9	27.56	4.30	0.44	9.15
16 <sup>th</sup> Oct	931.4	37.12	3.86	0.56	9.45
23 <sup>rd</sup> Nov	997.5	34.92	4.41	0.61	10.54
18 <sup>th</sup> Dec	1099.3	22.51	5.67	0.49	11.02

 Table 4.2: Electrical PV module performance parameters.

#### **4.3.2 Energy Production Performance parameters**

Energy production performance indicators include  $E_{DC}$ ,  $Y_{r}$ ,  $Y_{A}$ ,  $L_{A}$ , PR and CF as defined and discussed in Section 3.5. Table 4.3 gives tabulation of the calculated monthly average values of these quantities.

The experimental results in Table 4.3 show that the  $E_{DC}$  varied from 206.86 kWh to 328.91 kWh and were recorded in the months of June and February respectively with an annual average value of 266.90 kWh while the theoretical results varied from 285.5 kWh in the month of January and 345.2 kWh in the month of October with an annual average value of 312.19 kWh. This shows that the theoretical results are higher than the experimental results, which is attributed to the fact that the former assumes optimized system. However, the deviation between the average values of the experimental and theoretical calculations is ~14.5% which, though slightly high, is acceptable. Form these results, it can be seen that the daily  $E_{DC}$  for the experimental

results is 8.89 kWh and that for theoretical results is 10.41 kWh as compared to daily energy load demand of 2.076 kWh as given in Appendix IV during times of blackouts. Therefore, it can be inferred that the produced  $E_{DC}$  is sufficient to serve the facility loads and in addition the PV system can be used not only as a backup system but also to alleviate part of the grid demand for the facility in the institution.

	E <sub>DC</sub>		Y <sub>r</sub> (kWh/k		Y <sub>A</sub>		L <sub>A</sub>		PR		FF		CF		Efficiency (%)	
Month	(kWh)		W)		(kWh/kWp)		(kWh/kW)		(%)				(%)			
	Exp.	Theo.	Exp.	Theo.	Exp.	Theo.	Exp.	Theo.	Exp.	Theo.	Exp.	Theo.	Exp.	Theo.	Exp.	Theo.
Jan	284.64	285.50	5.64	5.99	4.80	3.60	0.81	0.76	0.769	0.689	0.70	0.68	19.53	18.45	11.95	14.92
Feb	328.91	296.51	6.50	5.95	5.46	3.26	1.04	0.85	0.805	0.659	0.68	0.71	19.45	19.96	13.96	14.56
Mar	310.41	305.72	6.09	5.69	5.10	4.15	0.99	0.72	0.796	0.745	0.70	0.75	19.56	20.95	12.93	14.68
Apr	240.25	300.51	5.11	5.72	4.43	4.15	0.68	0.65	0.728	0.76	0.65	0.65	19.56	21.95	11.64	13.35
May	221.87	345.22	4.97	5.12	4.76	4.22	0.21	0.46	0.730	0.749	0.65	0.68	20.83	21.66	10.95	12.54
Jun	206.86	325.62	4.80	4.98	4.78	5.21	0.12	0.49	0.745	0.75	0.61	0.67	20.55	25.66	10.52	12.85
Jul	217.04	318.60	4.88	4.95	4.72	5.38	0.16	0.38	0.730	0.792	0.62	0.65	18.97	24.99	10.75	12.45
Aug	253.12	302.40	5.38	5.52	5.20	5.02	0.18	0.40	0.720	0.785	0.62	0.64	19.56	24.95	11.04	12.03
Sep	304.25	336.34	5.97	5.68	5.40	4.10	0.57	0.66	0.798	0.745	0.67	0.69	20.33	23.95	12.53	12.17
Oct	295.81	324.50	5.81	5.62	4.09	3.94	0.62	0.72	0.776	0.736	0.65	0.66	18.98	21.65	11.78	12.28
Nov	274.98	309.10	5.59	5.52	5.00	3.64	0.59	0.61	0.784	0.695	0.70	0.69	19.58	18.62	11.22	12.56
Dec	264.76	296.46	5.41	5.84	4.92	3.51	0.47	0.69	0.779	0.730	0.69	0.67	20.67	11.99	11.45	13.95
Yearly	266.90	312.20	5.51	5.55	4.89	4.18	0.54	0.61	0.76	0.736	0.66	0.68	19.80	21.23	11.70	13.19

 Table 4.3: Monthly average values of energy performance parameters.

The experimental values of  $Y_A$  varied from a minimum value of 4.09 kWh/kWp in April to a maximum value of 5.46 kWh/kWp in February with an annual average of 4.89 kWh/kWp. On the other hand, the theoretical values varied from 3.26 kWh/kWp to 5.38 kWh/kWp with an annual average of 4.182 kWh/kWp. From these results, the

number of hours per year at which the system effectively operates at its rated power was ~4 hours.

The experimental and theoretical annual average values of L<sub>A</sub> were 0.52 kWh/kW and 0.61 kWh/kW respectively. These results take into account all the system losses which include the PV loses due to irradiance level and ambient temperature, mismatch losses, ohmic wiring losses, unused energy losses, converter losses, battery efficiency losses and charging/discharging losses. They range from the characteristics of the modules themselves (e.g., tilt and orientation angles) which can be rectified and environmental factors (e.g., shading and soiling). All these losses have an impact on efficiency and output energy of a PV system. To minimize or avoid the system energy losses, the institution should carry out preventive maintenance periodically to detect and avoid energy losses due to failures from components such as the inverter and the batteries. The theoretical losses (source and quantities) are given in a loss diagram presented in Appendix XVI.

The experimental values of  $Y_r$  vary from 4.8 kWh/kW to 6.5 kWh/kW which were realized in the months of February and June respectively with an average annual value of 5.513 kWh/kW. On the other hand, the theoretical values of  $Y_r$  varied between 4.95 kWh/kW to 5.99 kWh/kW with annual average value of 5.548 kWh/kW. Thus, the average annual values for both experimental and theoretical results are relatively close and agree with similar studies carried out in Kenya that gave a value of 5.5 kWh/kW (Oloo *et al.*, 2015 and Kiplagat *et al.*, 2011). In other words, these results imply that the daily site peak sun hours were slightly over 5 hours and can be improved with optimal orientation of the PV array. The experimental values of PR varied from 72% in the month of August to 80.5% in the month of February with an average annual value of 76.3% while the theoretical values varied from 65.9% in the month of February to 79.2% in the month of July with an average annual value of 73.6%. This means that ~25% of the incident solar energy is not converted into usable energy which indicates a possible fault in components, conduction losses and thermal losses. The obtained PR value acts as an indicator and can prompt more detailed inspection of the PV plant so that, for example, soiling of the PV modules is removed or defective BOS components can be repaired or replaced. Both values are within the agreed range in literature (e.g., Rekhashree *et al.*, 2018; Seme *et al.*, 2019 and Znidarec *et al.*, 2019). The corresponding theoretical PR data are given in Appendix XIV.

The solar PV backup system has a CF of 19.80% whereas PVsyst modeled a CF of 21.23%. Fixed-tilt PV systems (the case for the studied system) are expected to have CF values between 20.8% and 26% in high solar radiation regions according to literature (e.g., Gottschalg *et al.*, 2005; Dena *et al.*, 2019 and Ding *et al.*, 2019), hence the obtained results compare well with the reported range. From these results, generally location has got an impact on CF; the site has very good solar energy potential sufficient for exploitation for electricity generation.

# 4.4 Economic performance

The economic performance methods that were implemented were the LCC, PBT and LCOE for PV system and the diesel generator. The PV system and the DG cost parameters that were considered in this study are given in Appendices VI and VII respectively. Assessment period was based on the long-lasting component, which is the PV module in this case. The expected life span of the PV systems is reported in

literature and also by manufacturers to be between 25 to 30 years, but in this study, a period of 25 years was used.

Figure 4.6 presents LCC comparison between the PV backup system and DG over the PV modules lifetime of 25 years.



Figure 4.6: PV backup system versus diesel generator life cycle costs

From these results, the DG has low initial capital cost as shown in year one (2016) and is as low as ~ 27% of the value obtained for the PV system. The PV system high initial costs are constituted in a large percentage by the PV modules, inverter/charge controller unit and battery bank costs. These costs can be reduced by lowering the battery capacity since many loads are not classified as critical loads and can be reduced during times they are not necessarily in use. The relatively higher LCC of the DG which is more than twice that of the PV backup system is attributed to the cost of fuel and routine maintenance while the PV system is characterized by low operation and maintenance cost over the 25 years project life. It can be seen from Figure 4.6 that there is gradual increase in LCC of the DG with time. This is attributed to yearly rise

in the cost of fuel as a result of inflation and routine maintenance of the diesel generator.

Table 4.5 gives the economic comparative performance results based on LCC, PBT and LCOE analysis of the PV backup system and that of an equivalent diesel generator. The PV arrays normally do not require any maintenance over their lifetime apart from module surface cleaning. For the studied PV system natural cleaning through rain is sufficient to remove any accumulated dust and other particles that may fall on the PV array surface. However, the batteries, inverter/charge controller unit require replacement and maintenance regularly. According to literature, the total operation and maintenance (O & M) cost for a PV system is 2% of PV array cost per year (e.g., Shah *et al.*, 2018 and Shaw-Williams *et al.*, 2018). Therefore, the O & M costs for the PV system under study with a lifespan of 25 years was \$433.00or \$17.32 yearly. On the other hand, the O & M cost for the diesel generator with lifespan of 5 years was \$5220.25 or \$26250.35 over the PV array lifespan assuming that it operates 2 hours per day of blackouts.

The total replacement cost for the PV system was \$5240, and out of these costs, the batteries' replacement cost was 91.4% of the total costs because they are replaced four times over the system lifespan and the remaining 8.6% is the replacement cost for inverter/charge controller unit. The diesel generator replacement costs over 25 years of the PV system life span with four times replacements was \$6482.60 or \$259.30 annually.

Parameter	PV system	Diesel generator		
Initial cost	\$1774.02	\$494.55		
Power rating of the facility	63.145 kWh/year	63.145 kWh/year		
O & M cost	\$433	\$26250.35		
Replacements	\$5240	\$6482.60		
Price of fuel	-	1.0 \$/L		
Annual fuel consumption	-	\$253.20		
Total electrical units produced in one Month	266.90 kWh,	318.21 kWh,		
LCC	\$3057.93	\$7792.75		
PBT	6.38 years	4.25 years		
LCOE	0.045 \$/kWh	0.325 \$/kWh		

Table 4.4: Summary of PV system versus diesel generator economic performance

It can be seen from the Table 4.4 that the cost of energy (determined using equation (3.15) for the PV system was 0.045 \$/kWh which was far much cheaper compared to that of the diesel generator which was 0.325 \$/kWh. These results represent the average revenue per unit of electricity generated where the PV backup system LCOE value was 12.2% lower than that of the diesel generator which makes PV backup power a competitive alternative as a primary source of energy during power outages in comparison with other technologies.

The PBT for the PV backup system was determined using equation (3.13) and results give 6.38 years with a lifetime of 25 years compared to that of the diesel generator with a lifetime of 5 years and a PBT of 4.25 years. Improvements on individual component conversion efficiencies will reduce the PBT of PV systems which will eventually improve the LCOE value. The LCOE values for the PV system and diesel generator were 0.045 \$/kWh and 0.325 \$/kWh respectively. This is a clear indication that the cost of solar power as a backup system is cheaper by a factor of seven than that of the diesel generator when used as a backup system.

From these economic performance results, as evident from all the considered parameters, the PV backup system is more cost effective as a long-term backup power supply system. The economic benefits of the PV system assume that the price of electricity remains constant (except for inflation) throughout the analysis period. In the long term, deployment of PV backup systems will lead to decrease in price of electricity, which in turn will spur economic growth and completely replace deployment of expensive diesel generators as power backup option.

# 4.5 Environmental performance of the PV system

Figure 4.7 presents the experimental and theoretical (corresponding data is given in Appendix XII) monthly variation of  $CO_2$  avoided by use of PV system as a backup system and the environmental cost parameter of the PV system. The amount of  $CO_2$  emission avoided and their costs were determined using equations (3.17) and (3.18) respectively. These results only consider the emissions avoided during PV array operation stage and exclude production, transportation and end-of-life process.



Figure 4.7: Monthly carbon reduction and environmental cost

It is seen from Figure 4.7 that mitigation of CO<sub>2</sub> emission rate varies from the minimum of 0.48 tCO<sub>2</sub>/month in the month of June to a maximum of 0.67 tCO<sub>2</sub>/month in the month of February. Generally, high insolation months (i.e., January to March and then September to December) give the highest amount of CO<sub>2</sub> avoided while low insolation months (i.e., April to August) give the lowest amount of CO<sub>2</sub> mitigated. On the other hand, theoretical results show that the amount of CO<sub>2</sub> emission avoided varies from 0.517 tCO<sub>2</sub>/month in the month of July to 0.79 tCO<sub>2</sub>/month in the month of February. The total annual amount of CO<sub>2</sub> emission avoided when PV is used instead of DG power backup system are 5.84 tCO<sub>2</sub>/year and 7.95 tCO<sub>2</sub>/year for experimental and theoretical results show that use PV backup systems have considerable potential for reduction of GHGs emissions which may lead to saving on fossil fuel use and its importation. In addition, the amount of CO<sub>2</sub>avoided

is a factor of solar radiation received at a given location hence, reduction in the emissions of  $CO_2$  signifies an environmental improvement when a PV backup system is used as a substitute to diesel backup generators.

The experimental cost parameter of the avoided CO<sub>2</sub> emissions (i.e., the cost of avoided CO<sub>2</sub> emissions) varies from a minimum of \$6.95 in the month of June to a maximum of \$9.72 in the month of February with average annual value of \$9.61 as shown in Figure 4.7. The theoretical cost parameter varied from \$8.27 to \$11.46 with an average annual value of \$116.18. This is the amount of money that one should be paid as compensation for using a PV system instead of a DG as a backup system. In addition, these findings can provide policymakers with useful information for both evaluating and planning the PV power plant-related policies and to deal with the environmental damage that is caused by use of DGs as backup systems.

# **CHAPTER FIVE**

# CONCLUSIONS AND RECOMMENDATIONS

# 5.1 Conclusions

The results show that the site is in a sun-belt region with high solar irradiance throughout the year and therefore ideal for PV applications. The region registers as high as 800 W/m<sup>2</sup> as early as 9:00 am and can be as high as 1200 W/m<sup>2</sup> on clear sky days. The site has high solar irradiance during the months of January to March and August to December and relatively low during the months of April to July with average  $Y_r$  or PSH of ~ 4 -5 hours. The MAPE test on the performance of the four models selected gave absolute uncertainties of 8.5%, 11.0%, 12.6% and 8.9% while RMSE test gave standard deviations of 0.252, 0.463, 0.403 and 0.302 respectively for the Angstrom-Prescott, ASHRAE, Hargreaves-Samanni and Iqbal models. These results show that the models are suitable for the site and have relatively good estimation of the monthly average of daily global solar radiation, but Angstrom-Prescott and Iqbal models predict the monthly average solar radiation with higher accuracy than the other two models for this site.

The electrical performance of the PV system showed that the PV system annual experimental efficiency was 11.71% with monthly range of 10.52% to 13.96% while the simulation results annual efficiency was 13.19% with monthly range of 12.03% to 14.92%. The PV backup system is not operating optimally since the experimental and the simulation FF values are << 1 with annual values of 0.66 and 0.68 respectively. Theselow efficiency and FF values are attributed to non-optimized installation of the PV array with regard to tilt angle (should be equal to latitude of the site) and orientation and losses from BOS components and degradation. The annual effective

energy output of the PV backup system was 3412.94 kWh where the month of February gave the highest value of 328.91 kWh and the month of June produced the lowest value of 206.63 kWh. The annual average array yield was 4.92 kWh/kW whereas the array losses average annual value was 0.54 kWh/kW. The average yearly reference yield was 5.51 kWh/kW, annual average PR was 76.0% and annual capacity factor of 19.80% was attained. On the other hand, simulation results showed that annual PV system PR was 73.33% and the total produced energy was 4615 kWh/year. These results indicate that the system performance is adequate with regard to yield and PR values. All the electrical performance parameters considered, PV system has showed potential and proved its reliability to meet the energy demands of the KTTC social hall during periods of power outages.

The results show that the DG has low initial capital cost and is as low as ~ 27 % of the value obtained for the PV system. The O & M cost for the PV system was \$433.00 or \$17.32 yearly over its lifetime while that of the DG with lifespan of 5 years was \$5220.25 or \$26250.35 over the PV array lifespan assuming that it operates 2 hours per day of blackouts. Total replacement costs are \$5240 with batteries' value taking 91.4% of the total PV system replacement costs. It is further shown that the investment will be recovered in 6.38 years for the PV system and 4.25 years for the DG. The LCC value of the DG is \$7792.75 and that of the PV system is \$3057.93hence, the DG value is more than twice that of the PV system due to the cost of fuel and routine maintenance. The experimental and simulation results showed that LCOE values are 0.045 \$/kWh and 0.059 \$/kWh respectively for the PV system which indicates that the cost of solar power is far much below the cost of electricity from the DG with annual LCOE value of 0.325 \$/kWh. These economic indicators show that the PV backup system is worthy of investment compared to diesel backup

generators which have for long been the traditional way to backup grid electricity sources.

Mitigation of  $CO_2$  emission rate varied from minimum of 0.48 t $CO_2$ /month to a maximum of 0.67 t $CO_2$ /month. The emissions are taken as the quantity of  $CO_2$  that can be saved by use of PV system where cost parameters vary from minimum of \$6.95 to a maximum of \$9.72. This shows that the PV system can contribute significantly to the mitigation of  $CO_2$  emission. In addition, the total amount of carbon dioxide emissions that are directly saved is 5.3 t $CO_2$  equivalent to 0.44 t $CO_2$ /month with LCE value of 334 Kg $CO_2$ /module and a subtotal of 1336 Kg $CO_2$  for the four modules. The PV power presents net benefits over diesel power in all economic performance indices evaluated, and hence can be used as a reliable and affordable replacement for DG backup systems.

# 5.2 **Recommendations**

- 1. The site is adequate for harnessing PV power as the primary source of energy since the region has abundant solar radiation throughout the year. For optimal performance realization, the PV modules should be installed properly for optimum operation through orientation and tilt angles adjustments. The PV backup systems should be used as an alternative to DGs because PV powered system have economic advantage after 6.38 years of installation compared to the DG. By taking PV systems as backup systems instead of DG will contribute to increased use of renewable energy.
- 2. Angstrom-Prescott and Iqbal models are recommended for the site because they estimate the solar radiation data which are closer to the measured values than the ASHRAE and Hargreaves-Samanni models.
- 3. There is need to set a national degradation test facility for further research to investigate the effect of degradation on PV system performance. In addition, a test bed of a PV backup system can be set up to have the test for various PV technologies under uniform climatic conditions.

# LIST OF REFERENCES

- Abawi, Y; Rennhofer, M.; Berger, K.; Wascher, H. and Aichinger, M. (2016). Comparison of theoretical and real energy yield of direct DC- power usage of a Photovoltaic Façade system. *Renew. Energy*; 89, 616–626.
- Abdelhafez, M. H. H., Touahmia, M., Noaime, E., Albaqawy, G.A., Elkhayat, K., Achour, B. and Boukendakdji, M. (2021). Integrating Solar Photovoltaics in Residential Buildings: Towards Zero Energy Buildings in Hail City, KSA. Sustainability, 13, 1845.
- Abd El-Shafy, A.N. (2009). Design and Economic Analysis of a Stand-Alone PV System to Electrify a Remote Area Household in Egypt. *The Open Ren. Energy Journal*; 2, 33-37.
- AbdelHady, R. (2017). Modeling and simulation of a micro grid-connected solar PV system. *Water Science*; 31(1), 1–10.
- Abdullah, G., Nishimura, H and Fujita, T. (2021). An Experimental Investigation on Photovoltaic Array Power Output Affected by the Different Partial Shading Conditions. *Energies*; 14, 2344; https://doi.org/10.3390/en14092344.
- Aguilera J. and Espinoza R. (2019). Feasibility evaluation of residential photovoltaic self-consumption projects in Peru. *Renewable Energy*, 136; 414–427.
- Akarslan, F. (2012). Photovoltaic systems and applications, modeling and optimization of renewable energy systems, *InTech*.
- Akinyele D. O., Rayudu R. K., Chakrabarti B and Nair N. K. C. (2014). Decentralized benefits assessment, In Proc. IEEE/PES Innovative Smart Grid Technologies (ISGT) Conference: 1-4.
- Akinyele. D. O and Rayudu. R. K. (2016). Techno-economic and environmental performance analyses of a solar photovoltaic microgrid (SPM) system for developing countries", *Energy Conversion and Management*. 124: 266-279.
- Akinyele D. O, Rayudu R. K and Nair N. K. C. (2015). Global progress in photovoltaic technologies and the scenario of solar panel plant and module performance estimation-application in Nigeria. *Renewable and Sustainable Energy Reviews*, 48: 112-139.
- Akinyele D. O, Rayudu R. K and Nair N. K. C. (2015). Development of photovoltaic power plant for remote residential applications: the socio-technical and economic perspectives, *Applied Energy*, 155: 131-149.
- Akinyele D. O and Rayudu, R.K. (2014). Review of energy storage technologies for sustainable power networks, Sustainable Energy Technologies and Assessments, 8: 74-91.

- Al-Aboosi, F.Y. (2020). Models and hierarchical methodologies for evaluating solar energy availability under different sky conditions toward enhancing concentrating solar collectors use: Texas as a case study. *Int. J. Energy Environ. Eng*; 11, 177–205.
- Al-Aboosi, F.Y and Al-Aboosi, A.F. (2021). Preliminary Evaluation of a Rooftop Grid-Connected Photovoltaic System Installation under the Climatic Conditions of Texas (USA).*Energies*; 14, 586.
- Al-Addous M., Dalala Z., Class C.B., Alawneh F. and Al-Taani H. (2017). Performance analysis of off-grid PV systems in the Jordan valley. *Renewable Energy*, 113, 930-941.
- Allouhi, A.; Saadani, R.; Kousksou, T.; Saidur, R.; Jamil, A. and Rahmoune, M.
   (2016). Grid-connected PV systems installed on institutional buildings: Technology comparison, energy analysis and economic performance. *Energy Build*, 130, 188–201.
- Almarshoud, A. (2017). Technical and Economic Performance of 1MW Gridconnected PV system in Saudi Arabia. *Int. J. Eng. Res. Appl*; 7, 9–17.
- AfDB I SE4All Africa Hub Annual Report 2014 2015.
- Ali H.A, Al-Waeli, K. Sopian, Hussein A. Kazem, Miqdam T. and Chaichan. (2017). Photovoltaic/Thermal (PV/T) systems: Status and future prospects, *Renewable and Sustainable Energy Reviews*; 77; 109-130.
- Al-Kayiem, H. and Mohammad, S. T. (2019). Potential of renewable energy resources with an emphasis on solar power in Iraq: An outlook. *Resource*; 8, 42.
- Allouhi A., Saadani R., Buker M.S., Kousksou T., Jamil A. and RahmouneM.(2019). Energetic, economic and environmental (3E) analyses and LCOE estimation of three technologies of PV grid-connected systems under different climates.*Solar Energy*; 178, 25-36.
- Almarshoud, A. (2017). Technical and Economic Performance of 1MW Gridconnected PV system in Saudi Arabia. *Int. J. Eng. Res. Appl*; 7, 9–17.
- Alpesh D, Vatsal S, Indrajit M and Abhijit R. (2020). Annual performance of Multiple MPPT based String Inverter under Partial Shadowing: Observations at Utility scale Solar PV Plants.47th IEEE Photovoltaic Specialists Conference (PVSC), June 2020, India.
- Al-Sanea S. A., Zedan M. F. and Al-Ajlan S. A. (2004). Adjustment factors for the ASHRAE clear-sky model based on solar-radiation measurements in Riyadh. *Applied Energy*. 79, 215-237.
- Alsema E.A. (2000). Energy pay-back time and CO<sub>2</sub> emissions of PV systems. *Prog. Photovolt: Res. Appl*; 8 (1): 17-25.

- AlYahya, S. and Irfan, M.A. (2016). The techno-economic potential of Saudi Arabia's solar industry. *Renew. Sustain. Energy Rev*; 55, 697–702.
- Ammu, S.J., Jan, D., Ajit, P.A., Rangan, B and Prakash C.G. (2017). Cost and Energy analysis of PV battery grid backup system for a residential load in urban India. *Energy Procedia*, 118, 88–94.
- Ani, V.A. (2016). Design of a Reliable Hybrid (PV/Diesel) Power System with Energy Storage in Batteries for Remote Residential Home. *Journal of Energy*, 1-16.
- Anis W.R, Mertens R.P and van Overstraeten R.J. (1983). Calculation of solar cell operating temperature in a flat plate PV array. *Proc. 5th European Photovoltaic Solar Energy Conf., Athens, Greece;* 520 - 524.
- Ångström A. (1924). Solar and terrestrial radiation.*Q. J. R. Meteorol. Soc.* 50; 121-125.
- Archer, M. D and Hill, R. (2001). Clean Electricity from photovoltaics. Series of photo conversion of solar energy; ISBN1-86094-161-3.Imperial College Press, 2001.
- Ascencio-V'asquez, J., Bevc, J., Reba, K., Brecl, K., Jankovec, M and Topi<sup>°</sup>c, M. (2020). Advanced PV Performance Modeling Based on Different Levels of Irradiance Data Accuracy. *Energies*; 13(9), 21-66.
- Assad, A. (2010). A Stand-Alone Photovoltaic System, Case Study: A Residence in Gaza. J. of Applied Sciences in Environmental Sanitation, 5 (1): 81-91.
- Awasthi, J. and Poudyal, K.N (2018). Estimation of Global Solar Radiation Using Empirical Model on Meteorological Parameters at Simara Airport, Bara, Nepal. *Journal of the Institute of Engineering*, 14 (1): 143-150
- Aydin, E.; Allen, T.G.; De Bastiani, M.; Xu, L.; Ávila, J.; Salvador, M.; Van Kerschaver, E. and De Wolf, S. (2020). Interplay between temperature and bandgap energies on the outdoor performance of perovskite/silicon tandem solar cells. *Nat. Energy*; 5, 851–859.
- Ayompea, L., Duffya, A., McCormackb, S. and Conlonc, M. (2011). Measured Performance of a 1.72 kW Rooftop Grid Connected Photovoltaic System in Ireland. *Energy Conversion and Management*, 52, 816-825.
- Babajide, A. and Brito, M. C. (2021). Solar PV systems to eliminate or reduce the use of diesel generators at no additional cost: A case study of Lagos, Nigeria, *Renewable Energy*, 172(3), 209-218.

- Bai, A, Popp. J., Balogh P., Gabnai Z., Plyi B., Farkas I., Pinter G. and Zsiboracs H. (2016). Technical and economic effects of cooling of monocrystalline photovoltaic modules under Hungarian conditions; *Renewable and Sustainable Energy Reviews*. 60; 1086-1099.
- Bajpai U. and Yadav S. (2018). Performance evaluation of a rooftop solar photovoltaic power plant in Northern India. *Energy Sustain. Dev.* 43, 130-138.
- Bakirci, K. (2009). Estimation of Solar Radiation by Using ASHRAE Clear-Sky Model in Erzurum, Turkey. Energy Sources, Part A: Recovery, Utilization, and Environmental Effects. 31, 208 - 216.
- Bandou, F., Hadj Arab, A and Belkaid, M.S. (2013). Test performance electrical of the photovoltaic module in two different environments. *Energy Procedia*, 36, 1018-24.
- Bandyopadhyay, A., Bhadra, A., Raghuwanshi, N.S. and Singh, R. (2008). Estimation of monthly solar radiation from measured air temperature extreme. *Agricultural and Forest Meteorology*; 148; 1707–1718.
- Barnes, D and Foley, G. (2004). Rural Electrification in the Developing World: A Summary of Lessons from Successful Programs, World Bank, Washington DC.
- Basharat, J. and Mohd, M.K. (2014). Estimation of Clear-Sky Solar Radiation Using ASHRAE Model for Aligarh, India.*International Journal of Engineering Research and Technology*. 7, (3), 227-236.
- Bashir, N., Mohammed, Y. S. and Mustafa, M.W. (2014). Energy storage systems for renewable energy power sector integration and mitigation of intermittency. *Renewable and Sustainable Energy Reviews*; 35, 499-514.
- Belmahdi, B. and Bouardi, A.E. (2020). Solar Potential Assessment Using PVsyst Software in the Northern Zone of Morocco. *Procedia Manuf*; 46, 738–745.
- Beyer, H.G., Yordanov, G. H., Midtgård, O.M., Saetre, T. O. and Imenes, A. G. (2011). Contributions to the knowledge base on PV performance: Evaluation of the operation of PV systems using different technologies installed in southern Norway. *Conference Paper in Conference Record of the, 38<sup>th</sup>IEEE Photovoltaic Specialists Conference*, USA, June 2021.
- Bhakta.S and Mukherjee.V. (2017). Performance indices evaluation and techno economic analysis of photovoltaic power plant for the application of isolated India's island. *Sustainable Energy Technologies and Assessments*; 20: 9-24.

- Bhandari K. P., Jennifer M.C., Randy J.E. and Defne S.A. (2015). Energy payback time (EPBT) and energy return on energy invested (EROI) of solar photovoltaic systems: A systematic review and meta-analysis. *Science Direct*, 47(7), 133-141.
- Bhattacharyya, S.C. and Debajit, P. (2016). Mini-grid based off-grid electrification to enhance electricity access in developing countries: What policies may be required? *Energy Policy*, 94, 166–178.
- Bhowmik, H. and Amin, R. (2019). Performance Analysis of PV Cells under Monsoon Climate. *IAPE '19, Oxford, United Kingdom conference paper*, ISBN: 978-1-912532-05-06.
- Bhuvaneswari C., Vijay B. and Natarajan P. (2018). Estimation and Performance Analysis of a 15kW Off-Grid Solar PV System. *International Journal of Engineering & Technology*, 7(2): 143-147.
- Bocca, A.; Bergamasco, L.; Fasano, M.; Bottaccioli, L.; Chiavazzo, E.; Macii, A. and Asinari, P. (2018). Multiple-Regression Method for Fast Estimation of Solar Irradiation and Photovoltaic Energy Potentials over Europe and Africa.*Energies*.11, 3477.
- Bosch J. L., Batlles F. J., Zarzalejo L. F. and López G. (2010). Solar resources estimation combining digital terrain models and satellite images techniques. *Renewable Energy*; 35, (12), 2853-2861.
- Boyle G. (2004). Renewable Energy: Power for a Sustainable Future, 2<sup>nd</sup> ed. Oxford, UK: Oxford University Press.
- Braker, K.; Pathak, M.; Pearce, J. (2011). A review of solar photovoltaic levelized cost of electricity. *Renew. Sustain. Energy Rev*;15, 4470–4482.
- Branker. K; PathakM. J.M. and Pearce. J. M. (2011). A Review of Solar Photovoltaic Levelized Cost of Electricity", *Renewable & Sustainable Energy Reviews*; 15, 4470-4482.
- Burger, B. and Rüther, R. (2006). Inverter sizing of grid-connected photovoltaic systems in the light of local solar resource distribution characteristics and temperature, *Sol. Energy*, 80, 32–45.
- Burke A.F. (2007). Batteries and ultra-capacitors for electric, hybrid, and fuel cell vehicles. *In: Proceedings of the IEEE*; 95(4): 806 -820.

- Burzynski, R., Crane, M. and Yao, R. (2010). A review of domestic hot water demand calculation methodologies and their suitability for estimation of the demand for Zero Carbon houses: Proceedings of Conference: TSBE EngD Conference, TSBE Centre, University of Reading. Whiteknights Campus, RG6 6AF, 6<sup>th</sup> July 2010.
- Butti.K and Perlin .J. (1980). A golden thread 2500 years of solar architecture and technology", *ed. Marion Boyars, London.*
- Carr, A.J. and Pryor, T.L. (2004). A comparison of the performance of different PV module types in temperate climates. *Solar Energy*; 76, 285-294.
- Castaner, L. (2002). Modeling Photovoltaic Systems Using PSpice. CRC Press.
- Cebecauer T., Šúri M. and Perez R. (2010). High performance MSG satellite model for operational solar energy applications. ASES National Solar Conference, May 2010, Phoenix, USA.
- Cebecauer T., Šúri M. and Guyemard C.H. (2011). Uncertainty Sources in Satellitederived Direct Normal Irradiance: How can prediction accuracy be improved globally? Proceedings of the Solar PACES 2011 Conference, September 2011, Granada, Spain.
- Chanchangi, Y.N.; Ghosh, A.; Sundaram, S. and Mallick, T.K. (2020). Dust and PV Performance in Nigeria: A Review. *Renew. Sustain. Energy Rev.* 121, 109704.
- Chantana, J., Tai, K., Hayashi, H., Nishimura, T., Kawano, Y. and Minemoto, T. (2020). Investigation of carrier recombination of Na-doped Cu2SnS3 solar cell for its improved conversion efficiency of 5.1%, *Solar Energy Materials and Solar Cells*, 206, ISSN 0927-0248,
- Chen, H., Cong, T. N., Yang, W., Tan, C., Li, Y and Ding Y. (2009). Progress in electrical energy storage system: A critical review. *Progress Nat Science*, 19:291-312.
- Chen H, Zhang X, Liu J and Tan C. (2013).Compressed air energy storage. *InTech*; 101-112.
- Cherfa, F.; Arab, A.H.; Oussaid, R.; Abdeladim, K. and Bouchakour, S. (2015). Performance Analysis of the Mini-Grid Connected Photovoltaic System at Algiers. *Energy Proc*; 83, 226–236.
- Chioncel, C.P.; Kohake, D.; Augustinov, L.; Chioncel, P. and Tirian, G.O. (2010). Yield factors of a photovoltaic plant. *Acta Tech. Corviniensis Bull. Eng.* (*Fasc.*); 2, 63–66.

- Choubey, P.C., Oudhia, A. and Dewangan, R. (2012). A review: Solar cell current scenario and future trends. *Recent Res. Sci. Technol*; 4, 99–101.
- Chow.T.T. (2010). A review on photovoltaic/thermal hybrid solar technology. *Applied Energy*, 87(2), 365-379.
- Cissokho, L and Seck, A. (2013). Electric power outages and the productivity of small and medium enterprises in Senegal. *Investment Climate and Business Environment Research Fund Report*, 77, 13.
- Collado, J and Guallar, J. (2013). A review of optimized design layouts for solar power tower plants with campo code. Renew. *Sustain. Energy Rev*, 20, 142 154.
- Congedo P. M., Malvoni M., Mele M. and De Giorgi M. G. (2013). Performance measurements of monocrystalline silicon PV modules in south-eastern Italy. *Energy Conversion and Management;* 68, 1-10.
- Conibeer, G. (2007). Third- generation photovoltaics. Mater. Today. 10, 42-50.
- Cornaro, C. and Musella, D. (2010). Performance Analysis of PV Modules of Various Technologies After More than One Year of Outdoor Exposure in Rome, Italy. *Solar Energy*, 88 (6), 421-428.
- Cucumo.M., De Rossa. A., Ferraro. V., Kaliakatkos.D. and Marinelli.V. (2007). Experimental testing of models for the estimation of hourly solar radiation on vertical surfaces at Arcavacata di Rende. *Solar Energy*, 81 (5), 692-695.
- Da Silva. P, Maurício. B, Escobedo. J. F, João. S, Cicero. R, Taiza. M, Gorla. S and Sílvia.H. (2017).Performance of the Angstrom-Prescott Model (A-P) and SVM and ANN Techniques to estimate the daily Global Solar Irradiation in Botucatu/ SP/Brazil. *Journal of Atmospheric and Solar-Terrestrial Physics*; 160, 11–23.
- D'Orazio M., Di Perna C. and Di Giuseppe E. (2014). Experimental operating cell temperature assessment of BIPV with different installation configurations on roofs under Mediterranean climate. *Renewable Energy*. (68), 378-396.
- Dadjé, A., Djongyang, N. and Tchinda, R. (2017). Electrical Power Losses in a Photovoltaic Solar Cell Operating under Partial Shading Conditions. *Journal* of Power and Energy Engineering, 5, 19-33.
- Dahmani K, Notton G, Voyant C, Dizene R, Nivet M.L and Paoli C. (2016). Multilayer Perceptron approach for estimating 5-min and hourly horizontal global irradiation from exogenous meteorological data in locations without solar measurements. *Renew Energy*; 90:267-282.

- Dally, E. (2013). Life cycle cost analysis of a stand-alone PV system in rural Kenya. M.Scdise, Indiana.
- Danandeh, M. A. and Mousavi. S. M. G. (2018). Comparative and comprehensive review of maximum power point tracking methsertation, Purdue University, West Lafayettods for PV cells. *Renew. Sustain. Energy Rev*, 82, 2743-2767.
- Darling S. B., You F., Veselka, T. and Velosa A. (2011). Assumptions and the levelized cost of energy for photovoltaics. *Energy & Environmental Science*; 4, 3133-3139.
- Das, S., Sudhagar, P., Kang, Y.S. and Choi, W. (2014). Graphene synthesis and application for solar cells. J. Mater. Res. 29, 299–319.
- Debbarma, M. Sudhakar, K and Baredar, P. (2017). Thermal modeling, exergy analysis, performance of BIPV and BIPVT, *A review, Renew. Sustain. Energy Rev.* 73, 1276–1288.
- Dena, H.; Sinaga, E. and Djajadi, A. (2019). Experimental Testbed and Performance Evaluation for Rooftop Solar PV System and Generator Set. In Proceedings of the Advances in Science and Engineering Technology International Conferences (ASET), Dubai, UAE; 1–8.
- Despotovic, M.; Nedic, V.; Despotovic, D. and Cvetanovic, S. (2016). Evaluation of empirical models for predicting monthly mean horizontal diffuse solar radiation. Renew. Sustain. *Energy Rev.* 56, 246–260.
- De Lima, L. C., De Araújo Ferreira, L., and De Lima Morais, F. H. B. (2017).
   Performance analysis of a grid connected photovoltaic system in northeastern Brazil. *Energy for Sustainable Development*; 37, 79–85.
- Díez-Mediavilla, M., Alonso-Tristán, C., Rodríguez-Amigo, M., García-Calderón, T. and Dieste-Velasco, M. (2012). Performance analysis of PV plants: Optimization for improving profitability. *Energy Convers. Management*; 54, 17–23.
- Ding, K.; Feng, L.; Zhang, J.; Chen, X.; Chen, F. and Li, Y. (2019). A Health Status-Based Performance Evaluation Method of Photovoltaic System. *IEEE Access*; 7, 124055–124065.
- Doorga, J.R.S.; Rughooputh, S.D.D.V. and Boojhawon, R. (2019).Modeling the global solar radiation climate of Mauritius using regression techniques. Renew. Energy, 131, 861–878.
- Duffie J.A. and Beckman W.A. (2013). Solar Energy of Thermal Process, 2nd edition. *Wiley & Sons, New York.*

- Economic survey. (2021). Kenya National Bureau of Statistics; ISBN: 978-9966-102-30-0, Available at; https://www.knbs.or.ke; Accessed on 25<sup>th</sup> Jan 2021.
- El-Adawi, M.K. and Al-Nuaim, I.A. (2007). The Temperature Functional Dependence of VOC for a Solar Cell in Relation to Its Efficiency New Approach. *Desalination*, 209, 91–96.
- El Chaar L, lamont L. A and El Zein N. (2011). Review of photovoltaic technologies. *Renewable and Sustainable Energy Reviews*, 15:2165-75.
- Elhodeiby, A.S.; Metwally, H.M.B and Farahat, M.A. (2011). Performance analysis of 3.6 kW rooftop grid connected photovoltaic system in Egypt. *International Conference on Energy Systems and Technologies (ICEST 2011)*: 151–157.
- Emilio M, Giovanni P, and Giovanni S. (2014). A two steps algorithm improving the P&O steady state MPPT efficiency. *Appl. Energy*; 113, 414-421.
- Emziane, M. and Al Ali, M. (2015). Performance Assessment of Rooftop PV Systems in Abu Dhabi. *Energy Build*. 2015, 108, 101–105.
- Energy Regulatory Commission. (2018). Least Cost Power Development Plan 2017 2037; Energy Regulatory Commission: Nairobi, Kenya.
- Energy, R. Renewable Energy Policies in a Time of Transition. (2018). Available online:http://energyaccess.org/wpcontent/uploads/2018/04/IRENA\_IEA\_REN 21\_Policies-2018.pdf, Accessed on 02/05/2021.
- Enerray. (2017). The advantages of rooftop solar systems. http://www.enerray.com/ blog/ advantages- rooftop- solar- systems.
- Enterprise Survey. The World Bank. (2018). Available online: http://www.enterprisesurveys.org, Accessed on 21/04/2021
- EPRA, Energy and Petroleum Statistics Report. (2019). Available online: https://www.epra.go.ke/download/the-energy-act-2019/. Accessed on 25/08/2021
- ERC, Energy Regulatory Commission Biomass. 2019. Available online: https://www.epra.go.ke/ (accessed on 12 January 2021).
- Esmaili.A., and Nasiri.A. (2009). A case study on improving ELCC by utilization of energy storage with solar PV, *35th Annual. Conf. IEEE Ind. Electron, Porto, Portugal*, November 2009; 3957–3962.

- Ezenugu. I. A., Umoren M. A., and Okpura. I. (2016). Performance Analysis of Stand-Alone Photovoltaic (SAPV) System for Category I Health Clinic in Orlu, Imo State, Nigeria. *Mathematical and Software Engineering*: 2(1): 35-47.
- Faninger, G. (2010). The Potential of Solar Heat in the Future Energy System, Faculty for Interdisciplinary Research and Continuing Education, IFF-University of Klagenfurt, Austria.
- Farquharson, D., Jaramillo, P and Samaras, C. (2018). Sustainability implications of electricity outages in sub-Saharan Africa. *Nature Sustainability*, 1(10).
- Felten, H., Kreutzmann, A. and Welter, P. (2006). Increase in grid-connected PV system power in Germany. *Photovoltaic Energy Conversion, Conference Record of the 2006 IEEE 4th World Conference, December 21<sup>st</sup>, Sydney, pages 2494-2496.*
- Fesharaki. V. J.; Dehghani. M. and Fesharaki. J. J. (2011). The effect of temperature on photovoltaic cell efficiency. In Proceedings of the 1st International Conference on Emerging Trends in Energy Conservation (ETEC '11), Tehran, Iran, November 2011.
- Ferrada, P., Araya, F., Marzo, A. and Fuentealba, E. (2015). Performance analysis of photovoltaic systems of two different technologies in a coastal desert climate zone of Chile. *Sol. Energy*, 114, 356–363.
- Fouad, M.M.; Shihata, L.A. and Morgan, E.I. (2017). An Integrated Review of Factors Influencing the Performance of Photovoltaic Panels. Renew. Sustain. Energy Rev. 80, 1499–1511.
- Freris, L, and Infield D. (2008). Renewable energy in power systems, *1st ed., John Wiley and Sons.* ISBN 978-0-470-01749-4.
- Frohlich, C; Geist, Z; Kendall J. and Marchgraber, R. M. (1973). The third international comparison of pyrheliometers and a comparison of radiometric scales. *Solar energy* 14, 157-166.
- Fu; Ran; David F, and Robert M. (2020). U.S. Solar Photovoltaic System Cost Benchmark: Q1 2020. Golden, CO: National Renewable Energy Laboratory.
- Fukushima Renewable Energy Institute. 2014. Available online: https://www.japantimes.co.jp/news/2018/03/11/ national/fukushima-powerstoward-100-goal-renewables-grid-cost-woes-linger/ (accessed on 30 May 2021).

- Gaglia, A.G.; Lykoudis, S. and Argiriou, A.A.; Balaras, C.A.; Dialynas, E. (2017).
  Energy efficiency of PV panels under real outdoor conditions—An experimental assessment in Athens, Greece. Renew. *Energy*; 101, 236–243.
- Gairaa, K.and Bakelli, Y. A. (2013). Comparative study of some regression models to estimate the global solar radiation on a horizontal surface from sunshine duration and meteorological parameters for Ghardaïa Site, Algeria. *ISRN Renew. Energy*, 13, 1–11.
- Garg H. P. and Garg. S. N (1993). Measurement of solar radiation, radiation instruments. *renewable energy*; 3, (4/5), 321- 333.
- Gavalian, P., Lorite, I.J., Tornero, S. and Berengena, J. (2005). Regional calibration of Hargreaves equation for estimating reference ET in a semiarid environment. *Agricultural Water Management*; 81; 257–281.
- George A, Boxiong S, Arowo M, Ndolo P, Chebet C and Shimmon J. (2019). Review of solar energy development in Kenya: opportunities and challenges, *Renew. Energy Focus*.29; 123–40.
- Ghaffarzadeh, N and Azadian, A. (2019). A Comprehensive Review and Performance Evaluation in Solar (PV).*Journal of Solar Energy Research*, 4(4), 252-272.
- Ghafoor, A and Munir, A. (2015). Design and economics analysis of an off-grid PV system for household electrification. *Renewable and Sustainable energy Reviews*, 42: 496-502.
- Ghani F, Rosengarten G, Duke M and Carson JK. (2015). On the influence of temperature on crystalline silicon solar cell characterization parameters. *Solar Energy*; 112:437-445.
- Ghazi, S. and Ip, K. (2014). The effect of weather conditions on the efficiency of PV panels in the southeast of UK. *Renew. Energy*; 69, 50–59.
- Gongsin I. E. and Saporu W. O. (2020). Solar Energy Potential in Yola, Adamawa State, Nigeria. *International Journal of Renewable Energy Sources*; 4; 1420-1429.
- González-Mahecha. R.E., Lucena. A.F.P. and R. Garaffa. (2019). Greenhouse gas mitigation potential and abatement costs in the Brazilian residential sector, *Energy & Buildings*.184; 19–33.
- Gorter T and Reinders A.H. (2012). A comparison of 15 polymers for application in photovoltaic modules in PV powered boats. *Applied Energy*; 92: 286–97.
- Goswami, Kreith and Kreider, Taylor and Francis. (2000). Principles of Solar Engineering. CRC Press of Taylor & Francis Group. ISBN-13: 978 – 1 – 4665 – 6379-7.

- Gottschalg R; Betts T. R; Infield D. G. and Kearney M.J. (2005). The effect of spectral variations on the performance parameters of single and double junction amorphous silicon solar cells. *Solar Energy Materials and Solar Cells*; 85(3), 415–428.
- Grant. B., Mariam, R and Blake. S. (2020). Electricity prices vary across provinces, as well as across industrial, commercial and residential consumers. Overall electricity costs and rate structures matter for economically efficient power consumption and for provincial competitiveness. *Commentary No. 582, C. D. Howe Institute. October 2020;* ISBN 978 – 1 – 989483 – 46-6
- Green, M. A. (2009). The path to 25% silicon solar cell efficiency: History of silicon cell evolution," Progress in Photovoltaics: *Research and Applications*, 17(3), 183–189.
- Griffith, J. S, Rathod, N. S, and Paslaski J. (1991). Some tests of flat plate photovoltaic module cell temperatures in simulated field conditions. *In: Proceedings of the IEEE 15th photovoltaic specialist's conference*, Kissimmee, FL, May 12-15, 1981, Conference Record. (A82-44928 23-44) New York.
- Guerra, A.T., Amador G.J., Tabernero, O.B. and De la Cruz García, G. (2017).
  Comparative Energy Performance Analysis of Six Primary Photovoltaic Technologies in Madrid (Spain). *Energies*, 10, 772. https://doi.org/10.3390/en10060772.
- Guerra, N., Guevara, M., Palacios A., and Crupi, F. (2018). Operation and physics of photovoltaic solar cells: an overview, *I+D Tecnológico*. (14), 84-95.
- Gulaliyev, M.G.; Mustafayev, E.R. and Mehdiyeva, G.Y. (2020). Assessment of Solar Energy Potential and Its Ecological-Economic Efficiency: Azerbaijan Case. *Sustainability*, 12, 1116.
- Habibi M. H, Karimi B, Zendehdel M and Habibi M. (2014). Preparation of nanostructure mixed copper-zinc oxide via co-precipitation rout for dyesensitized solar cells: The influence of blocking layer and Co(II)/Co(III) complex redox shuttle. *Journal of Industrial and Engineering Chemistry*; 20(4):1462-1467.
- Harrouni, S., Guessoum, A. and Maafi, A. (2005). Classification of Daily Solar Irradiation by Fractional Analysis of 10-Min-Means of Solar Irradiance. *Theoretical and Applied Climatology*; 80, 27-36.
- Hemza, A., Abdeslam, H., Rachid, C and Aoun, N. (2019). Simplified methods for evaluating the degradation of photovoltaic module and modelling considering partial shading, *Measurement*, 138, 217-224, ISSN 0263-2241.

- Herbert H. K. and Hermann E. H. (1923). A New Form of Thermoelectric Recording Pyrheliometer; *J. Opt. Soc. Am.* **7**, 707-718.
- Hibberd, B. (2011). PV Reliability & Performance A project Developers Experience. In Photovoltaic Module Reliability Workshop (NREL), Colorado, USA, *February*, 2011.
- Hofmann, M. and Seckmeyer, G. (2017). A New Model for Estimating the Diffuse Fraction of Solar Irradiance for Photovoltaic System Simulations. *Energies*, 10, 248.
- Hossain. M. I, Hongsingthong. A, Qarony. W, Sichanugrist. P, Konagai. M, Salleo. A, Knipp.D and Tsang. Y. H. (2019). Optics of Perovskite Solar Cell Front Contacts. ACS Applied Materials & Interfaces; 11 (16), 14693-14701.
- Huang, G, Li, Z, Li, X, Liang, S, Yang, K, Wang, D, and Zhang, Y, (2019). Estimating surface solar irradiance from satellites: Past, present, and future perspectives. *Remote Sensing of Environment*. 233,111 - 371.
- Hubble, A.H.; Ustun, T.S. (2018). Composition, placement, and economics of rural microgrids for ensuring sustainable development. Sustain. Energy Grids Networks. 13, 1 – 18.
- Idika, O. (1995). Cost comparison of Alternative Electric Power Supply Systems (Grid Extension, Stand - alone Diesel Generator and photovoltaic systems) for Rural and Remote Areas. B. Eng (Thesis), Department of Mechanical Engineering, University of Nigeria, Nsukka.
- IEA, IRENA, UNSD, World Bank, WHO. (2021). Tracking SDG 7: The Energy Progress Report. World Bank, Washington DC.
- IEA PVPS Annual Report (2020). Photovoltaic power systems programme.
- IEC61724-1. (2021). Photovoltaic System Performance- Part 1: Monitoring; *International Electrotechnical Commission*: Geneva, Switzerland.
- IRENA. (2021). World Adds Record New Renewable Energy Capacity in 2020, available at: https://www.irena.org/newsroom/pressreleases/2021/Apr/World-Adds-Record-New-Renewable Energy-Capacity.
- International standard IEC 61724. (1998). Photovoltaic System Performance Monitoring - Guidelines for Measurement, Data Exchange and Analysis.
- IPCC, (2018): Global warming of 1.5 °C. An IPCC Special Report on the impacts of global warming of 1.5 °C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty. Waterfield (eds). In Press.

- Irfan, M., Zhao, Z., Ahmad, M., and Rehman, A. (2019). A Techno-Economic Analysis of Off-Grid Solar PV System: A Case Study for Punjab Province in Pakistan. *Processes*; 7, 708.
- Irwanto M., Daut I., Sembiring M., Ali R., Champakeow S. and Shema S. (2010). Effect of Solar Irradiance and Temperature on Photovoltaic Module Electrical Characteristics. *International Postgraduate Conference on Engineering (IPCE)*, 16-17October 2010, Perlis, Malaysia.
- Jacobson, A. (2007). Connective Power: Solar Electrification and Social Change in Kenya. *World Development*, 35 (1): 144–162.
- Jacovides, C. P; Tymvios, F. S,: Assimakopoulos, V. D and Kaltsounides, N. A. (2006).Comparative study of various correlations in estimating hourly diffuse fraction of global solar radiation. *Renewable Energy*, vol. 31, no. 15, pp. 2492–2504, 2006.
- Jain A. (2004). Exact analytical solutions of the parameters of real solar cells using Lambert W- function. Solar Energy Materials and Solar Cells, 81, (2); 269 -277.
- Jamil, B. and Khan, M.M. (2014). Estimation of Clear-Sky Solar Radiation Using ASHRAE Model for Aligarh, India. *Environmental Science, Engineering*, *Physics.31*, 256-264.
- Jiménez -Castillo, C., Talavera, D.L., Muñoz-Rodriguez, F.J. and Rus-Casas, G. (2019). A new approach to sizing the photovoltaic generator in selfconsumption systems based on cost-competitiveness, maximizing direct self-consumption, *Renewable Energy*, 130, 1021-1035.
- Jordan, D. C. and Kurtz, S. R. (2013). Photovoltaic Degradation Rates—an Analytical Review. *Progress in Photovoltaics: Research and Applications;* 21, (1); 12-29.
- Joseph, A. and Kamala, J. (2013). PV Array Characteristics Analysis under Partial Shading & Modeling of P&O MPPT Applied Boost Convertor Using Matlab/Simulink. In Proceedings of the 2013 International Conference on Energy Efficient Technologies for Sustainability, Nagercoil, India, 596–601.
- Joshua D.R., Charles R.U., Wesley J.C., Chris L.H. and Michael E.W. (2014). A multi- objective assessment of the effect of solar PV array orientation and tilt on energy production and system economics" *Solar Energy*; 108, 28-40.
- Jungbluth N. (2005). Life cycle assessment of crystalline photovoltaics in the Swiss Ecoinvent Database. *Progress in Photovoltaics: Research and Applications*; 13 (5): 429–446.

- Kaldellis J. K and Zafirakis D. (2007). Optimum energy storage techniques for the improvement of renewable energy sources-based electricity generation economic efficiency. *Energy*; 32; 2295-2305.
- Kalgirou, S.A and Tripanagnostopoulos, Y. (2007). Industrial application of PV/T solar energy systems. Applied Thermal Engineering; 27 (8-9), 1259-1270.
- KAM, (2019). Manufacturing priority agenda 2019: Closing the manufacturing gap through the Big 4 Agenda for shared prosperity. *Kenya Association of manufacturers*. Accessed February 16, 2019.
  <a href="http://kam.co.ke/kam/wpcontent/uploads/2019/02/KAM-Priority-Agenda">http://kam.co.ke/kam/wpcontent/uploads/2019/02/KAM-Priority-Agenda</a> 2019.pdf>.
- Kan, S.Y., Verwaal, M. and Broekhuizen, H. (2006). The use of battery–capacitor combinations in photovoltaic powered products. J. Power Sources, 162, 971– 974.
- Kaplanis, S. and Kaplani, E. (2011). Energy Performance and Degradation over 20 years Performance of BP C-Si PV Modules. *Simul. Model. Pract. Theory*; 19, 1201–1211.
- Karakaya, E. and Sriwannawit, P. (2015). Barriers to the adoption of photovoltaic systems: The state of the art. *Renew. Sustain. Energy Rev*, 49, 60–66.
- Kariuki, B. W., and Sato, T. (2018). Inter- annual and spatial variability of solar Radiation energy potential in Kenya using Meteosat satellite. *Renewable* energy, 116, 88-96.
- Kawajiri K, Oozeki T and Genchi, Y. (2011). Effect of Temperature on PV Potential in the World. *Environmental Science and Technology*; 45, 9030-9035.
- Ketjoy, N., Sirisamphanwong, C. and Khaosaad, N. (2013). Performance evaluation of 10 kWp photovoltaic power generator under hot climatic condition. *Energy Proc*; 34, 291–297.
- Kessaissia, F. Z., Zegaouia, A., Hadj Arab, A., Loukarfi, L and Aillerie, M. (2015). Comparison of two PV modules technologies using analytical and experimental methods. *Energy Procedia* 74; 389 – 397.
- Khatoon S and Ibraheem J. M. F. (2014). Analysis of solar photovoltaic array under partial shading conditions for different array configurations. *In: Proceedings of the international conference on innovative applications of computational intelligence on power, energy and controls with their impact on humanity, CIPECH 2014, IEEE*; 452–456.
- Kiplagat, J. K., Wang, R. Z., and Li, T. X. (2011). Renewable energy in Kenya: Resource potential and status of exploitation. *Renewable and Sustainable Energy Reviews*, 15(6): 2960–2973.

- Kiprop.E, Kenichi.M. and Maundu.N. (2018). Can Kenya Supply Energy With 100% Renewable Sources? *International Scientific Journal of Environmental Science*, 2, 25-36.
- Kim, M.; Kim, H. and Jung, J.H. (2021). A Study of Developing a Prediction Equation of Electricity Energy Output via Photovoltaic Modules. *Energies*; 14, 1503.
- Kolhe M, Kolhe S, and Joshi J.C. (2002). Economic Viability of Stand-alone Solar Photovoltaic System in Comparison with Diesel-powered System for India. J. Energy Economics, 24 (2): 155-165.
- Kost, C., Mayer, J. N., Thomsen, J., Hartmann, N., Senkpiel, C., Philipps, S, and Schlegl, T. (2013). Levelized cost of electricity renewable energy technologies. *Fraunhofer institute for Solar Energy Systems* ISE.
- Kristie L Ebi, Luisa F. Cabeza and Jin Chen. (2019). We believe in a brighter future. Introducing Global Transitions – Connecting Forward-Thinking. *Global Transitions*; 1.1-2. 10.1016/j.glt.2019.03.001.
- Kumar, S., Kumar, N. and Vivekadhish, S. (2016). Millennium Development Goals (MDGs) to Sustainable Development Goals (SDGs): Addressing Unfinished Agenda and Strengthening Sustainable Development and Partnership, *Indian journal of community medicine*, 41(1): 1–4.
- Kumar, R., Rajoria, C.S., Sharma A. and Suhag, S. (2020). Design and simulation of standalone solar PV system using PVsyst Software: A case study, *Materials Today*, *Proceedings*, 46 (4).
- Kumar, K., Sundareswaran, K. and Venkateswaranc, P. (2014). Performance Study on a Grid Connected 20 kWp Solar Photovoltaic Installation in an Industry in Tiruchirappalli (India). *Energy for Sustainable Development*, 23; 294-304.
- Kumaresh.V., Malhotra.M., Ramakrishna, N and Saravanaprabu. R. (2014). Literature review on solar MPPT system. Advance in Electronic and Electric Engg. 4, 285-296.
- Kurtz, S. R. and Jordan, D. C. (2013). Photovoltaic Degradation Rates-an Analytical Review. *Prog. Photovolt. Res. Appl*, 21, 12–29.
- Kougias, I., SándorSzabó, F., Monforti-Ferrario, T. and Huld, K.H. (2016). A methodology for optimization of the complementarily between smallhydropower plants and solar PV systems, *Renewable Energy*, 87(2), 1023-1030.

- Lai, C.S. and McCulloch, M.D. (2017). Levelized cost of electricity for solar photovoltaic and electrical energy storage. *Appl. Energy;* 190, 191–203.
- Lalwani, M., Kothari, D. P and Singh, M. (2010). Investigation of solar photovoltaic simulation softwares, *International journal of applied engineering research*, 1(3), 585-601.
- Lang, T.; Ammann, D. andGirod, B. (2016). Profitability in absence of subsidies: A techno-economic analysis of rooftop photovoltaic self-consumption in residential and commercial buildings. *Renew. Energy.* 19, 126–131.
- Lau, K.Y., Muhamad, N.A., Arief, Y.Z., Tan, C.W and Yatim, A.H.M. (2016). Grid connected photovoltaic systems for Malaysian residential sector: Effects of component costs, feed-in tariffs, and carbon taxes. *Energy*, 102, 65–82.
- Lee K, Brewer E, Carson C, Meyo F, Miguel E, Podolsky M, Rosa J, and Wolfram C. (2014). Barriers to Electrification for "Under Grid" Households in Rural Kenya, National Bureau of Economic Research, Cambridge, MA.
- Lee, H.Y.; Yoon, S.H. and Park, C.S. (2015). The Effect of Direct and Diffuse Split Models on Building Energy Simulation. J. Arch. Inst. Korea Plan. Des; 31, 221–229.
- Li, D.H. and Lam, J.C. (2001). An Analysis of Climatic Parameters and Sky Condition Classification. *Building and Environment*; 36, 435-445.
- Lim S.D, Mazzoleni A.P, Park J, Ro P.I and Quinlan B. (2012). Conceptual design of ocean compressed air energy storage system. *In Proc: Oceans*; 1-8.
- Longley, P.A.; Goodchild, M.F.; Maguire, D.J. and Rhind, D. (2001). Geographic Information Systems and Science; *John Wiley & Sons Ltd.*: Hoboken, NJ, USA.
- Lorenzo, E.; Moretón, R.; Luque, I. (2014). Dust effects on PV array performance: Infield observations with non-uniform patterns. *Prog. Photovolt. Res. Appl*; 22, 666–670.
- Louwen A, de Waal A. C, Schropp R. E, Faaij A. P and van Sark W. G. (2017). Comprehensive characterization and analysis of PV module performance under real operating conditions. *Prog. Photovoltaics Res. Appl.* 25 (3); 218-232.
- Luceño-Sánchez, J., Díez-Pascual, A. and Peña Capilla, R. (2019). Materials for Photovoltaics: State of Art and Recent Developments. *Int. J. Mol. Sci*, 20(4), 976.

- Ludin NA, Mustafa NI, Hanafiah MM, Ibrahim MA, Asri Mat Teridi M and Sepeai S. (2018). Prospects of life cycle assessment of renewable energy from solar photovoltaic technologies: A review. Renewable and Sustainable Energy Reviews; 96: 11-28.
- Luque A, Hegedus S. (2003). Handbook of photovoltaic science and engineering. *1st* ed. John Wiley; 154-204.
- Luzzi.A. and Lovegrove, K. (2004). Solar thermal power generation. *In Encyclopedia of Energy*, 5; Elsevier: Amsterdam, U.K.
- McCollum, D.L., Zhou, W., Després, J., Drouet, L., Emmerling, J., Fay, M., Fricko,
  O., Fujimori, S., Gidden, M., Harmsen, M., Huppmann, D., Iyer, G., Krey, V.,
  Kriegler, E., Nicolas, C., Pachauri, S., Parkinson, S., Poblete-Cazenave, M.,
  Rafaj, P., Rao, N., Rozenberg, J., Schmitz, A., Schoepp, W., van Vuuren, D.
  and Riahi, K. (2018). Energy investment needs for fulfilling the Paris
  Agreement and achieving the Sustainable Development Goals. *Nat Energy*, 3, 589–599.
- McCann M.J, Catchpole K.R, Weber K.J and Blakers A.W. (2001). A review of thinfilm crystalline silicon for solar cell applications. Part 1: Native substrates. *Solar Energy Materials and Solar Cells*; 68: 135–71.
- Maende, S., and Alwanga, M. (2020). The Cost of Power Outages on Enterprise Performance in Kenya, *International Journal of Research and Innovation in Social Science*, 4(3), 2454-6186.
- Mahmoud, M.M., Ibrik, I.H. (2006). Techno-economic Feasibility of Energy Supply to Remote Villages in Palestine by PV-Systems, Diesel Generator and Electri Grid. *Renewable Sustainable Energy Rev.*, 10: 128-138.
- Mahmoud, Y., Xiao, W and Zeineldin, H. H. (2012). A simple approach to modeling and simulation of photovoltaic modules. IEEE Trans. *Sustainable Energy*, 3(1):185-6.
- Makrides G, Zinsser B, Schubert M and Georghiou G.E. (2014). Performance loss rate of twelve photovoltaic technologies under field conditions using statistical techniques. *Sol Energy*; 103: 28–42.
- Malvoni M, Leggieri A, Maggiotto G, Congedo P. M and Giorgi M. G. (2017). Long term performance, losses and efficiency analysis of a 960 kW P photovoltaic system in the Mediterranean climate. *Energy Conver. Manage*.145; 169–81.
- Martinez S. M., Carreton M. C., Escribano A.H., Lazaro E. G., 2019. Performance evaluation of large solar photovoltaic power plants in Spain. *Energy Convers. Manage*. 183, 515-528.

- Matteocci.F., Vesce. L., Kosasih. F. U., Castriotta. L. A., Cacovich. S., Palma.A.L., Divitini. G., Ducati. C. and Di Carlo.A. (2019). Fabrication and Morphological Characterization of High-Efficiency Blade-Coated Perovskite Solar Modules. ACS Applied Materials and Interfaces. 11 (28): 25195–25204.
- Mayfield R (2012). The highs and lows of photovoltaic system calculationsunderstanding calculations for proper system design and installation of photovoltaic systems, *Electrical, Construction and Maintenance (ECM):* 1-6.
- Mcnutt P., Kroposki B., Hansen R., Deblasio R., Lynn K., Wilson W., Rosenthal A., Boulanger P and Malbranche, P. (2000).Procedures for determining the performance of stand-alone photovoltaic systems. *Conference Record of the IEEE Photovoltaic Specialists Conference*. 1556 - 1559.
- Mellit, A., Kalogirou, S.A., Shaari, S., Salhi, H. and Arab, A.H. (2008). Methodology for Predicting Sequences of Mean Monthly Clearness Index and Daily Solar Radiation Data in Remote Areas: Application for Sizing a Stand-Alone PV System. *Renewable Energy*; 7, 1570-1590.
- Messenger, R. and Ventre, G. (2003). Photovoltaic Systems Engineering, Second Edition. *Wiley*.
- Meteonorm handbook, Version 6.12, Part II: Theory. Meteotest, 2010
- Micheli, D.; Alessandrini, S.; Radu, R. and Casula, I. (2014). Analysis of the outdoor performance and efficiency of two grid connected photovoltaic systems in northern Italy. *Energy Convers. Manag*; 80, 436–445.
- Mittag, M.; Vogt, L.; Herzog, C. and Neuhaus, H. (2019). Thermal modeling of photovoltaic modules in operation and production. *In Proceedings of the 36th EUPVSEC, Marseille, France*.892–900.
- Mohammed Al-Smairan, Rida Al-Adamat, and Omar Al-Nhoud. (2012). Techno-Economic Feasibility of Energy Supply of Remote Dump Site in Jordan Badia by Photovoltaic Systems, Diesel Generators and Electrical Grid. *Research Journal of Applied Sciences, Engineering and Technology*; 4(9), 1073-1081.
- Mohamed A. Eltawil T, and Zhengming Z. (2013). MPPT techniques for photovoltaic applications. *Renew. Sustain. Energy Rev*; 25, 793-813.
- Mohandes, N., A. Sanfilippo, M. Al Fakhri. (2019). Modeling Residential Adoption of Solar Energy in the Arabian Gulf Region. *Renewable Energy, Elsevier*; 131; 381-389.
- Moharil R.M and Kulkarni P.S. (2009). A case study of solar photovoltaic power system at Sagardeep Island, India. *Renewable and Sustainable Energy Reviews* 2009; 13:673-681.

- Moriarty, P. and Honnery, D. (2012). What is the global potential for renewable energy? *Renewable and Sustainable Energy Reviews*, 16(1) 244-252.
- Motahhir.S., Chalh. A., Ghzizal. A. E. and Derouich. A. (2018). Development of a low-cost PV system using an improved INC algorithm and a PV panel Proteus model. *Journal of Cleaner Production*, 204: 355-365.
- Müller, B., Hardt, L., Armbruster, A., Kiefer, K. and Reise, C. (2016). Yield Predictions for Photovoltaic Power Plants: Empirical Validation, Recent Advances and Remaining Uncertainties: Yield Predictions for Photovoltaic Power Plants. *Prog. Photovolt. Res. Appl*, 24, 570–583.
- Mulligan, C. J., Bilen, C., Zhou, X., Belcher, W. J., and Dastoor, P. C. (2015). Levelized cost of electricity for organic photovoltaics. *Solar energy materials and solar cells*; 133, 26-31.
- Muñoz J V, Nofuentes G, Fuentes M, De la Casa J, and Aguilera J (2016). DC energy yield prediction in large monocrystalline and polycrystalline PV plants: Time-domain integration of Osterwald's model. *Energy*, 114; 951-960.
- Musanga, L.M, Barasa, W.H. and Mageto, M. (2018). The Effect of Irradiance and Temperature on the Performance of Monocrystalline Silicon Solar Module in Kakamega. *Physical Science International Journal*; 19 (4): 1-9.
- Museruka, C. and Mutabazi, A. (2007). Assessment of Global Solar Radiation over Rwanda. International Conference on Clean Electrical Power 2007 (ICCEP'07); 670–676. http://dx.doi.org/10.1109/ICCEP.2007.384312.
- Muslim, Hasan Noaman, AfaneenAlkhazraji, and Mohammed Ahmed Salih. (2018).
  Feasibility study of using 2kWp residential PV system comparing with 2.5 kV.
  A gasoline generator (Case study: Baghdad city). *International Journal of Energy and Environment;*9(1): 57-62.
- Myers. D. R., Emery. K. and Gueymard. C. (2002). Revising and validating spectral irradiance reference standards for photovoltaic performance. *In Proceedings of the ASES/ASME solar conference*, Reno, Nevada.
- Myers D. R, Stoffel T. L, Wilcox S, Reda I and Andreas A. (2002). Recent progress in reducing the uncertainty in and improving pyranometer calibrations. *Journal of Solar Energy Engineering*. 124; 44-50.
- Ndwali, P. K., Njiri, J. G, and Wanjiru, E. M. (2020). Optimal Operation Control of Micro grid Connected Photovoltaic-Diesel Generator Backup System Under Time of Use Tariff, *Journal of Control, Automation and Electrical Systems*, 1– 14.

- Nerini, F., Tomei, J., Long Seng, T., Bisaga, I., Parikh, P., Black, M., Borrion, A., Spataru, C., CastánBroto, V., Anandarajah, G., Milligan, B. and Mulugetta, Y. (2018). Mapping synergies and trade-offs between energy and the Sustainable Development Goals. *Nat Energy*, 3, 10–15.
- Nieuwenhout, F.D.J., van Dijk, A., Lasschuit, P.E., Van Roekel, G., Van Dijk,
  V.A.P., Hirsch, D. and Wade, H. (2001). Experience with solar home systems in developing countries: *A review. Prog. Photovolt. Res. Appl.*, 9, 455–474.
- Nikolaidou. K., Sarang. S. and Ghosh.S. (2019). Nano-structured photovoltaics. *Nano Futures*. 3 (1): 012-022.
- Njeru. G, Gathiaka. J and Kimuyu.P. (2020). Technical Efficiency of Thermal Electricity Generators in Kenya. *International Journal of Energy Economics and Policy*; 10 (3), 340-347.
- Nkhonjera, L. and Wu, J. (2013). Performance Analysis of Battery Based Stand-Alone Solar Home Systems in Malawi. *Journal of Energy and Power Engineering*, 7, 866-875.
- Odeh, S. and Nguyen, T.H. (2021). Assessment Method to Identify the Potential of Rooftop PV Systems in the Residential Districts. *Energies*; 14, 42-48.
- Okello D, van Dyk E.E and Vorster F.J. (2015). Analysis of measured and simulated performance data of a 3.2 kWp grid-connected PV system in Port Elizabeth, South Africa. *Energy Convers Manage*; 100:10–15.
- Olatomiwa L.J., Mekhilef, S and Huda A.S.N. (2014). Optimal sizing of hybrid energy system for a remote telecom tower: A case study of Nigeria. *In Proc: IEEE Conf. on Energy Conversion (CENCON)*, 243-247.
- Olchowik, J.M.; Gulkowski, S.; Cieslak, K.; Banas, J.; Jozwik, I.; Szymczuk, D.; Zabielski, K.; Mucha, J.; Zdrojewska, M. and Adamczyk, J. (2006). Influence of temperature on the efficiency of monocrystalline silicon solar cells in the South-eastern Poland conditions. *Mater. Sci.-Pol*; 24 (4), 1127–1132.
- Oliver, M. and Jackson, T. (2001). Energy and economic evaluation of buildingintegrated photovoltaics. *Energy*, 26, 431–439.
- Oloo, F., Olang, L. and Strobl, J. (2015). Spatial Modeling of Solar Energy Potential in Kenya. *Int. J. Sustain. Energy Plan. Manag.* 6: 17–30.
- Omar, M.A. and Mahmoud, M.M. (2018). Grid connected PV-home systems in Palestine: A review on technical performance, effects and economic feasibility. *Renew. Sustain. Energy Rev*, 82, 2490–2497.

- Oparaku, O. U. (2003). Rural Area Power Supply in Nigeria. A cost comparison of the photovoltaic, Diesel/Gasoline Generator and Grid Utility Options. *Pergamon, Renewable Energy;* 28, 2089-2098.
- Oyelami, S., Azeez, N. A., Adedigba, S. A., Akinola, O. J and Ajayi, R.M. (2020). A Pyranometer for Solar Radiation Measurement-Review.*Journal of Engineering and Technology;* 3, (1), 61-68.
- Ozden, T.; Akinoglu, B.G. and Turan, R. (2017). Long term outdoor performances of three different on-grid PV arrays in central Anatolia—An extended analysis. Renew. Energy; 101, 182–195.
- Pacala, S. and Socolow, B. (2004). Stabilization wedges: Solving the climate problem for the next 50 years with current technologies, *Science*, 305(5686), 968–972.
- Pacca S, Sivaraman D and Keoleian G.A. (2007). Parameters affecting the life cycle performance of PV technologies and systems. *Energy Policy*; 35 (6), 3316-3326.
- Panwar K, Agarwal K and S. Pandey S. (2017). Indian Climate Based Weighted Conversion Efficiency of Solar Photovoltaic Inverter for North and South Zone: 7 (4), 1499–1502.
- Pearsall, N. (2016). The Performance of Photovoltaic (PV) Systems: Modelling, Measurement and Assessment. Elsevier, Woodhead Publishing, pp.
- PengJ; LuL.and Yang H. (2013). Review on life cycle assessment of energy payback and greenhouse gas emission of solar photovoltaic systems. *Renew Sust Energy Rev*; 19, 255–274.
- Perez. R.R; Ineichen P; Maxwell E.L; Seals R.D and Zelenka A. (1992). Dynamic Global to Direct Conversion Models, ASHRAE Transactions Research Series; 154-168.
- Pessanha, M.S., dos Santos, L.M., Lyra, G.B. (2021). Interpolation methods applied to the spatialisation of monthly solar irradiation in a region of complex terrain in the state of Rio de Janeiro in the southeast of Brazil. *Model. Earth Syst. Environ.* 7, 1565–1578.
- Phinikarides A, Kindyni N, Makrides G and Georghiou G.E. (2014). Review of photovoltaic degradation rate methodologies. Renew Sustain Energy Rev; 40:143–52.
- Pietruszko S. M., Fetlinski B. and Bialecki M. (2012). Analysis of the Performance of Grid Connected Photovoltaic System. Power and Energy Engineering Conference (APPEEC), 1-4.
- Pillai G, Naser HAY. (2018). Techno-economic potential of large scale photovoltaics in Bahrain. *Sustainable Energy Technologies and Assessments;* 27: 40-45.
- Pless, J., and Fell, H. (2017). Bribes, bureaucracies, and blackouts: Towards understanding how corruption at the firm level impacts electricity reliability. *Resource and Energy Economics*, 47, 36-55.
- Preston, B. L., Backhaus, S. N., Ewers, M., Phillips, J. A., Silva-Monroy, C. A., Dagle, J. E. and King, J. T. (2016). Resilience of the U.S. Electricity System: A Multi-Hazard Perspective. *Argonne National Laboratory*, 48(2).
- Pundir, K.S.S.; Varshney, N.; Singh, G.K. (2017). Comparative Study of Performance of Grid Connected Solar Photovoltaic Power System in IIT Roorkee Campus. In Proceedings of the International Conference on Innovative Trends in Science, Engineering and Management, New Delhi, India, 7 January 2017; 423–431.
- PVSyst photovoltaic software. http://www.pvsyst.com/en/ (accessed: 25/12/2021)
- Qing, X. and Niu, Y. (2018). Hourly day-ahead solar irradiance prediction using weather forecasts by LSTM. Energy.148, 461–468.
- Rached.D and Mostefaoui.R. (2008). Influence of the front contact barrier height on the indium tin oxide/hydrogenated p-doped amorphous silicon hetero-junction solar cells. *Thin Solid Films*: 516 (15); 5087–5092.
- Rahil, A., Madziga, M. and Mansoor, R. (2018). Comparison between Three Off-Grid Hybrid Systems (Solar Photovoltaic, Diesel Generator and Battery Storage System) for Electrification for Gwakwani Village, South Africa.*Environments;* 5, (57).
- Rabah K.V. (2005). Integrated solar energy systems for rural electrification in Kenya. *Renewable Energy*, 30 (1), 23–42.
- Rajput, S.K and Singh, O. (2017). Reduction in CO<sub>2</sub> Emission through Photovoltaic System: A Case Study. In Proceedings of the 3rd IEEE International Conference on Nanotechnology for Instrumentation and Measurement, Greater Noida, India, 16–17.
- Rakhi S. and Tiwari G.N. (2012). Technical performance evaluation of stand-alone photovoltaic array for outdoor field conditions of New Delhi. *Applied Energy*, 92; 644-652.
- Ramirez C. L; Nitsch, F; Gruber, K and Dorner W. (2018). Electricity self-sufficiency of single-family houses in Germany and the Czech Republic. *Applied Energy*; 228: 902–915.
- Rao, C.N.; Bradley, W.A. and Lee, T.Y. (1984). The diffuse component of the daily global solar irradiation at Corvallis, Oregon (USA). *Sol. Energy*, 32, 637–641.

- Rehman, S., Bader, M. A. and Al- Moallem, S. A. (2007). Cost of solar energy generated using PV panels. *Renewable and sustainable energy reviews*;11, 1843–1857.
- Reich, N.H.; Mueller, B.; Armbruster, A.; Van Sark, W.G.; Kiefer, K.andReise, C. (2012). Performance ratio revisited: Is PR > 90% realistic? *Prog. Photovolt. Res. Appl*; 20, 717–726.
- Rekhashree A, Rajashekar J.S and Naganagouda H. (2018). Study on Design and Performance Analysis of Solar PV Rooftop Standalone and On Grid System Using PVsyst" *IRJET*.5 (7).
- Republic of Kenya. (2018). Kenya Electricity Sector Investment Prospectus 2018-2022. Nairobi: Ministry of Energy Renewable resource data center (http://rredc.nrel.gov/)
- Research and Markets. (2017). Available Online: http://www.researchandmarkets.com/research/xfhx6m/kenya\_diesel. Accessed on 21/8/2021.
- Rodrigues.S., Chen.X. and Morgado-Dias.F. (2017). Economic analysis of photovoltaic systems for the residential market under China's new regulation. *Energy Policy*. 101, 467–472.
- Rodrigues S, Torabikalaki R, Faria F, Cafôfo N, Chen X and Ivaki A.R. (2016). Economic feasibility analysis of small scale PV systems in different countries. *Solar Energy*; 131: 81-95.
- Ross, R.G. (1980). Flat-plate photovoltaic array design optimization. *14th IEEE Photovoltaic Specialists Conference*, 1126-1132.
- Ruiz-Arias J. A., Cebecauer T., Tovar-Pescador J. and Šúri M. (2010). Spatial disaggregation of satellite-derived irradiance using a high-resolution digital elevation model. *Solar Energy*, 84, 1644-1657.
- Sabzipavar, A.A., Mousavi, R., Marofi, S. (2013). An improved estimation of the Angstrom-Prescott radiation coefficients for the FAO 56 Penman- Monteith evapotranspiration method.*Water. Manag*; 27, 2839–2854.
- Sagadevan S. (2012). Recent Trends on Nanostructures Based Solar Energy Applications: *A Review. Rev. Advance Material Sci.* 34; 44-61.
- Sah, R. L. Y., Noyce, R. N and Shockley, W. (1957). Carrier generation and recombination in P-N junctions and P-N junction characteristics, *Proceedings* of the IRE; 45(9), 1228–1243.

- Salih S. M, Salih F. F., and Hasan M. L. (2012). Performance evaluation of photovoltaic models based on a solar model tester. *I.J. Information Technology and Computer Science*; 7; 1-10.
- Sahu, B.K. (2015). A study on global solar PV energy developments and policies with special focus on the top ten solar PV power producing countries. *Renew. Sustain. Energy Rev.* 43, 621–634.
- Saiful, I., Achim, W., Ronnie, B., Heskes, P.J.M. and Rooij, P.M.
   (2006). Investigating performance, reliability and safety parameters of photovoltaic module inverter: Test results and compliances with the standards. *Renewable Energy*, 31(8), 1157-1181
- Samoita, D., Nzila, C., Poul, A.O and Remmen, A. (2020). Barriers and Solutions for Increasing the Integration of Solar Photovoltaic in Kenya's Electricity Mix. *Energies*, 13, 5502; doi: 10.3390/en13205502.
- Sampaio, P.G.V.; González, M.O.A. (2017). Photovoltaic solar energy: Conceptual framework. *Renew. Sustain. Energy Rev.* 74, 590–601.
- Sanajaoba, S. (2019). Optimal Sizing of off-Grid Hybrid Energy System Based on Minimum Cost of Energy and Reliability Criteria Using Firefly Algorithm. *Solar Energy*, 188: 655–666.
- Sangram B.R.P. (2016). Saini, A mathematical modeling framework to evaluate the performance of single diode and double diode based SPV systems. *Energy Rep*;2, 171-187.
- Sengupta, M.; Habte, A.; Kurtz, S.; Dobos, A.; Wilbert, S.; Lorenz, E.; Stoffel, T.; Renné, D.; Gueymard, C.A. and Myers, D. (2015). Best Practices Handbook for the Collection and Use of Solar Resource Data for Solar Energy Applications; *National Renewable Energy Laboratory*: Golden, CO, USA.
- Satish, M.; Santhosh, S.; Yadav, A. (2020). Simulation of a Dubai Based 200 KW Power Plant Using PVsyst Software. In Proceedings of the 2020 7th International Conference on Signal Processing and Integrated Networks (SPIN), Noida, India, 27–28 February 2020; 824–827.
- Satsangi P., Das B., Babu S. and Saxena A.K. (2018). Performance analysis of grid interactive solar photovoltaic plant in India. *Energy Sustain. Dev*; 47, 9-16.
- Seme, S, Sredensek, K, Stumberger, B. and Hadziselimovic, M. (2019). Analysis of the Performance of Photovoltaic Systems in Slovenia. *Solar Energy*; 180, 550–558.
- Serrano E, Rus G and Garcia-Martinez J. (2009). Nanotechnology for sustainable energy. *Renewable and Sustainable Energy Reviews*; 13: 2372-84.

- Sethi VK, Pandey M and Shukla P. (2011).Use of nanotechnology in solar PV cell. International Journal of Chemical Engineering and Applications; 2(2): 77-80.
- Shafey, H.M. and Ismail, I.M. (1990). Thermodynamics of the conversion of solar radiation. *Journal of Solar Energy Engineering*, 112, 140.
- Shah, S.; Valasai, G.; Memon, A.; Laghari, A.; Jalbani, N. and Strait, J. (2018). Techno-economic analysis of solar PV electricity supply to rural areas of Balochistan, Pakistan. *Energies*, 11, 1777.
- Shah AV, Schade H, Vanecek M, Meier J, Vallat-Sauvain E and Wyrsch N. (2004). Thin-film silicon solar cell technology. Progress in Photovoltaics: *Research and Applications*;12 (23):113-142.
- Shaw-Williams, D.; Susilawati, C. and Walker, G. (2018). Value of residential investment in photovoltaics and batteries in networks: A techno-economic analysis. *Energies*; 11, 10-22.
- Shenoy, V., Tripathi, P., Mahadik, A., Nalawade, P. and Mahajan, A. (2018). Devices Used for Measuring Solar Radiation – A Review. International Conference on Innovative and Advanced Technologies in Engineering..IOSR Journal of Engineering (IOSRJEN); 7, 01-04.
- Shubbak, M. H. (2019). The technological system of production and innovation: The case of photovoltaic technology in China. *Research Policy*. 48 (4): 993–1015.
- Shukla, A.K., Sudhakar, K and Baredar, P. (2016). Exergetic assessment of BIPV module using parametric and photonic energy methods, A review. *Energy Build*. 119, 62–73.
- Siddiqui.R and Bajpai, U. (2012). Deviation in the performance of solar module under climatic parameter as ambient temperature and wind velocity in composite climate. *International Journal of Renewable Energy Research*; 2(3), 486–490.
- Siddiqui, M.U. and Said, S.A.M. (2015). A review of solar powered absorption systems, *Renew Sustain Energy Rev*, 42, 93–115.
- Sidi, C. E., Ndiaye, B. E., Bah, M. L., Mbodji, M. E., Ndiaye, A., and. Ndiaye, P. A. (2016). Performance analysis of the first large-scale (15MWp) grid-connected photovoltaic plant in Mauritania. *Energy Conversion and Management*, 119 (1): 411–421.
- Sifat, M., Shoaib, M.A., Tanzil, H.C., Anikur, R.S. and Rahman, A. (2015).
  Designing of a 2 kW stand-alone PV system in Bangladesh using PVsyst, Homer and SolarMAT, *Conference Paper, DOI:* 10.1109/ICGET.2015.7315090.

- Skoczek A., Virtuani A., Cebecauer T., Chianese D. (2011). Energy Yield Prediction of Amorphous Silicon PV Modules Using Full Time Data Series of Irradiance and Temperature for Different Geographical Locations. 26th European Photovoltaics Solar Energy Conference, September 2011, Hamburg, Germany.
- Skoplaki E and Palyvos J.A. (2009). On the temperature dependence of photovoltaic module electrical performance: A review of efficiency/power correlations. *Solar Energy*; 83, 614-624.
- Smets, A., Jger., K., Isabella, O., Swaaij, R. and Zeman, M. (2016). Solar Energy: The Physics and Engineering of Photovoltaic Conversion, Technologies and Systems. UIT Cambridge.
- Spea.S.R. and Khattab.H.A. (2019). Design sizing and performance analysis of stand- alone PV system using PVSyst software for a location in Egypt. *In: Proc. of 21st Int. Middle East Power Syst. Conf. MEPCON 2019 – Proc*, 9: 927–932.
- Spooner, E.D and Harbidge, G. (2001). Review of international standards for grid connected photovoltaic systems, *Renewable Energy*, 22, 235-239.
- Sreenath, S.; Sudhakar, K.; Yusop, A.F.; Solomin, E. and Kirpichnikova, I.M. (2020). Solar PV Energy System in Malaysian Airport: Glare Analysis, General Design and Performance Assessment. *Energy Rep*; 6, 698–712.
- Solar energy perspectives. (2011). *International Energy Agency*, accessed on 15 January 2021.
- SolarGIS, Global Horizontal Irradiation Kenya, 2013. Available: http://solargis.info/doc/\_pics/freemaps/1000px/ghi/SolarGIS-Solar-map-Kenya-en.png. Accessed on 3/05/2021
- Tanesab, J.; Parlevliet, D.; Whale, J. and Urmee, T. (2017). Seasonal effect of dust on the degradation of PV modules performance deployed in different climate areas. *Renew. Energy*; 111, 105–115.
- Thong, L. W., Murugan, S., Ng, P. K. and Sun, C. C. (2016). Analysis of Photovoltaic Panel Temperature Effects on Its Efficiency. *Paper presented at the 2nd International Conference on Electrical Engineering and Electronics Communication System. Ho Chi Minh, Vietnam.* 18-19.
- Tian, H., Mancilla-David, F., Ellis, K., Muljadi, E and Jenkins, P. A. (2012). Cell-tomodule-to-array detailed model for photovoltaic panels. *Solar Energy*; 86(9):2695-706.

- Timilsina, G. R., Kurdgelashvili, L. and Narbel, P.A. (2011). A Review of Solar Energy: Markets, Economics and Policies. Paper is funded by the Knowledge for Change Program (KCP); Available: https://openknowledge.worldbank.org/handle/10986/3612. Accessed on 6/07/2021
- Tiwari. G. N., Yadav. J. K., Singh. D. B., Al-Helal. I. M and Abdel-Ghany. A. M. (2015). Exergoeconomic and enviroeconomic analyses of partially covered photovoltaic flat plate collector active solar distillation system, *Desalination*, 367, 186–196.
- Toffalis, C. (2015). A better Measure of Relative Prediction Accuracy for Model Selection and model Estimation. J. of Operational Research Society, 66(8), 1352-1362.
- Tohsing, K., Phaisathit, K., Pattarapanitchai, D., Masiri, S., Buntoung, I., Aumporn, S.O. and Wattan, R. (2019). A development of a low-cost pyranometer for measuring broadband solar radiation, *Journal of Physics*: Conference Series 1380.
- Tonui, J.K and Tripanagnostopoulos, Y., (2008). Performance improvement of PV/T solar collectors with natural air flow operation. *Solar Energy*, Vol. 82 (1), 1-12.
- Tonui, J.K and Tripanagnostopoulos, Y., (2007). Air-cooled PV/T solar collectors with low-cost performance improvements. *Solar Energy*, Vol. 81(4), 498-511,
- Topic, M., Kurnik, J., Jankovec, M. and Brecl, K. (2011). Outdoor testing of PV module temperature and performance under different mounting and operational conditions, *Solar Energy Materials and Solar Cells*, 95 (1), 373-376.
- Trillo-Montero, D.; Santiago, I.; Luna-Rodriguez, J.J. and Real-Calvo, R. (2014). Development of a software application to evaluate the performance and energy losses of grid-connected photovoltaic systems. *Energy Convers. Manag*; 81, 144–159.
- Tymvios, F.S., Jacovides, C.P. and Michaelides, S.C. (2005). Comparative study of Angstrom and artificial neural networks methodologies in estimating global solar radiation. *Sol. Energy*; 78, 752–762.
- Ulgen, K. and Hepbasli, A. (2002), Solar Radiation Models. Part 1: A Review. *Energy Sources*, 26: 413-430.
- Usman, Z.; Tah, J.; Abanda, H. and Nche, C. (2020). A Critical Appraisal of PV-Systems'*Performance.Buildings*.10,192. https://doi.org/10.3390/buildings10110192.
- Vazquez, S., Lukic, S. M., Galvan, E., Franquelo, L. G and Carrasco, J. M. (2010).

Energy storage systems for transport and grid applications. *IEEE Trans Ind Electron*, 57(12):3881-95. http://dx.doi.org/10.1016/j.apenergy.2015.11.037.

- Venkata R. K and Muralidhar N. B. (2016). A simple and efficient MPPT scheme for PV module using 2-Dimensional Lookup table. *IEEE power and energy conference at Illinois;* 24. 1-7
- VignolaFotis. F, MavromatakisFotis and MavromatakisJim K. (2008). Performance of PV inverters. *American Solar Energy Society*.
- Warner, K.J and Jones, G.A. (2017). A Population-Induced Renewable Energy Timeline in Nine World Regions. *Energy Policy*, 101, 65–76.
- Weis, C. (2013, January). Considerations for Off-Grid PV Systems. Retrieved from HomePower:https://www.homepower.com/articles/solarelectricity/designinstal lation/considerations-grid-pv-systems
- Whittingham M.S. (2012). History, evolution and future status of energy storage. *Proceedings of the IEEE*; 100:1518-34.
- Wohlgemuth, J.H. (2012). Standards for PV Modules and Components- Recent Developments and Challenges, 27th European Photovoltaic Solar Energy Conf. and Exhibition; 2976-2980.
- World Bank (2013), World Development Indicators, Number of power outages in a month.http://databank.worldbank.org/data/views/reports/tableview.aspx. accessed on 25/04/2021.
- Woyte, A., Belmans, R. and Nijs, J. (2007). Fluctuations in Instantaneous Clearness Index: Analysis and Statistics. *Solar Energy*; 2, 195-206.
- Yali L, Xu. Ming Li, Xi Luo J, Meidi W and Yng Z. (2014). A comparative study of the maximum power point tracking methods for PV systems. *Energy Convers. Manage*; 01, 49-56.
- Yang. Y. M., Yu. A., Hsu. B., Hsu. W. C., Yang. A., and Lan, C. W. (2015). Development of high-performance multicrystalline silicon for photovoltaic industry. *Progress in Photovoltaics: Research and Application*; 23 (3), 340 – 351.
- Young-Min K, Jang-Hee L, Seok-Joon K and Daniel F. (2012). Potential and evolution of compressed air energy storage: energy and exergy analyses. *Entropy*; 14:1501-1521.

- Zaoui F., Titaouine A., Becherif M., Emziane M. and Aboubou A. (2015). A combined experimental and simulation study on the effects of irradiance and temperature on photovoltaic modules. *Energy Procedia*; 75: 373-380.
- Zhang, F., Zhou, X., Zhang, H., Peng, X. and Wang, Z. (2014). On the relationship between direct and anisotropic diffuse radiation. *Infrared Phys. Technol.* 65, 5–8.
- Žnidarec, M.; Šljivac, D.; Došen, D. and Dumnin, B. (2019). Performance Assessment of Mono and Poly Crystalline Silicon Photovoltaic Arrays under Pannonian Climate Conditions. *In Proceedings of the IEEE EUROCON 2019–* 18th International Conference on Smart Technologies, Novi Sad, Serbia, 1–4 July 2019; IEEE: Piscataway, NJ, USA, 2019; 1–6.
- Zhu, Z.; Mao, K. and Xu, J. (2020).Pervoskite tandem solar cells with improved efficiency and stability. *J. Energy Chem*; 58, 219–232.

#### **APPENDICES**

Year		INST	ALLED C	АРАСІІ	ſΥ		Total		EFF	ECTIVE	CAPACI	TY		Total
	Hydro	Thermal Oil	Geo- thermal	Wind	Co- gene- ration	Solar	MW	Hydro	Thermal Oil	Geo- thermal	Wind	Co- gene- ration	Solar	MW
2016	818.7	801.6	652	26.1	28	0.6	2327	797.5	762.9	644	26	23.5	0.2	2254.1
2017	826.2	806.9	652	26.1	28	0.7	2339.9	805	765.8	644	25.5	23.5	0.6	2264.4
2018	826.2	807.7	663	336.1	28	50.7	2711.7	805	768.2	655	335.5	23.5	50.6	2637.8
2019	826.2	749.3	828.4	336.1	28	51	2818.9	805	716	816	325.5	23.5	50.4	2736.4
2020	834	749.1	863.1	336.1	2	52.5	2836.7	805	715.5	805.1	325.5	2	52.2	2705.3
							Notes:							
	1. Megawatt = 1,000 kilowatts = 1,000,000 watts													
	2. Install	ed capacity	refers to the	e maximu	im theore	tical elec	etric outpu	t a power	station could	d produce v	when ope	rating at	100 perce	ent
	3. Effecti	ve capacity	refers to th	e maximu	um electri	ic output	a power s	tation is e	xpected to a	chieve give	en current	t operatin	g constra	int

# APPENDIX I: INSTALLED CAPACITY OF ELECTRICITY BY SOURCE, 2016-2020

Source: Kenya Power & Lighting Company Ltd and Kenya Electricity Generation Company Ltd (2020).

Parameter	Value
Power requirement (kW)	0.08
Rated AC output voltage(V)	230±5
Maximum DC Input Voltage (V)	1000
Input voltage (V)	750
Maximum output current (A)	22
Maximum input current (A)	15
Efficiency (%)	96
AC line frequency (Hz)	50
Power factor	0.9
Power Control	MPPT
Operating Temperature ( <sup>0</sup> C)	0~60

### APPENDIX II: INVERTER TECHNICAL CHARACTERISTICS

Parameter	Magnitude					
Nominal system voltage (V)	12-24					
Maximum input voltage (V)	24					
Max. PV input power (W)	1040					
Rated battery current (A)	20					
Rate load current (A)	15					
Self-consumption	≤16mA					
Working temperature	$-25^{\circ}C \sim +60^{\circ}C$					
Power conversion efficiency (%)	95%					
Safety factor	1.25					

### APPENDIX III: CHARGE CONTROLLER SPECIFICATIONS

### APPENDIX IV: BATTERY TECHNICAL PARAMETERS

Parameter	Magnitude
Battery capacity (Ah)	200
Nominal voltage (V)	25.4
Energy (WH)	5120
Depth of discharge (DOD)	0.8
Efficiency (%)	96
Discharge temperature ( <sup>0</sup> C)	-20~60
Charge temperature ( <sup>0</sup> C)	0~45
Self-discharge	<1% per month
Peak discharge current (A)	200
Discharge cut-off voltage (V)	4.2±0.05
	1

		Power rating	Operation	Daily power		
Appliance	Quantity	(W)	time (h/day)	load (kWh/day)		
Light bulbs	4	8.5	2	0.068		
Fluorescent tube	12	35	2	0.840		
Desktop computer	1	30	2	0.060		
PC (Laptop)	2	18.5	2	0.074		
Television	1	60	2	0.120		
Fridge	1	150	2	0.300		
Phone/ mobile	6	6	2	0.072		
Fan	6	30	2	0.360		
Printer	1	21	2	0.022		
Sound system	4	20	2	0.160		
Total				2.076		

### APPENDIX V: LOAD POWER RATINGS

### APPENDIX VI: PYRANOMETER (SOLAR METER) SPECIFICATIONS

Parameter	Value
Timer	Sampling from 1 second to 3600 seconds
Spectral response	400 to 1100 nm
Solar power range	0~2000 W/m <sup>2</sup> or 634 Btu/ $ft^2 \times h$
Solar power accuracy	$\pm$ 10 W/m <sup>2</sup> or 5 % reading
Angular accuracy	Cosine corrected $<5\%$ for angles $< 60^{\circ}$
Operating temperature	0~50 <sup>0</sup> C
Operating humidity	Less than 85% R.H

Item	Value						
Module cost (\$)	1.11/W						
Battery cost (\$)	@114.28						
Charge controller/inverter (\$)	165.24						
4 x 16mm <sup>2</sup> cable (\$/m)	1.0						
Length of cable (m)	235						
Installation cost (\$)	5% of component cost						
O&M/year (\$)	2% of PV array cost						
No. of replacement for PV	0						
Battery lifespan (years)	10						
No. of replacement for battery	2						
Charge-cum-inverter lifespan (years)	10						
No. of replacement for charge-cum-inverter	2						
PV derating factor (%)	80						
Interest rate (%)	6.5						
Inflation rate (%)	2.0						
Average hours of operation	9						
Continuous cloudy days	4						
Temperature correction factor	0.95						
Cost of electricity (\$) (grid)	0.21/kWh						
Inverter replacement periodic cost (\$)	220						

### APPENDIX VII: PV SYSTEM COST PARAMETERS

Parameter	Value							
Interest rate (%)	7.5							
Inflation rate (%)	6.0							
Generator cost (\$)	240							
Installation cost (\$)	5% of components cost							
O and M cost (\$ per op. Hr)	0.03							
Generator lifespan (years)	5							
Number of generators	1							
Number of replacements	4							
Fuel cost- average (\$)	1.0 /L							
Length of cable (m)	235							
Cost of cable (\$/m)	1.0							

### APPENDIX VIII: DIESEL GENERATOR COST PARAMETERS

# APPENDIX IX: REPRESENTATIVE MONTHLY SOLAR IRRADIANCE

Time	Time	21st	18th	11th	16th	20th	10th	14th	9th	11th	16th	23rd	18th
7.04	07.04.00	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec
7.34	07:34:00 AM	25.3	21.6	24.3	42.3	18.6	32.1	54	28.6	29	65.2	12.3	56.3
7.39	07:39:00 AM	11.2	106.5	101	56	25.3	31	212	128.3	45	24.1	11.2	62.9
7.44	07:44:00 AM	15.2	125.2	204	95	100.2	35	154	100.5	81	135.9	15.2	54.9
7.49	07:49:00	58.7	35	296	124	95.6	78.5	276	142.6	56	56.3	58.7	61
7.54	07:54:00 AM	46.9	78.5	201	96.3	85	66.7	103	81	95	62.9	46.9	36.6
7.59	07:59:00 AM	115.8	66.7	158	99.2	84	135.6	245	56	124	54.9	115.8	135
8.04	08:04:00 AM	127.6	78.5	308	154	168	147.4	165	95	127.6	61	202	111.2
8.09	08:09:00 AM	137.7	66.7	312	276	166.1	166.1	179	124	137.7	36.6	189	122.3
8.14	08:14:00 AM	350.2	135.6	296	103	144.1	144.1	157	127.6	350.2	135	212	653.1
8.19	08:19:00 AM	407.4	147.4	356	245	212	218.6	254	235.1	407.4	422.3	155	604.9
8.24	08:24:00 AM	421.6	166.1	245	102.3	215.6	312.1	325	235.6	421.6	262.8	339	674.2
8.29	08:29:00 AM	389.7	144.1	476	205.6	196	118.5	214	304.2	389.7	541.6	413	500.9
8.34	08:34:00 AM	212.6	212	321	185.7	243.4	356.1	256.3	369	212.6	425.9	256.3	720
8.39	08:39:00 AM	349	312.1	478.2	323.2	122.1	312.1	135	325	310.1	471.8	213	724.5
8.44	08:44:00 AM	266.7	196	498.2	504.7	142.3	148	422.3	157	319	455.6	395.7	632.4
8.49	08:49:00 AM	299.3	356.1	600.4	398.1	249.9	144.1	262.8	315.6	230.1	500.7	287.6	852.9
8.54	08:54:00 AM	356.9	635.1	639.5	456.4	256.3	312	541.6	308	461.1	412.3	400.8	694.2
8.59	08:59:00 AM	351.1	826.4	522	363.9	152.7	310.1	425.9	325	199.1	350.2	321.6	736.8
9.04	09:04:00 AM	395.3	736.4	803.4	421.5	448	319	471.8	356	397.1	407.4	256.9	745.2
9.09	09:09:00 AM	496.8	115.8	728.5	407.4	569	230.1	455.6	245	312	421.6	501.2	769.8
9.14	09:14:00 AM	468.1	127.6	805	625.1	412	461.1	500.7	476	345	389.7	200.1	722.6
9.19	09:19:00 AM	95.5	137.7	849	856.1	322	196	241	241	423	426.1	459.7	705.7
9.24	09:24:00 AM	123.8	350.2	725	566.5	356	311.1	326	326	496	437.3	93.3	803.4
9.29	09:29:00 AM	560.9	407.4	800	656.3	296	199.1	214	214	653	726.8	478.2	728.5
9.34	09:34:00 AM	586.2	421.6	823	956.3	425	397.1	412	412	589	657.1	498.2	835.6
9.39	09:39:00 AM	495.7	389.7	931.9	822.1	500	312	514	514	625	720.6	600.4	806.6
9.44	09:44:00 AM	505.8	212.6	932.1	722.6	412	345	200	615.8	411	658.7	639.5	812.4
9.49	09:49:00 AM	564.9	349	930.8	755.8	312	423	612	612	716	703	722.6	822.7
9.54	09:54:00 AM	569.1	266.7	981.9	721.3	267	496	596	596	685	725.6	625.9	704.5
9.59	09:59:00 AM	635.9	299.3	836	722.6	625	509	625	653	635.4	698.3	803.4	704.6
10.04	10:04:00 AM	541.1	356.9	901	658.2	500	632.6	611	589	631.3	705.4	728.5	722.6
10.09	10:09:00 AM	524.3	322.1	732	865.6	656	633.6	612	625	635.6	856.3	903.5	748.1
10.14	10:14:00 AM	567.4	351.1	901.9	845.7	523	635	589	411	639.4	689.5	859.6	729.7
10.19	10:19:00 AM	541.1	395.3	856	826.4	574	590.6	569	856.5	712	756.4	933.5	897.3

10.24	10:24:00 AM	596.1	496.8	928.5	726.5	584	605.6	584	584	603	689.9	478.5	869.3
10.29	10:29:00 AM	562.3	468.1	912.1	864.1	546	567.6	546	587.9	656	754.1	951.7	877.6
10.34	10:34:00 AM	601.9	95.5	894.2	956.4	603	598	603	459.7	632	896.5	951.3	901.5
10.39	10:39:00	596.2	123.8	905.7	505.8	589	677.6	656	512.4	689	854.3	754.3	867.2
10.44	10:44:00	564.2	560.9	884.8	896.5	611	653.6	632	478.2	568	876.2	1024.5	865.3
10.49	10:49:00	603.5	586.2	903.5	433.2	612.3	710.6	689	498.2	359	867.2	869.2	789.6
10.54	10:54:00	631.8	495.7	924.4	635.9	589	584	623	600.4	400	825.6	956.5	789.6
10.59	10:59:00	635.4	505.8	933.5	989.5	672.7	633.6	635.4	639.5	718	796.3	1004.5	894.5
11.04	11:04:00	631.3	564.9	639	844	641.4	485	631.3	623	436	814.1	957.6	901
11.09	11:09:00	635.6	569.1	613	872.1	598.6	459	635.6	876.2	623	856.4	959	898.4
11.14	11:14:00 AM	639.4	635.9	653.2	896.2	611.8	587.6	639.4	867.2	690	786.3	958.8	910.3
11.19	11:19:00 AM	712	541.1	613	956.1	611.4	603.2	745	825.6	512	856.2	1054.3	856.2
11.24	11:24:00	755	524.3	672	854.2	578.5	518	525	796.3	895.3	804.1	1002.5	705.2
11.29	11:29:00 AM	698	567.4	718	689.5	614.8	564	695	814.1	803.5	829.7	953.3	689.5
11.34	11:34:00 AM	702.6	541.1	865.9	458.6	505.6	652.8	725	856.4	400.8	863.1	856.3	585.2
11.39	11:39:00 AM	736.9	596.1	924.6	745.8	534.9	625	763	635.2	653.2	845.9	758.6	925.6
11.44	11:44:00 AM	865.9	562.3	932.7	796.5	586.2	778.7	654	589.7	502.1	821.1	960.7	945
11.49	11:49:00 AM	924.6	601.9	994.1	854.6	523.3	589	612	872.1	501.2	852.6	965.3	973.1
11.54	11:54:00 AM	932.7	596.2	986.4	877.3	621.3	554	603	896.2	700.2	789.6	967.8	922.8
11.59	11:59:00 AM	994.1	564.2	826.7	965.4	845.3	672.7	639	956.1	994.1	854.2	970.3	945.6
12.04	12:04:00 PM	986.4	603.5	956.8	562.1	823.1	641.4	613	925.6	986.4	946.5	789.5	1000.7
12.09	12:09:00 PM	826.7	631.8	1055	639.4	846.3	705.6	613	857.4	800.3	926.6	857.6	986.5
12.14	12:14:00 PM	956.8	635.4	1054	789.6	895.3	611.8	613	859.6	994.1	906.7	984.6	989.9
12.19	12:19:00 PM	956.4	631.3	1046	933.6	803.5	611.4	672	907.7	986.4	639.1	754.8	1026.8
12.24	12:24:00 PM	926.4	645.5	1026.2	766.6	718.3	588.5	722.8	963.2	841.9	931.4	997.5	1099.3
12.29	12:29:00 PM	965.8	639.4	1043	702.6	736.9	612	436	986.4	956.8	856.2	973	1122.5
12.34	12:34:00 PM	1032.9	725.6	1055	736.9	865.9	687.9	623	826.7	857.3	911.7	975.3	1126.3
12.39	12:39:00 PM	1051.1	632.5	1033	798.4	924.6	706.1	690	956.8	895.6	722.8	900.5	1230.5
12.44	12:44:00 PM	1033.7	511.6	1022	854.6	932.7	589	512	956.4	825.6	1026	923.3	1194.3
12.49	12:49:00 PM	1045.2	702.6	1010	789.5	865.2	696	610	923.5	869.5	994.1	954.4	512.3
12.54	12:54:00 PM	939.5	736.9	1001	857.3	700.5	594.5	436	965.8	621.3	965.3	964.8	1235
12.59	12:59:00 PM	1096.3	865.9	1008	854.6	700.25	535	600	1032.9	826.7	985.6	925.3	1215.7
13.04	01:04:00 PM	1065.2	924.6	1002	752.1	800.25	588	653	968.7	956.8	956.8	935.6	1266.2
13.09	01:09:00 PM	1145.9	932.7	990.4	1065.2	700.25	558	623	854.6	1055	1045.2	967.8	1258.6
13.14	01:14:00 PM	1186.3	994.1	990.9	956.4	990.4	588	653	752.1	1054	1056.8	970.3	1239.8
13.19	01:19:00 PM	1134.2	986.4	976.7	847.1	990.9	800	865	1065.2	905.2	732.5	1000.2	1206.4
13.24	01:24:00	1045.6	826.7	1000	741.6	976.7	660	725	956.4	845.6	1032.9	1023.3	1214.2

	PM												
13.29	01:29:00 PM	1196.5	956.8	997.8	854.7	700.25	594	659	847.1	856.3	1056.3	1024.5	1228.1
13.34	01:34:00 PM	1231.9	956.4	822.2	624.3	826.7	624	689	879.2	852.4	986.2	1005.2	1245.5
13.39	01:39:00 PM	1154.6	923.5	1073	725.1	956.8	561	685	859.1	741.5	733.8	1045.2	1196.9
13.44	01:44:00 PM	758.2	965.8	1052	648.5	956.4	540	735	785.4	706.1	300.2	1022.3	1205.9
13.49	01:49:00 PM	1135.9	1032.9	1059	837.7	923.5	547	825	687.9	968.35	856.2	800.1	1189.3
13.54	01:54:00 PM	1066.3	1051.1	1102	330.5	700.25	590	725	706.1	958.41	856.3	924.6	1223.1
13.59	01:59:00 PM	1012.9	1033.7	1063	256.1	852.3	551	625	968.35	725	789.5	1002.3	1238.6
14.04	02:04:00 PM	1033.5	1045.2	1072	355.7	700.25	560	695	958.41	785.6	415.7	1038.1	1125.4
14.09	02:09:00 PM	1012.9	939.5	965.2	287.1	700.25	408	639	618.2	692	789.5	1023.4	1209.5
14.14	02:14:00 PM	1189.6	1096.3	718.6	556.9	600	494	695.6	725	605	896.5	984.2	1123.5
14.19	02:19:00 PM	1089.7	1065.2	736.9	324.4	635	414	605.6	645	700.2	985.4	985.9	1245.9
14.24	02:24:00 PM	1046.8	1145.9	865.9	986.4	831.6	396	567.6	692	986.5	859.1	942.4	856.2
14.29	02:29:00 PM	989.5	1186.3	721.5	925.6	800	374	598	605	937.4	722.5	923.2	704.1
14.34	02:34:00 PM	1053.6	1134.2	1106	899.9	632	392	677.6	623	901.1	542.6	985.6	1178.6
14.39	02:39:00 PM	1024.9	1045.6	1154.6	985.6	623	395.6	653.6	623	926.4	845.5	880.1	1208.3
14.44	02:44:00 PM	1027.3	1196.5	758.2	900.6	700.25	304.2	702.1	687.3	865.1	986.5	946.3	1144.7
14.49	02:49:00 PM	856.9	1231.9	1135.9	561.9	589	290	636.3	784.1	847.7	926.7	824.6	1232.3
14.54	02:54:00 PM	829.4	1154.6	1066.3	432.5	700.25	425	656	656	900.5	1033.5	901.3	1178.9
14.59	02:59:00 PM	964.3	758.2	1012.9	524.1	635	521.6	721	721	986.5	1012.9	869.8	1100.2
15.04	03:04:00 PM	836	1135.9	1033.5	665.4	831.6	254	612	865.1	937.4	1032.4	845.2	1034.4
15.09	03:09:00 PM	765	1066.3	985	638.8	800	385	600	847.7	901.1	1089.7	824.9	1145.2
15.14	03:14:00 PM	986.5	1012.9	965	465.2	632	365	635	900.5	964.3	1072.6	833.4	803.5
15.19	03:19:00 PM	937.4	1033.5	865.9	327.9	451	265.9	831.6	986.5	836	989.5	920.3	824.9
15.24	03:24:00 PM	901.1	1012.9	924.6	336.1	362	324.6	800	937.4	765	1006.7	936.4	935.6
15.29	03:29:00 PM	926.4	1189.6	932.7	325.4	525	189	632	901.1	831.6	1028.3	845.3	926.2
15.34	03:34:00 PM	865.1	1089.7	994.1	432.1	489	394.1	623	725	800	935.6	823.1	856.4
15.39	03:39:00 PM	847.7	1046.8	986.4	973	478	386.4	512	645	632	726.3	846.3	505.4
15.44	03:44:00 PM	856.9	989.5	826.7	975.3	425	154	502	692	623	829.4	895.3	562.3
15.49	03:49:00 PM	865.2	1053.6	956.8	901.1	458	356.8	514.5	605	523.6	719.4	803.5	710.1
15.54	03:54:00 PM	732.6	1024.9	956.4	964.3	458	356.4	332	901.3	583.6	986.1	824.9	300.6
15.59	03:59:00 PM	625.9	1027.3	923.5	836	522.3	186	322	869.8	421	925.4	785.6	536.8
16.04	04:04:00 PM	754.8	856.9	965.8	765	485	365.8	412	412	436	628.3	793.5	938.2
16.09	04:09:00 PM	735.9	829.4	725	831.6	368	313	352	536.8	445	652.9	800.3	935.3
16.14	04:14:00 PM	752.1	964.3	645	800	356	233	312	938.2	365.8	523.4	720.4	924.7
16.19	04:19:00 PM	789.5	758.6	692	632	415	280	456	935.3	313	226.1	601.3	899.1
16.24	04:24:00 PM	741.6	695.3	605	623	385	193	312	924.7	233	217	710.1	426.9

16.29	04:29:00 PM	426.7	986.5	901.3	225.1	653	489.3	421	899.1	280	234.6	833.4	428.6
16.34	04:34:00 PM	505.6	937.4	869.8	465.1	700.5	457.8	436	426.9	193	331.8	831.6	637.1
16.39	04:39:00 PM	534.9	901.1	845.2	541.6	289	305	445	428.6	489.3	216.7	800	824.6
16.44	04:44:00 PM	461.5	926.4	824.9	330.3	360	412.9	415	637.1	541.6	94.5	632	505.4
16.49	04:49:00 PM	523.3	865.1	833.4	326.1	298	421.4	423	289	330.3	137.1	623	496.2
16.54	04:54:00 PM	517.6	847.7	920.3	330.1	256	508.3	362	360	326.1	126.2	560.25	702.3
16.59	04:59:00 PM	514.8	856.9	936.4	356.7	359	325	312	298	325	98.7	725	455.6
17.04	05:04:00 PM	465.1	865.2	845.3	286.7	458	312	310	256	312	186.2	645	696.1
17.09	05:09:00 PM	541.6	732.6	823.1	295.7	475	276	316	296	456.2	152.3	692	435.8
17.14	05:14:00 PM	330.3	625.9	846.3	186.3	322	212	300	300	365.2	168.1	605	765.9
17.19	05:19:00 PM	326.1	754.8	895.3	145.9	269	96	324	137.1	126	249.3	623	582.2
17.24	05:24:00 PM	330.1	735.9	803.5	359.4	299	101	369	126.2	254	196.5	365.3	402.6
17.29	05:29:00 PM	356.7	752.1	400.8	368.1	351	124	278	98.7	385	314.6	450.2	321.9
17.34	05:34:00 PM	286.7	789.5	321.6	425.6	362	86	221	186.2	365	125.9	325	300.7
17.39	05:39:00 PM	295.7	741.6	256.9	154	311	89	145	152.3	265.9	86.4	312	201.9
17.44	05:44:00 PM	186.3	426.7	501.2	276	412	135	125	168.1	324.6	109.1	276	385.6
17.49	05:49:00 PM	145.9	505.6	200.1	103	478	154	110	249.3	185	219.3	212	206.4
17.54	05:54:00 PM	359.4	534.9	459.7	245	452	85	92	196.5	256	126.2	96	198.6
17.59	05:59:00 PM	368.1	461.5	330	165	189	169	96	314.6	300	128.7	325	188.5
18.04	06:04:00 PM	425.6	523.3	524		412	95	123	102.3	98	87	255	145.5

# APPENDIX X: MONTHLY CLEARNESS INDEX

Month	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec
Clearness index	0.694	0.672	0.599	0.524	0.412	0.386	0.397	0.496	0.522	0.497	0.596	0.631

Month	Wind Speed (m/s)	Ambient temperature ( <sup>0</sup> C)				
Jan	3.71	17.9				
Feb	3.41	18.9				
Mar	3.76	19				
Apr	1.07	18.2				
May	1.66	17.8				
Jun	1.54	16.6				
Jul	1.96	16.5				
Aug	1.77	16.3				
Sep	2.08	16.9				
Oct	2.87	17.7				
Nov	3.78	17.5				
Dec	3.39	17.9				

### APPENDIX XI: MONTHLY AVERAGE WIND SPEED AND AMBIENT TEMPERATURE

	Experin	nental	Theoret	tical
Month	Experimental	Experimental	Theoretical	Theoretical
WOnu	CO <sub>2</sub> Emission	Cost (\$)	CO <sub>2</sub> Emission	Cost (\$)
Jan	0.56928	8.25456	0.671	9.7295
Feb	0.65782	9.53839	0.793	11.4985
Mar	0.62082	9.00189	0.6114	8.8653
Apr	0.4805	6.96725	0.601	8.7145
May	0.44374	8.57849	0.6904	10.0108
Jun	0.41372	6.43423	0.6112	8.8624
Jul	0.43408	5.99894	0.5172	7.4994
Aug	0.50624	6.29416	0.6148	8.9146
Sep	0.6085	7.34048	0.7126	10.3327
Oct	0.59162	8.82325	0.749	10.8605
Nov	0.54996	7.97442	0.6882	9.9789
Dec	0.52952	7.67804	0.6928	10.0456

APPENDIX XII: MONTHLY CO<sub>2</sub> EMISSION AND ENVIRONMENTAL COST

### APPENDIX XIII: INPUT AND OUTPUT SIMULATION PARAMETERS

PVSYST V6.70								14/09/21	Page 4/7
		Star	nd Alon	e Syst	em: Ma	in resul	ts		
Project :	KAI	BOI							
Simulation varia	ant: Nev	v simula	tion vari	ant					
	Sim	ulation fo	or the first	t year of	operation				
Main system para PV Field Orientatio PV modules PV Array Battery Battery Pack User's needs	ameters on	aily house	Syster Nb. of m Batter Nb. o ehold cons	m type tilt Model odules y module of units sumers	Stand alor 15° LX-195M/12 4 Li-lon, 26V 4 Constant o	ne with b 25-72+ 180 Ah Volta ver the yea	ack-up ge azimi Pnom to Technolo age / Capao ar Glo	nerator uth 180° om 195 Wp otal <b>780 Wp</b> ogy Lithium- city <b>26 V / 8</b> bal 4614 kV	ion, LFP 00 Ah Wh/year
Main simulation of System Production Back-Up energy fro	results n om generator	Perfo	vailable E Used I rmance Ra Back-Up	Energy Energy atio PR energy	<b>1092 kWh/</b> 4615 kWh/ 73.63 % 3636 kWh/	<b>'year</b> year Exc Sola year Fuel	Specific pro cess (unuse ar Fraction Consumpt	od. 1401 k ed) 0 kWh/ SF 21.23 % ion 2181/ye	Wh/kWp/year year sar
Investment Yearly cost Energy cost	Annui	( ties (Loan	Global incl 5.0%, 20	. taxes years)	3568 US\$ 286 US\$/yr <b>0.59 US\$/k</b>	- F Wh	Spec Running Co	ific 4.57 US sts 2447 US	S\$/Wp S\$/yr
The second secon	d energy (full battery) d energy (full battery) in Loss (P-varry losses n losses and battery chan supplied to the user	a) DAWNIKA 0 DE I I AWNIKA 0 DE I AWNI 0 DE I AWNI Juli Aug S	Abday WANDay MANAday MANAday MANAday	lew simulat	and the set of the set	PR: Pedoma OF: Golar Fra	Apr May Jun	UT20	Oct Nov Dec
			Ba	lances and	main results				
		GlobHor	GlobEff	EAvail	EUnused	EUser	E Load	SolFrac	
	January	95.7	55.2	40.7	0.000	392.0	391.9	0.087	
	February	106.2	72.4	53.0	0.005	354.0	354.0	0.126	
	April	185.0	160.3	111.1	0.000	379.3	379.3	0.268	
	May	220.2	202.4	135.1	0.005	392.0	391.9	0.317	
	July	238.3	226.5 238.3	146.0 150.7	0.000	379.4	379.3	0.363	
	August	227.7	200.6	129.3	0.005	392.0	391.9	0.304	
	September	171.4	138.3	93.4	0.000	379.3	379.3	0 223	
	November	101.5	60.7	44.2	0.000	379.3	379.3	0.091	
	December	83.3	44.9	32.9	0.005	392.0	391.9	0.057	
	Year	1981.8	1620.9	1092.5	0.019	4615.4	4614.5	0.212	
	Legends: GlobHor GlobEff E Avail EUnused	Horizontal Effective G Available 3 Unused er	global irradiation Slobal, corr. for IA Solar Energy hergy (full battery)	) M and shading ) loss	E User E Load SolFrac	Energy su Energy ne Solar frac	pplied to the user ad of the user (Los tion (EUsed / ELos	d) d)	

#### **APPENDIX XIV: MAIN RESULT OF PV SYSTEM PRODUCTION**

	Stand Alone Sys	stem: Main results	
Project :	KAIBOI		
Simulation variant :	New simulation variant Simulation for the first year of	of operation	
Main system parameters	System type	Stand alone with back-up genera	ntor
PV Field Orientation	tilt	15° azimuth	180°
PV modules	Model	LX-195M/125-72+ Pnom	195 Wp
PV Array	Nb. of modules	4 Pnom total	780 Wp
Battery	Battery modul	e Li-lon, 26V 180 Ah Technology	Lithium-ion, LFP
Battery Pack	Nb. of units	4 Voltage / Capacity	26 V / 800 Ah
User's needs	Daily household consumers	Constant over the year Global	4614 kWh/year
Main simulation results			
System Production	Available Energy Used Energy Performance Ratio PR	1092 kWh/yearSpecific prod.4615 kWh/yearExcess (unused)73.63 %Solar Fraction SF3636 kWh/yearFuel Consumption	1401 kWh/kWp/year 0 kWh/year 21.23 % 2181/year
buok op energy nom gene	Buck-op chergy	oboo kuniyear ruer oonsumption	210119001
Investment	Global incl. taxes	3568 US\$ Specific	4.57 US\$/Wp
Yearly cost Energy cost	Annuities (Loan 5.0%, 20 years)	286 US\$/yr Running Costs 0.059US\$/kWh	2447 US\$/yr

### APPENDIX XV: PERFORMANCE RATIO AND ANNUAL SOLAR FRACTION



Performance Ratio PR and Solar Fraction SF

ı.

#### APPENDIX XVI: PV SYSTEM LOSS DIAGRAM



#### APPENDIX XVII: PV SYSTEM PERFORMANCE COEFFICIENTS

	Yr	Lu	Yu	Lc	Ya	Ls	Yf	PR
	kWh/m².day		kWh/kWp/d		kWh/kWp/d		kWh/kWp/d	
January	2.00	0.000	2.00	0.215	1.78	0.377	1.40	0.703
February	2.81	0.000	2.80	0.243	2.56	0.523	2.04	0.727
March	4.20	0.000	4.20	0.377	3.83	0.627	3.20	0.761
April	5.57	0.000	5.57	0.547	5.03	0.680	4.35	0.780
Мау	6.75	0.000	6.75	0.832	5.92	0.789	5.13	0.760
June	7.80	0.000	7.80	1.188	6.61	0.726	5.88	0.755
July	7.94	0.000	7.94	1.332	6.60	0.763	5.84	0.736
August	6.71	0.000	6.71	1.054	5.66	0.727	4.93	0.735
September	4.86	0.000	4.86	0.639	4.22	0.611	3.61	0.743
October	3.41	0.000	3.41	0.416	3.00	0.580	2.42	0.708
November	2.25	0.000	2.25	0.260	1.99	0.519	1.47	0.654
December	1.65	0.000	1.65	0.212	1.44	0.513	0.93	0.562
Year	4.67	0.000	4.67	0.612	4.06	0.620	3.44	0.736

#### Normalized Performance Coefficients

### APPENDIX XVIII: DAILY ENERGY OUTPUT OF THE PV ARRAY



#### Daily Input/Output diagram

### APPENDIX XIX: SIMULATION ECONOMIC PERFORMANCE RESULTS

	Stand Alone System:	Economic evaluation	
Project:	AIBOI		
Simulation variant:	New simulation variant Simulation for the first year of	operation	
Main system parameters PV Field Orientation PV modules PV Array Battery Battery Pack User's needs	System type tilt Model Nb. of modules Battery module Nb. of units Daily household consumers	Stand alone with back-up ger           15°         azimu           LX-195M/125-72+         Pno           4         Pnom to           Li-lon, 26V 180 Ah         Technolog           4         Voltage / Capac           Constant over the year         Glob	terator ath 180° om 195 Wp tal <b>780 Wp</b> gy Lithium-ion, LFP ity <b>26 V / 800 Ah</b> bal 4614 kWh/year
Investment			
PV modules (Pnom = 195Wp Supports/Integration Batteries (26 V/3 Ah) controller Settings, wiring, . Substitution underworth Gross investment (witho	o) 4 units 4 units ut taxes)	215 US\$/ unit 0 US\$/module 450 US\$/ unit	858 US\$ 0 US\$ 1800 US\$ 210 US\$ 235 US\$ 0 US\$ <b>3103 US\$</b>
Financing			
Gross investment (without tax Taxes on investment (VAT) Gross investment (including V Subsidies Net investment (all taxes in	es) Rate 15.0 % /AT)		3103 US\$ 465 US\$ 3568 US\$ 0 US\$ <b>3568 US\$</b>
Annuities Maintenance insurance, annual taxes Provision for battery replacem Fuel for Back-Up generator	( Loan 5.0 % o nent (lifetime 6.9 ye (Back-Up ener	over 20 years) ears) rgy 3636 kWh/year)	286 US\$/year 17 US\$/year 0 US\$/year 20 US\$/year 2410 US\$/year
Total yearly cost			2734 US\$/year
Energy cost Used solar energy Excess energy (battery full) Used energy cost			46 15 kWh / year 0.0 kWh / year 0.059 US\$/ kWh

# APPENDIX XX: CO<sub>2</sub> EMISSION BALANCE.

	Stand Alone Syst	em: CO2 Balance	
Project :	KAIBOI		
Simulation variant :	New simulation variant		
	Simulation for the first year o	f operation	
Main system parameters	System type	Stand alone with back-up genera	itor
PV Field Orientation	tilt	15° azimuth	180°
PV modules	Model	LX-195M/125-72+ Pnom	195 Wp
PV Array	Nb. of modules	4 Pnom total	780 Wp
Battery	Battery module	e Li-Ion, 26V 180 Ah Technology	Lithium-ion, LFP
Battery Pack	Nb. of units	4 Voltage / Capacity	26 V / 800 Ah
User's needs	Daily household consumers	Constant over the year Global	4614 kWh/year
Produced Emissions	Total:	0.00 tCO2	
	Source:	Custom value supplied by User	
Replaced Emissions	Total:	9.0 tCO2	
•	System production:	1092.41 kWh/yr Lifetime:	25 years
		Annual Degradation:	1.0 %
	Grid Lifecycle Emissions:	331 gCO2/kWh	
	Source:	IEA List Country:	Kenya
CO2 Emission Balance	Total:	8.0 tCO2	
	Saved CO2 Em 10 10 10 10 10 10 10 10 10 10	nission vs. Time	

### APPENDIX XXI: I-V CHARACTERISTIC DATA

21 <sup>st</sup> Ja	in	18 <sup>th</sup> F	eb	11 <sup>th</sup> N	1ar	16 <sup>th</sup> A	pr	20 <sup>th</sup> N	1ay	10 <sup>th</sup> Ju	un	14 <sup>th</sup> Ju	ıl	9 <sup>th</sup> Au	g	11 <sup>th</sup> S	ер	16 <sup>th</sup> O	ct	23 <sup>rd</sup> N	ov	18 <sup>th</sup> D	ec
Voltage (V)	Current (A)	Voltage (V)	Current (A)	Voltage (V)	Current (A)	Voltage (V)	Current (A)	Voltage (V)	Current (A)	Voltage (V)	Current (A)	Voltage (V)	Current (A)	Voltage (V)	Current (A)	Voltage (V)	Current (A)	Voltage (V)	Current (A)	Voltage (V)	Current (A)	Voltage (V)	Current (A)
0.00	5.38	29.99	0.00	0.01	5.62	0.00	5.24	0.00	5.02	0.01	4.85	0.01	5.15	37.8	0.01	0.00	5.36	0.00	5.4	0.00	5.42	0.00	5.65
3.12	5.36	29.86	0.53	1.52	5.49	3.12	5.19	3.12	4.99	1.52	4.81	5.62	5.12	37.51	0.22	5.62	5.34	2.56	5.39	3.21	5.39	3.25	5.55
5.26	5.3	29.35	1.02	3.21	5.46	5.26	5.11	5.26	4.96	3.21	4.78	12.3	5.05	36.77	0.52	12.75	5.31	15.2	5.26	9.56	5.3	10.32	5.42
10.36	5.29	29.05	1.23	6.52	5.41	10.36	5.01	10.36	4.85	6.52	4.62	20.66	5	36.29	0.75	16.06	4.96	19.56	5.19	21.71	5.28	17.72	5.37
16.62	5.28	28.89	1.5	10	5.36	16.62	4.98	16.62	4.76	10	4.51	26.92	4.96	35.84	0.82	19.54	4.85	25.82	5.08	27.97	5.11	21.2	5.32
18.35	5.23	28.75	2.11	11.11	5.31	18.35	4.56	18.35	4.71	15.62	4.49	28.65	4.85	35.69	1.05	20.65	4.76	27.55	4.98	29.7	4.93	22.31	5.15
19.63	5.22	28.65	2.24	13.2	5.34	20.63	4.24	20.63	4.69	16.35	4.39	29.93	4.64	35.52	1.23	22.74	4.56	29.83	4.62	31.98	4.56	24.4	4.95
20.03	5.21	28.05	2.44	14.7	5.24	23.23	4.19	23.23	4.56	17.35	4.29	30.33	4.35	35.46	1.92	24.24	4.49	32.43	4.25	34.58	4.34	25.9	4.68
20.55	5.13	27.85	2.76	18.9	5.22	25.55	3.95	24.55	4.25	24.95	4.18	30.85	4.25	34.8	2.05	28.44	4.21	34.75	3.95	35.9	3.92	30.1	4.18
21.06	5.14	27.65	2.85	19.4	5.19	26.86	3.85	25.86	4.05	25.24	4.1	31.36	4.14	34.18	2.62	28.94	4.11	36.06	3.85	37.21	3.76	30.6	3.95
21.55	5.12	27.65	2.94	19.95	5.15	27.55	3.73	27.55	3.95	25.89	3.95	31.85	3.96	33.83	3.23	29.49	3.95	36.75	3.64	38.9	1.92	30.55	3.75
21.95	5.01	27.02	2.95	20.44	5.16	28.95	3.62	27.95	3.84	26.02	3.94	32.25	3.72	33.48	3.64	29.98	3.62	38.15	1.82	39.3	1.81	30.64	3.35
23.36	4.93	26.58	3.02	20.75	5.09	29.06	3.41	28.36	3.71	26.45	3.85	33.66	3.53	33.1	3.85	30.29	3.26	38.26	1.61	39.71	1.68	30.95	3.05
24.96	4.92	26.25	3.12	21.03	5.04	30.16	3.23	28.96	3.21	26.98	3.56	35.26	3.26	32.89	4.01	30.57	3.00	39.36	1.43	40.31	1.18	31.13	2.84
25.26	4.82	25.23	3.56	21.88	4.72	31.26	3.12	29.26	3.1	27.04	3.42	35.56	2.22	32.3	4.12	31.42	2.53	40.46	1.32	40.61	1.07	31.38	2.54
26.49	4.08	24.56	3.72	22.23	4.08	31.49	2.94	29.49	2.94	27.56	3.35	36.79	1.48	32.14	4.35	31.77	2.21	40.69	1.14	40.84	0.91	31.43	2.36
27.79	3.84	23.25	3.84	23.35	3.84	31.79	2.76	29.79	2.76	27.95	3.25	38.09	1.24	31.16	4.71	32.89	1.96	40.99	0.96	41.14	0.73	31.55	2.01
28.95	3.58	22.15	3.95	24.51	3.58	31.85	2.44	29.95	2.44	28.01	3.15	39.25	0.98	25.66	4.84	34.05	1.56	41.05	0.64	41.3	0.41	31.71	1.92
29.39	3.34	18.26	4.01	24.87	3.34	32.09	2.24	30.09	2.24	28.36	3.05	39.69	0.74	24.15	4.95	34.41	1.42	41.29	0.44	41.44	0.21	32.27	1.82
29.85	2.83	15.63	4.12	25.14	2.83	32.65	2.11	30.65	2.11	28.65	2.83	40.15	0.23	22.06	5.05	34.68	0.95	41.85	0.31	42	0.08	32.94	1.35
29.96	2.73	10	4.65	25.79	2.73	32.86	1.5	30.96	1.5	29.01	2.73	40.26	0.13	20.95	5.16	35.33	0.84	42.06	0.3	42.31	0.07	32.99	1.08
30.26	2.66	6.52	4.89	26.34	2.66	33.06	1.23	31.06	1.23	29.36	2.66	40.56	0.06	17.47	5.2	35.88	0.23	42.26	0.29	42.41	0.06	33.54	0.74
30.85	2.36	3.21	4.92	26.82	2.36	33.85	1.02	31.85	1.02	29.56	2.36	41.15	0.05	10.25	5.26	36.36	0.06	43.05	0.28	43.2	0.05	34.02	0.36
31.25	1.02	1.52	4.95	27.56	1.02	34.25	0.53	32.25	0.53	30.14	1.02	41.55	0.04	5.12	5.42	37.1	0.04	43.45	0.25	43.6	0.04	34.76	0.22
32.83	0.01	0.01	4.99	28.5	0.01	34.83	0.00	32.83	0.00	30.25	0.01	43.13	0.01	0.00	5.59	38.04	0.03	44.03	0.01	44.18	0.00	35.13	0.00

# APPENDIX XXII: SIMILARITY REPORT

Driginality I	Report
Tu	mitin Originality Report
IN PH Ch	SITU TECHNO-ENVIROECONOMIC ANALYSIS OF A ROOF OF IOTOVOLTAIC (PV) BACKUP SYSTEM IN THE TROPICAL CLIMATE by Wilkins eruiyot
Fre	om Theses (Theses)
:	Processed on 14-Nov-2022 12:24 EAT ID: 1953487098 Word Count: 41029
Simila 29%	rity Index
Simila	rity by Source
Interne Public Studer	et Sources: 23% ations: 19% the present the source of
sour	Ces:
1	3% match (Internet from 23-Feb-2022) https://researcharchive.vuw.ac.nz/xmlui/bitstream/handle/10063/5450/thesis.pdf?sequence=
2	1% match (Internet from 20-Aug-2022) https://www.knbs.or.ke/wp-content/uploads/2021/09/Economic-Survey-2021.pdf
3 'Info	1% match () <u>"Advances in Phytochemistry, Textile and Renewable Energy Research for Industrial Growth</u> <u>brma UK Limited</u>
	< 1% match (Internet from 15-Apr-2021)
4	https://www.mdpi.com/2071-1050/13/4/1845/htm
5	< 1% match (Internet from 11-Nov-2020)
	https://www.mdpi.com/1996-1073/13/20/5502/html
6	< 1% match (Internet from 16-Mar-2020)
harmen d	https://www.mdpi.com/2227-9717/7/10/708/html
7	< 1% match (Internet from 07-Feb-2022)
	https://www.mdpi.com/1996-1073/14/9/2344/html
Land	
Land	

